### **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

<u>,                                      </u>	FORM 10-Q	_
<b>☑</b> QUARTERLY REPORT PURSUANT TO SECTION 13 C For the quarterly period ended March 31, 2021	OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF 1934
	OR	
☐ TRANSITION REPORT PURSUANT TO SECTION 13 C For the transition period from to	OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF 1934
Comm	ission File Number: 001-3939	9
	<b>j</b> amf	
	F HOLDING COR of registrant as specified in its of	
	Washington Ave S, Suite 1100 Minneapolis, MN 55401 ss of principal executive offices	82-3031543 (I.R.S. Employer Identification No.)
(Registrant's	(612) 605-6625 telephone number, including are	ea code)
Securities registe	ered pursuant to Section 12(b)	of the Act:
<u>Title of each class</u> Common Stock, \$0.001 par value per share	<u>Trading symbol</u> JAMF	Name of each exchange on which registered The NASDAQ Stock Market LLC
Indicate by check mark whether the registrant (1) has filed all reduring the preceding 12 months (or for such shorter period that requirements for the past 90 days. Yes $\boxtimes$ No $\square$		
Indicate by check mark whether the registrant has submitted ele Regulation S-T (§ 232.405 of this chapter) during the preceding files). Yes $\boxtimes$ No $\square$		
Indicate by check mark whether the registrant is a large accelerate emerging growth company. See the definitions of "large acceler company" in Rule 12b-2 of the Exchange Act.		
Large accelerated filer $\square$ Smaller reporting company $\square$	Accelerated filer□	Non-accelerated filer ⊠ Emerging growth company ⊠
If an emerging growth company, indicate by check mark if the r new or revised financial accounting standards provided pursuan		
Indicate by check mark whether the registrant is a shell compan	y (as defined in Rule 12b-2 of t	he Exchange Act). Yes □ No ⊠
On May 7, 2021, the Registrant had 117,725,132 shar	es of common stock, \$0.001 par	r value, outstanding.

### **Jamf Holding Corp.**

### INDEX

		PAGE
PART I.	FINANCIAL INFORMATION	
<u>Item 1</u> .	<u>Financial Statements (unaudited)</u>	3
	Consolidated Balance Sheets as of March 31, 2021 and December 31, 2020	3
	Consolidated Statements of Operations for the Three Months Ended March 31, 2021 and 2020	4
	Consolidated Statements of Stockholders' Equity for the Three Months Ended March 31, 2021 and 2020	5
	Consolidated Statements of Cash Flows for the Three Months Ended March 31, 2021 and 2020	6
	Notes to Consolidated Financial Statements	7
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	25
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	42
Item 4.	Controls and Procedures	42
PART II	. OTHER INFORMATION	
<u>Item 1</u> .	<u>Legal Proceedings</u>	43
Item 1A.	Risk Factors	43
<u>Item 2</u> .	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	45
Item 3.	Defaults Upon Senior Securities	45
Item 4.	Mine Safety Disclosures	45
Item 5.	Other Information	45
<u>Item 6</u> .	<u>Exhibits</u>	46
	Signatures	47

#### PART I. FINANCIAL INFORMATION

#### **Item 1. Financial Statements**

# JAMF HOLDING CORP. CONSOLIDATED BALANCE SHEETS (In thousands, except share and per share amounts)

	N	March 31, 2021 (unaudited)	Dec	ember 31, 2020
Assets				
Current assets:				
Cash and cash equivalents	\$	196,190	\$	194,868
Trade accounts receivable, net of allowances of \$603 and \$530 at March 31,				
2021 and December 31, 2020, respectively		75,882		69,056
Income taxes receivable		632		632
Deferred contract costs		11,155		9,959
Prepaid expenses		15,009		13,283
Other current assets		2,325		1,113
Total current assets		301,193		288,911
Equipment and leasehold improvements, net		16,965		12,755
Goodwill		541,850		541,480
Other intangible assets, net		197,504		202,878
Deferred contract costs		28,774		26,770
Other assets		28,898		5,359
Total assets	\$	1,115,184	\$	1,078,153
Liabilities and stockholders' equity				
Current liabilities:				
Accounts payable	\$	5,744	\$	6,967
Accrued liabilities		28,131		31,574
Income taxes payable		1,153		713
Deferred revenues		167,868		160,443
Total current liabilities		202,896		199,697
Deferred revenues, noncurrent		53,711		45,507
Deferred tax liability, net		5,475		6,422
Other liabilities		33,839		11,046
Total liabilities		295,921	-	262,672
Commitments and contingencies (Note 8)				- ,-
Stockholders' equity:				
Preferred stock, \$0.001 par value, 50,000,000 shares authorized at March 31,				
2021 and December 31, 2020; no shares issued and outstanding at March 31,				
2021 and December 31, 2020		_		_
Common stock, \$0.001 par value, 500,000,000 shares authorized at March 31,				
2021 and December 31, 2020; 117,705,895 and 116,992,472 shares issued and				
outstanding at March 31, 2021 and December 31, 2020, respectively		118		117
Additional paid-in capital		909,966		903,116
Accumulated deficit		(90,821)		(87,752)
Total stockholders' equity		819,263		815,481
Total liabilities and stockholders' equity	\$	1,115,184	\$	1,078,153
rotal natimites and stockholders equity	Ψ	1,113,104	Ψ	1,070,103

# JAMF HOLDING CORP. CONSOLIDATED STATEMENTS OF OPERATIONS (In thousands, except share and per share amounts) (unaudited)

	T		Ended March 31, 2020		
Revenue:	_	2021	-	2020	
Subscription	\$	74,923	\$	54,618	
Services	Ψ	4,003		4,010	
License		2,242		1,762	
Total revenue		81,168		60,390	
Cost of revenue:		,		,	
Cost of subscription (exclusive of amortization expense shown below)		12,014		9,248	
Cost of services (exclusive of amortization expense shown below)		2,465		3,086	
Amortization expense		2,777		2,677	
Total cost of revenue		17,256		15,011	
Gross profit		63,912		45,379	
Operating expenses:					
Sales and marketing		29,332		22,282	
Research and development		15,626		12,617	
General and administrative		16,105		11,289	
Amortization expense		5,627		5,674	
Total operating expenses		66,690		51,862	
Loss from operations		(2,778)		(6,483)	
Interest expense, net		(55)		(4,778)	
Foreign currency transaction loss		(171)		(304)	
Other income, net				55	
Loss before income tax (provision) benefit		(3,004)		(11,510)	
Income tax (provision) benefit		(65)		3,220	
Net loss	\$	(3,069)	\$	(8,290)	
Net loss per share, basic and diluted	\$	(0.03)	\$	(0.08)	
Weighted-average shares used to compute net loss per share, basic and diluted	1	17,386,322		102,860,545	

# JAMF HOLDING CORP. CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (In thousands, except share amounts) (unaudited)

	Stock Com	 <u> </u>	Additional Paid-In																																						Ac	cumulated	Sto	ockholders'
	Shares	Amount		Capital		Deficit		Equity																																				
Three Months Ended March 31, 2021:																																												
Balance, December 31, 2020	116,992,472	\$ 117	\$	903,116	\$	(87,752)	\$	815,481																																				
Exercise of stock options	713,423	1		4,018		_		4,019																																				
Share-based compensation	_	_		2,832		_		2,832																																				
Net loss	_	_		_		(3,069)		(3,069)																																				
Balance, March 31, 2021	117,705,895	\$ 118	\$	909,966	\$	(90,821)	\$	819,263																																				
Three Months Ended March 31, 2020:																																												
Balance, December 31, 2019	102,843,612	\$ 103	\$	568,756	\$	(64,981)	\$	503,878																																				
Exercise of stock options	18,792	_		103		_		103																																				
Share-based compensation	_	_		811		_		811																																				
Net loss	_	_		_		(8,290)		(8,290)																																				
Balance, March 31, 2020	102,862,404	\$ 103	\$	569,670	\$	(73,271)	\$	496,502																																				

# JAMF HOLDING CORP. CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (unaudited)

	Three Months Ended March			March 31,
		2021		2020
Cash flows from operating activities				
Net loss	\$	(3,069)	\$	(8,290)
Adjustments to reconcile net loss to cash provided by (used in) operating activities:				
Depreciation and amortization expense		9,784		9,586
Amortization of deferred contract costs		3,296		2,090
Amortization of debt issuance costs		69		279
Non-cash lease expense		1,267		_
Provision for bad debt expense and returns		159		240
Share-based compensation		2,832		811
Deferred tax benefit		(758)		(2,564)
Adjustment to contingent consideration		300		_
Other		62		10
Changes in operating assets and liabilities:				
Trade accounts receivable		(7,066)		220
Income tax receivable/payable		463		(789)
Prepaid expenses and other assets		(3,317)		(2,255)
Deferred contract costs		(6,496)		(4,788)
Accounts payable		(1,191)		(1,989)
Accrued liabilities		(7,694)		(4,928)
Deferred revenue		15,472		5,025
Other liabilities		(90)		(13)
Net cash provided by (used in) operating activities	·	4,023		(7,355)
Cash flows from investing activities				
Acquisition, net of cash acquired		(3,041)		_
Purchases of equipment and leasehold improvements		(3,290)		(1,039)
Proceeds from sale of equipment and leasehold improvements		12		_
Net cash used in investing activities	_	(6,319)		(1,039)
Cash flows from financing activities		<u> </u>		· /
Cash paid for offering costs		_		(1,465)
Proceeds from the exercise of stock options		4,019		103
Net cash provided by (used in) financing activities		4,019		(1,362)
Effect of exchange rate changes on cash and cash equivalents		(401)		
Net increase (decrease) in cash and cash equivalents	_	1,322	_	(9,756)
Cash and cash equivalents, beginning of period		194,868		32,433
Cash and cash equivalents, end of period	\$	196,190	\$	22,677
•	Ψ	130,130	Ψ	22,077
Supplemental disclosures of cash flow information:	ď	<b>n</b>	φ	4 70 4
Cash paid for interest	\$	3	\$	4,734
Cash paid for income taxes, net of refunds		351		216
Offering costs accrued but not paid		(10)		2,378
Operating lease assets obtained in exchange for operating lease liabilities		(19)		_

#### Note 1. Basis of presentation and description of business

#### **Description of business**

Jamf Holding Corp. and its wholly owned subsidiaries, collectively, are referred to as the "Company," "we," "us" or "our." We are the standard in Apple Enterprise Management, and our cloud software platform is the only vertically-focused Apple infrastructure and security platform of scale in the world. We help organizations connect, manage and protect Apple products, apps and corporate resources in the cloud without ever having to touch the devices. With our products, Apple devices can be deployed to employees brand new in the shrink-wrapped box, automatically set up and personalized at first power-on and continuously administered throughout the life of the device. Our customers are located throughout the world.

#### Vista Equity Partners acquisition

On November 13, 2017, Vista Equity Partners ("Vista") acquired a majority share of all the issued and outstanding shares of the Company at the purchase price of \$733.8 million (the "Vista Acquisition"). As of March 31, 2021, funds controlled by Vista owned approximately 61.9% of our outstanding common stock. As a result, we are a "controlled company" under NASDAQ Global Select Market ("NASDAQ") corporate governance rules.

#### **Emerging growth company status**

We are an emerging growth company as defined in the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"). Under the JOBS Act, emerging growth companies can delay adopting new or revised accounting standards issued subsequent to the enactment of the JOBS Act, until such time as those standards apply to private companies.

We have elected to use this extended transition period for complying with new or revised accounting standards that have different effective dates for public and private companies until the earlier of the date that it is (i) no longer an emerging growth company or (ii) affirmatively and irrevocably opt out of the extended transition period provided in the JOBS Act. As a result, our consolidated financial statements may not be comparable to companies that comply with the new or revised accounting pronouncements as of public company effective dates.

We will remain an emerging growth company for the first five fiscal years after the completion of our initial public offering ("IPO"), unless one of the following occurs: (i) our total annual gross revenue is at least \$1.07 billion, (ii) we have issued more than \$1.0 billion in non-convertible debt securities during the prior three year period, or (iii) we are deemed to be a large accelerated filer, which means the market value of our common stock that is held by non-affiliates exceeds \$700.0 million as of the prior June 30.

#### Basis of presentation

The accompanying consolidated financial statements have been prepared in accordance with U.S. generally accepted accounting principles ("GAAP") and include all adjustments necessary for the fair presentation of the consolidated financial position, results of operations, and cash flows of the Company.

Certain reclassifications of prior period amounts have been made to conform to the current presentation. In the fourth quarter of 2020, the Company reclassified on-premise subscription revenue from license revenue to subscription revenue in the consolidated statements of operations on a retroactive basis. The amount reclassified for the three months ended March 31, 2020 was \$4.5 million. The revised presentation is consistent with our disaggregated revenue disclosure and is more consistent with how investors and other users of the financial statements evaluate overall subscription revenue. The reclassification had no impact on total revenue.

In 2020, the Company reclassified cash paid for offering costs of \$1.5 million during the three months ended March 31, 2020 from operating activities to financing activities in the consolidated statements of cash flows as a result of the completion of our IPO in July 2020. The impact of the reclassification was not material to current or prior period financial statements.

#### **Unaudited Interim Consolidated Financial Information**

The accompanying interim consolidated balance sheet as of March 31, 2021, the consolidated statements of operations, of stockholders' equity and of cash flows for the three months ended March 31, 2021 and 2020 and the related footnote disclosures are unaudited. These unaudited interim consolidated financial statements have been prepared on the same basis as the annual consolidated financial statements and, in management's opinion, include all adjustments necessary for the fair presentation of the consolidated financial position, results of operations, and cash flows of the Company. All adjustments made were of a normal recurring nature. The results for the three months ended March 31, 2021 are not necessarily indicative of the results to be expected for the year ending December 31, 2021 or for any future period.

#### Use of estimates

The preparation of the consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities as of the reporting date, and the reported amounts of revenues and expenses during the reporting period. These estimates are based on management's best knowledge of current events and actions that the Company may undertake in the future and include, but are not limited to, revenue recognition, stock-based compensation, commissions, goodwill and accounting for income taxes. Actual results could differ from those estimates.

#### Segment and geographic information

Our chief operating decision maker ("CODM") is our Chief Executive Officer, who reviews financial information presented on a consolidated basis for purposes of making operating decisions, assessing financial performance and allocating resources. We operate our business as one operating segment and therefore we have one reportable segment.

Revenue by geographic region as determined based on the end user customer address was as follows:

	Th	ree Months l	Ended	March 31,	
	<u></u>	2021		2020	
		(in tho	usand	ands)	
The Americas	\$	61,258	\$	47,026	
Europe, the Middle East, India, and Africa		14,523		10,121	
Asia Pacific		5,387		3,243	
	\$	81,168	\$	60,390	

### Note 2. Summary of significant accounting policies

The Company's significant accounting policies are discussed in Note 2 to the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2020. Except for the accounting policies for leases that were updated as a result of adopting the new accounting standard, there have been no significant changes to these policies that have had a material impact on the Company's consolidated financial statements and related notes for the three months ended March 31, 2021. The following describes the impact of certain policies.

#### Revenue recognition

The Company applies ASC Topic 606, *Revenue from Contracts with Customers* ("ASC 606") and follows a five-step model to determine the appropriate amount of revenue to be recognized in accordance with ASC 606.

#### <u>Disaggregation of Revenue</u>

The Company separates revenue into subscription and non-subscription categories to disaggregate those revenues that are term-based and renewable from those that are one-time in nature. Revenue from subscription and non-subscription contractual arrangements are as follows:

	Three Months Ended March				
		2021		2020	
		(in tho	ousands)		
SaaS subscription and support and maintenance	\$	66,669	\$	50,078	
On-premise subscription		8,254		4,540	
Subscription revenue		74,923		54,618	
Professional services		4,003		4,010	
Perpetual licenses		2,242		1,762	
Non-subscription revenue		6,245		5,772	
Total revenue	\$	81,168	\$	60,390	

#### Contract Balances

If revenue is recognized in advance of the right to invoice, a contract asset is recorded. The balances of contract assets were \$1.2 million and \$0.9 million as of March 31, 2021 and December 31, 2020, respectively.

Contract liabilities consist of customer billings in advance of revenue being recognized. The Company invoices its customers for subscription, support and maintenance and services in advance.

Changes in contract liabilities, including revenue earned during the period from the beginning contract liability balance and new deferrals of revenue during the period, were as follows:

	 nree Months I	March 31,		
	2021		2020	
	(in tho	usand	sands)	
Balance, beginning of the period	\$ 205,950	\$	140,710	
Revenue earned	(60,925)		(47,723)	
Deferral of revenue	76,554		52,748	
Balance, end of the period	\$ 221,579	\$	145,735	

There were no significant changes to our contract assets and liabilities during the three months ended March 31, 2021 and 2020 outside of our sales activities.

#### Remaining Performance Obligations

Revenue allocated to remaining performance obligations represents contracted revenue that has not yet been recognized, which includes deferred revenue and noncancelable amounts to be invoiced. As of March 31, 2021 and December 31, 2020, the Company had \$243.1 million and \$224.5 million, respectively, of remaining performance

obligations, with 78% and 80%, respectively, expected to be recognized as revenue over the succeeding 12 months, and the remainder expected to be recognized over the three years thereafter.

#### **Deferred Contract Costs**

Sales commissions as well as associated payroll taxes and retirement plan contributions (together, contract costs) that are incremental to the acquisition of customer contracts are capitalized using a portfolio approach as deferred contract costs in the consolidated balance sheets when the period of benefit is determined to be greater than one year.

Total amortization of contract costs for the three months ended March 31, 2021 and 2020 was \$3.3 million and \$2.1 million, respectively.

The Company periodically reviews these deferred costs to determine whether events or changes in circumstances have occurred that could affect the period of benefit of these deferred contract costs. There were no impairment losses recorded during the three months ended March 31, 2021 and 2020.

### Concentration of Credit Risk

For the three months ended March 31, 2021, the Company had one distributor that accounted for more than 10% of total net revenues. Total receivables related to this distributor were \$17.3 million as of March 31, 2021. For the three months ended March 31, 2020, the Company had two distributors that accounted for more than 10% of total net revenues. Total receivables related to these distributors were \$19.8 million as of December 31, 2020.

No single end customer accounted for more than 10% of total revenue for the three months ended March 31, 2021 and 2020.

#### Recently issued accounting pronouncements not yet adopted

From time to time, new accounting pronouncements are issued by the Financial Accounting Standards Board ("FASB") or other standard setting bodies and adopted by us as of the specified effective date. Unless otherwise discussed, the impact of recently issued standards that are not yet effective will not have a material impact on our financial position or results of operations upon adoption.

#### Financial Instruments — Credit Losses

In June 2016, the FASB issued Accounting Standards Update ("ASU") No. 2016-13, *Financial Instruments* — *Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"), which introduces a model based on expected losses to estimate credit losses for most financial assets and certain other instruments. In November 2019, the FASB issued ASU No. 2019-10 *Financial Instruments* — *Credit Losses (Topic 326)*, *Derivatives and Hedging (Topic 815)*, *and Leases (Topic 842): Effective Dates* ("ASU 2019-10"). The update allows the extension of the initial effective date for entities which have not yet adopted ASU No. 2016-02, *Leases (Topic 842)* ("ASU 2016-02"). The standard is effective for annual reporting periods beginning after December 15, 2022, with early adoption permitted. Entities will apply the standard's provisions by recording a cumulative-effect adjustment to retained earnings. The Company has not yet adopted ASU 2016-13 and is currently evaluating the effect the standard will have on its consolidated financial statements.

#### Reference Rate Reform

In March 2020, the FASB issued ASU No. 2020-04, *Reference Rate Reform (Topic 848): Facilitation of the Effects of Reference Rate Reform on Financial Reporting* ("ASU 2020-04"), which provides entities with temporary

optional financial reporting alternatives to ease the potential burden in accounting for reference rate reform and includes a provision that allows entities to account for a modified contract as a continuation of an existing contract. ASU 2020-04 is effective upon issuance and can be applied through December 31, 2022. The adoption of this standard is not expected to have a material impact on the Company's consolidated financial statements.

#### Adoption of new accounting pronouncements

Leases

In February 2016, the FASB issued ASU No. 2016-02, *Leases (Topic 842)* to increase transparency and comparability among organizations related to their leasing arrangements. The update requires lessees to recognize most leases on their balance sheets, with the exception of short-term leases if a policy election is made, while recognizing lease expense on their income statements in a manner similar to current GAAP. The guidance also requires entities to disclose key quantitative and qualitative information about its leasing arrangements. The Company adopted the new lease standard on January 1, 2021 using the optional transition method to the modified retrospective approach. Under this transition provision, results for reporting periods beginning on January 1, 2021 are presented under ASC Topic 842, *Leases* ("ASC 842") while prior period amounts continue to be reported and disclosed in accordance with the Company's historical accounting treatment under ASC Topic 840, *Leases* ("ASC 840").

To reduce the burden of adoption and ongoing compliance with ASC 842, a number of practical expedients and policy elections are available under the new guidance. The Company elected the "package of practical expedients" permitted under the transition guidance, which among other things, did not require reassessment of whether contracts entered into prior to adoption are or contain leases, and allowed carryforward of the historical lease classification for existing leases. The Company has not elected to adopt the "hindsight" practical expedient, and therefore measured the right-of-use ("ROU") asset and lease liability using the remaining portion of the lease term at adoption on January 1, 2021.

The Company made an accounting policy election under ASC 842 not to recognize ROU assets and lease liabilities for leases with a term of twelve months or less. For all other leases, the Company recognizes ROU assets and lease liabilities based on the present value of lease payments over the lease term at the commencement date of the lease (or January 1, 2021 for existing leases upon the adoption of ASC 842). The ROU assets also include any initial direct costs incurred and lease payments made at or before the commencement date and are reduced by any lease incentives.

Future lease payments may include fixed rent escalation clauses or payments that depend on an index (such as the consumer price index). Subsequent changes to an index and other periodic market-rate adjustments to base rent are recorded in variable lease expense in the period incurred.

The Company has made an accounting policy election to account for lease and non-lease components in its contracts as a single lease component for all asset classes. The non-lease components typically represent additional services transferred to the Company, such as common area maintenance for real estate, which are variable in nature and recorded in variable lease expense in the period incurred.

The Company uses its incremental borrowing rate to determine the present value of lease payments as the Company's leases do not have a readily determinable implicit discount rate. The incremental borrowing rate is the rate of interest the Company would have to pay to borrow on a collateralized basis over a similar term and amount in a similar economic environment. Judgement is applied in assessing factors such as Company specific credit risk, lease term, nature and quality of the underlying collateral, currency and economic environment in determining the incremental borrowing rate to apply to each lease.

Upon adoption, the Company recorded ROU assets and lease liabilities of approximately \$25.0 million and \$28.6 million, respectively, related to the Company's operating leases. The adoption of the new lease standard did not materially impact our consolidated statements of operations or consolidated statements of cash flows. See Note 6 for more information.

Debt with Conversion and Other Options and Contracts in Entity's Own Equity

In August 2020, the FASB issued ASU No. 2020-06, *Debt — Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging — Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity ("ASU 2020-06")*, which simplifies the accounting for certain financial instruments with characteristics of liabilities and equity. Among other changes, the standard eliminates the beneficial conversion and cash conversion accounting models for convertible instruments. As a result, entities will account for a convertible debt instrument wholly as debt unless the instrument contains features that require bifurcation as a derivative in accordance with ASC Topic 815, *Derivatives and Hedging*, or a convertible debt instrument was issued at a substantial premium. In addition, the amendments also require the if-converted method to be applied for all convertible instruments when calculating diluted earnings per share. The standard is effective for fiscal years beginning after December 15, 2021, including interim periods within those fiscal years. Early adoption is permitted. The Company early adopted the new standard on January 1, 2021. The adoption of the standard did not have a material impact on the Company's consolidated financial statements.

#### Note 3. Financial instruments fair value

We report financial assets and liabilities and nonfinancial assets and liabilities that are recognized or disclosed at fair value in the consolidated financial statements on a recurring basis in accordance with ASC Topic 820, *Fair Value Measurement* ("ASC 820"). ASC 820 defines fair value as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. When determining the fair value measurements for assets and liabilities, which are required to be recorded at fair value, we consider the principal or most advantageous market in which we would transact and the market-based risk measurements or assumptions that market participants would use in pricing the asset or liability, such as inherent risk, transfer restrictions and credit risk.

ASC 820 also establishes a fair value hierarchy, which prioritizes the inputs to valuation techniques used to measure fair value into three levels. Fair value represents the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. GAAP established a hierarchy framework to classify the fair value based on the observability of significant inputs to the measurement. The levels of the fair value hierarchy are as follows:

- Level 1: Fair value is determined using an unadjusted quoted price in an active market for identical assets or liabilities.
- Level 2: Fair value is estimated using inputs other than quoted prices included within Level 1 that are observable, either directly or indirectly.
- Level 3: Fair value is estimated using unobservable inputs that are significant to the fair value of the assets or liabilities

The Company invests in money market funds and U.S. Treasuries with original or remaining maturities at the time of purchase of three months or less, which are measured and recorded at fair value on a recurring basis. Money market funds are valued based on quoted market prices in active markets and classified within Level 1 of the fair value

hierarchy. U.S. Treasuries include treasury bills that generally mature within 30 days and are classified within Level 1 of the fair value hierarchy. The fair value of these financial instruments were as follows:

	March 31, 2021						
	Level 1	Level 2		l 2 Level 3		Total	
		(in thousands)					
Cash equivalents:							
Money market funds	\$ 150,006	\$	_	\$	_	\$ 150,006	
Total cash equivalents	\$ 150,006	\$	_	\$	_	\$ 150,006	
		]	Decembe	r 31, 2	2020		
	Level 1	L	evel 2	Le	vel 3	Total	
			(in tho	usand	s)		
Cash equivalents:							
Money market funds	\$ 100,000	\$	_	\$	_	\$ 100,000	
U.S. Treasuries	25,000					25,000	
Total cash equivalents	\$ 125,000	\$		\$		\$ 125,000	

The carrying value of accounts receivable and accounts payable approximate their fair value due to their short maturities and are excluded from the tables above.

#### **Note 4. Acquisitions**

cmdReporter

On February 26, 2021, the Company entered into an asset purchase agreement with cmdSecurity Inc. ("cmdSecurity") to acquire certain cmdSecurity assets, including cmdReporter, a suite of security and compliance tools purpose-built for macOS. With cmdReporter, the Company will further extend the security capabilities of its expansive Apple Enterprise Management platform. cmdSecurity's software complements the Company's existing product offerings. The Company accounted for the acquisition by applying the acquisition method of accounting for business combinations in accordance with ASC Topic 805, *Business Combinations* ("ASC 805"). The final aggregate purchase price was approximately \$3.4 million. This acquisition was funded by the Company's cash on hand and includes future contingent consideration due to the sellers. The goodwill represents the excess of the purchase consideration over the fair value of the underlying net identifiable assets. The goodwill recognized in this acquisition is primarily attributable to the cmdSecurity assembled workforce and was not material. The acquired intangible assets and goodwill are deductible for income tax purposes.

At the time of the acquisition, the contingent consideration was valued at \$0.4 million, which was based on the acquired business signing new business or renewing acquired contracts during the 90 days following the close of the acquisition. The estimated fair value of these contingent payments is determined using projected contract wins, which uses Level 3 inputs for fair value measurements, including assumptions about the probability of closing contracts based on their current stage in the sales process.

Substantially all of the purchase price consideration related to the fair value of the acquired separately identifiable intangibles assets, which related to acquired developed technology and in-process research and development ("IPR&D"). The fair value of the identifiable intangible assets was estimated using the replacement cost method, whereby the components of the acquired intangibles were reviewed to determine the cumulative cost of development for each component, inclusive of a developer's profit and an entrepreneurial incentive. The cumulative cost of development was not discounted to account for obsolescence factor as the replacement cost accounted for present day development. The developed technology will be amortized over its estimated weighted-average useful life, which was determined to be

5.0 years. The IPR&D is an indefinite lived intangible asset that is not amortized, but will be evaluated at least annually for impairment. For more details on the intangible assets, see Note 5.

Acquisition-related expenses were expensed as incurred and totaled \$0.1 million for the three months ended March 31, 2021. These expenses were recognized as acquisition costs in general and administrative expenses in the consolidated statement of operations.

The Company allocated the net purchase consideration to the net assets acquired based on their respective fair values at the time of the acquisition as follows (in thousands):

Cash consideration	\$ 3,041
Contingent consideration	359
Final aggregate purchase price	\$ 3,400
Intangible assets acquired:	
Developed technology	\$ 2,630
IPR&D	400
Goodwill	370
Total purchase consideration	\$ 3,400

#### Digita Security LLC

In 2019, the Company recorded contingent consideration in connection with its purchase of the outstanding membership interests of Digita. The maximum contingent consideration is \$15.0 million if the acquired business achieves certain revenue milestones by December 31, 2022. The estimated fair value of these contingent payments is determined using a Monte Carlo simulation model, which uses Level 3 inputs for fair value measurements, including assumptions about the probability of growth of subscription services and the related pricing of the services offered. During the three months ended March 31, 2021, the fair value of the contingent consideration was increased by \$0.3 million, which was reflected in general and administrative expenses in the consolidated statement of operations. The adjustment for the three months ended March 31, 2021 primarily reflects an increase in the liability due to updated market assumptions. As of March 31, 2021 and December 31, 2020, the fair value of the contingent consideration was \$8.5 million and \$8.2 million, respectively, which is included in other liabilities in the consolidated balance sheets.

### Note 5. Goodwill and other intangible assets

The change in the carrying amount of goodwill is as follows:

	T	Three Months Ended Marc			
		2021	2021		
		(in thousands)			
Goodwill, beginning of period	\$	541,480	\$	539,818	
Goodwill acquired		370			
Goodwill, end of period	\$	541,850	\$	539,818	

The gross carrying amount and accumulated amortization of intangible assets other than goodwill are as follows:

					Weighted- Average
	Useful Life	Gross Value	Accumulated Amortization	Net Carrying Value	Remaining Useful Life
			(in thousands)		
Trademarks	8 years	\$ 34,320	\$ 13,454	\$ 20,866	4.8 years
Customer relationships	2 - 12 years	214,428	55,810	158,618	8.7 years
Developed technology	5 years	54,563	31,173	23,390	2.3 years
Non-competes	2 years	90	86	4	0.1 years
Balance, December 31, 2020		\$ 303,401	\$ 100,523	\$ 202,878	
Trademarks	8 years	\$ 34,300	\$ 14,506	\$ 19,794	4.6 years
Customer relationships	2 - 12 years	214,408	60,342	154,066	8.5 years
Developed technology	5 years	57,193	33,949	23,244	2.4 years
Non-competes	2 years	90	90	_	_
In-process research and development	Indefinite	400		400	
Balance, March 31, 2021		\$ 306,391	\$ 108,887	\$ 197,504	

Amortization expense was \$8.4 million for both the three months ended March 31, 2021 and 2020.

There were no impairments to goodwill or intangible assets recorded for the three months ended March 31, 2021 and 2020.

#### Note 6. Leases

The Company determines if an arrangement is or contains a lease at inception, which is the date on which the terms of the contract are agreed to and the agreement creates enforceable rights and obligations. Under ASC 842, a contract is or contains a lease when (i) explicitly or implicitly identified assets have been deployed in the contract and (ii) the customer obtains substantially all of the economic benefits from the use of that underlying asset and directs how and for what purpose the asset is used during the term of the contract. The Company also considers whether its service arrangements include the right to control the use of an asset. See Note 2 for more information on the Company's accounting policies for leases.

The Company leases office facilities and vehicles under operating lease agreements that have initial terms ranging from 1 to 9 years. Some leases include one or more options to renew, generally at our sole discretion, with renewal terms that can extend the lease term up to 10 years. In addition, certain leases contain termination options, where the rights to terminate are held by either the Company, the lessor, or both parties. These options to extend or terminate a lease are included in the lease terms when it is reasonably certain that the Company will exercise that option. The Company's leases generally do not contain any material restrictive covenants or residual value guarantees. The Company also leases office equipment under a finance lease agreement with a term of 4 years. The Company's finance lease was not material to the consolidated financial statements as of March 31, 2021.

Supplemental balance sheet information related to the Company's operating leases is as follows:

Leases	Balance Sheet Classification	 March 31, 2021
		(in thousands)
Assets		
Operating lease assets	Other assets	\$ 23,868
Liabilities		
Operating lease liabilities - current	Accrued liabilities	\$ 4,870
Operating lease liabilities - non-current	Other liabilities	23,184
Total operating lease liabilities		\$ 28,054

The weighted-average remaining term of the Company's operating leases was 6.5 years as of March 31, 2021. The weighted-average discount rate used to measure the present value of the operating lease liabilities was 3.5% as of March 31, 2021.

The components of lease expense were as follows:

		ree Months Ended March 31, 2021 (in thousands)
		,
Operating lease cost	\$	1,514
Short-term lease cost		52
Variable lease cost		439
Total lease expense	\$	2,005

Operating lease cost is recognized on a straight-line basis over the lease term. The Company leases certain office facilities with a related party, including the office space in Eau Claire, Wisconsin. Operating lease cost with related parties was \$0.3 million for the three months ended March 31, 2021.

Total rent expense, including the Company's share of the lessors' operating expenses, was \$1.4 million for the three months ended March 31, 2020. Rent expense with related parties, including the Company's share of the lessors' operating expenses, was \$0.3 million for the three months ended March 31, 2020.

For the three months ended March 31, 2021, operating cash flows included \$1.2 million of cash paid for operating lease liabilities.

Maturities of the Company's operating lease liabilities as of March 31, 2021 were as follows:

	nting Leases housands)
Years ending December 31:	
2021 (remaining nine months)	\$ 4,326
2022	5,407
2023	5,331
2024	4,589
2025	2,519
Thereafter	9,482
Total lease payments	31,654
Less: imputed interest	3,600
Total present value of lease liabilities	\$ 28,054

#### Note 7. Debt

On July 27, 2020, the Company entered into a new secured credit agreement (the "New Credit Agreement") for an initial revolving credit facility of \$150.0 million (the "New Revolving Credit Facility"), which may be increased or decreased under specific circumstances, with a \$25.0 million letter of credit sublimit and a \$50.0 million alternative currency sublimit. In addition, the New Credit Agreement provides for the ability of the Company to request incremental term loan facilities in a minimum amount of \$5.0 million for each facility. The maturity date of the New Credit Agreement is July 27, 2025. The New Credit Agreement contains customary representations and warranties, affirmative covenants, reporting obligations, negative covenants and events of default. We were in compliance with such covenants as of both March 31, 2021 and December 31, 2020. As of both March 31, 2021 and December 31, 2020, we had \$1.0 million of letters of credit outstanding under our New Revolving Credit Facility. In the third quarter of 2020, the Company recorded debt issuance costs of \$1.3 million, which are amortized to interest expense over the term of the New Credit Agreement. As of both March 31, 2021 and December 31, 2020, debt issuance costs of \$1.1 million were included in other assets in the consolidated balance sheets.

### Note 8. Commitments and contingencies

#### Contingencies

From time to time, the Company may be subject to various claims, charges and litigation. The Company records a liability when it is both probable that a liability has been incurred and the amount of the loss can be reasonably estimated. The Company maintains insurance to cover certain actions and believes that resolution of such claims, charges, or litigation will not have a material impact on the Company's financial position, results of operations, or liquidity. The Company had no liabilities for contingencies recorded as of March 31, 2021 and December 31, 2020.

#### Note 9. Net loss per share

The following table sets forth the computation of basic and diluted net loss per share:

	Three Months Ended March 31,			March 31,
	2021 2			2020
	(in thousands, except share and p share amounts)			
Numerator:				
Net loss	\$	(3,069)	\$	(8,290)
Denominator:				
Weighted-average shares used to compute net loss per share, basic and diluted	117	7,386,322	1	02,860,545
Basic and diluted net loss per share	\$	(0.03)	\$	(80.0)

Basic net loss per share is computed by dividing the net loss by the weighted-average number of common shares outstanding for the period. Because we have reported a net loss for the three months ended March 31, 2021 and 2020, the number of shares used to calculate diluted net loss per common share is the same as the number of shares used to calculate basic net loss per common share because the potentially dilutive shares would have been antidilutive if included in the calculation.

The following potentially dilutive securities outstanding have been excluded from the computation of diluted weighted-average shares outstanding because such securities have an antidilutive impact due to losses reported:

	Three Months E	nded March 31,
	2021	2020
Stock options outstanding	6,521,067	7,742,158
Unvested restricted stock units	1,374,401	36,520
Total potential dilutive securities	7,895,468	7,778,678

#### Note 10. Share-based compensation

On July 21, 2020, the Company adopted the Jamf Holding Corp. Omnibus Incentive Plan (the "2020 Plan"). The 2020 Plan provides for grants of (i) stock options, (ii) stock appreciation rights, (iii) restricted shares, (iv) performance awards, (v) other share-based awards and (vi) other cash-based awards to eligible employees, non-employee directors and consultants of the Company. We initially reserved 14,800,000 shares of our common stock for issuance under the 2020 Plan. The total number of shares reserved for issuance under the 2020 Plan increases on January 1st of each of the first 10 calendar years during the term of the 2020 Plan by the lesser of: (i) a number of shares of our common stock equal to 4% of the total number of shares of our common stock outstanding on December 31st of the preceding calendar year or (ii) a number of shares of our common stock as determined by our board of directors. The maximum number of shares of common stock available for issuance under the 2020 Plan was 19,479,699 shares as of January 1, 2021. As of March 31, 2021, 18,105,298 shares of common stock are reserved for additional grants under the 2020 Plan.

The 2017 Stock Option Plan ("2017 Option Plan") became effective November 13, 2017 upon the approval of the board of directors and, prior to the adoption of the 2020 Plan, served as the umbrella plan for the Company's stock-based and cash-based incentive compensation program for its officers and other eligible employees. The aggregate number of shares of common stock that may be issued under the 2017 Option Plan may not exceed 8,470,000 shares. As of March 31, 2021, 128,928 shares of common stock are reserved for additional grants under the 2017 Option Plan. All stock options granted by the Company were at an exercise price at or above the estimated fair market value of the Company's common stock as of the grant date. No options were granted during the three months ended March 31, 2021.

The table below summarizes return target options activity for the three months ended March 31, 2021:

	Options	A E	eighted- verage xercise Price	Weighted- Average Remaining Contractual Term (Years)	Aggregate Intrinsic Value thousands)
Outstanding, December 31, 2020	3,687,664	\$	6.75	7.8	\$ 85,444
Granted	_		_		_
Exercised	_		_		_
Forfeitures	_		_		_
Outstanding, March 31, 2021	3,687,664	\$	6.75	7.5	\$ 105,358
Options exercisable at March 31, 2021		\$	_	_	\$ _
Vested or expected to vest at March 31, 2021	_	\$	_	_	\$ _

There was approximately \$33.0 million of unrecognized compensation expense related to these return target options as of March 31, 2021. The aggregate intrinsic value in the table above represents the total intrinsic value that would have been received by the optionholders had all optionholders exercised their options.

Restricted stock unit ("RSU") activity for the three months ended March 31, 2021 was as follows:

		P	er Unit
	Units	Fa	ir Value
Outstanding, December 31, 2020	1,293,107	\$	26.34
Granted	88,764		37.37
Restrictions lapsed	_		_
Forfeited	(7,470)		26.00
Outstanding, March 31, 2021	1,374,401	\$	27.05

RSUs under the 2020 Plan vest ratably over four years. RSUs under the 2017 Option Plan vest 100% on the one-year anniversary of the date of the grant. The estimated compensation cost of each RSU, which is equal to the fair value of the award on the date of grant, is recognized on a straight-line basis over the vesting period. There was \$31.3 million of total unrecognized compensation cost related to unvested restricted stock that is expected to be recognized over a weighted-average period of 3.4 years as of March 31, 2021.

The table below summarizes the service-based option activity for the three months ended March 31, 2021:

	Options	Av Ex	eighted- verage kercise Price	Weighted- Average Remaining Contractual Term (Years)	]	Aggregate Intrinsic Value thousands)
Outstanding, December 31, 2020	3,546,826	\$	5.65	7.1	\$	86,098
Granted	_		_			_
Exercised	(713,423)		5.63			21,328
Forfeitures	_		_			_
Outstanding, March 31, 2021	2,833,403	\$	5.65	6.8	\$	84,072
Options exercisable at March 31, 2021	1,790,693	\$	5.50	6.7	\$	53,395
Vested or expected to vest at March 31, 2021	2,833,403	\$	5.65	6.8	\$	84,072

The aggregate intrinsic value in the table above represents the total intrinsic value that would have been received by the optionholders had all optionholders exercised their options on the last date of the period. The total fair

value of service-based options vested during the three months ended March 31, 2021 was \$0.7 million. There was \$2.7 million of unrecognized compensation expense related to service-based options that is expected to be recognized over a weighted-average period of 1.5 years as of March 31, 2021.

The Company recognized stock-based compensation expense as follows:

	Thre	Three Months Ended March 31,			
		2021		2020	
		(in tho	usands)		
Cost of revenue:					
Subscription	\$	324	\$	38	
Services		77			
Sales and marketing		842		111	
Research and development		778		157	
General and administrative		811		505	
	\$	2,832	\$	811	

#### Note 11. Income taxes

The Company's effective tax rates for the three months ended March 31, 2021 and 2020 were (2.2)% and 28.0%, respectively. The effective tax rate for the three months ended March 31, 2021 was lower than the prior year period due to Section 162(m) of the Internal Revenue Code, stock option activity and the domestic valuation allowance. The effective tax rate for the three months ended March 31, 2021 differs from the statutory rate primarily as a result of the domestic valuation allowance. The effective tax rate for the three months ended March 31, 2021 was impacted by less than \$0.1 million of discrete income tax expense. The Company's annual effective tax rates for the three months ended March 31, 2021 and 2020 were (0.5)% and 25.2%, respectively.

#### Note 12. Related-party transactions

The Company made pledges to the Jamf Nation Global Foundation ("JNGF") of \$0.1 million for the three months ended March 31, 2021. The Company did not make any pledges to the JNGF for the three months ended March 31, 2020. As of March 31, 2021 and December 31, 2020, the Company accrued \$0.4 million and \$0.9 million, respectively, related to JNGF pledges, which are included in accrued liabilities in the consolidated balance sheets. The Company has an ongoing lease agreement for office space in Eau Claire, Wisconsin with an entity in which a related party is a minority owner. See Note 6 for further discussion of this lease agreement. The Company may engage in transactions in the ordinary course of business with other companies whose directors or officers may also serve as directors or officers for the Company. The Company carries out these transactions on customary terms.

Vista is a U.S.-based investment firm that controls the funds which own a majority of the Company. The Company has paid for consulting services and other expenses related to services provided by Vista and Vista affiliates. The total expenses incurred by the Company for these services were less than \$0.1 million and \$0.2 million for the three months ended March 31, 2021 and 2020, respectively. The Company had less than \$0.1 million in accounts payable related to these expenses as of both March 31, 2021 and December 31, 2020.

The Company also has revenue arrangements with Vista affiliates. The Company recognized revenue related to these arrangements of \$0.2 million and \$0.3 million for the three months ended March 31, 2021 and 2020, respectively. The Company had \$0.2 million in accounts receivable related to these agreements as of March 31, 2021 and \$0.3 million in accounts receivable related to these agreements as of December 31, 2020.

In addition, the Company pays for services with Vista affiliates in the normal course of business. The total expenses incurred by the Company for services with Vista affiliates were \$0.2 million and \$0.1 million for the three months ended March 31, 2021 and 2020, respectively. The Company had less than \$0.1 million in accounts payable related to these expenses as of March 31, 2021 and \$0.1 million in accounts payable related to these expenses as of December 31, 2020.

Prior to its termination and repayment in full on July 27, 2020, the Company was party to a term loan facility (the "Prior Term Loan") and revolving credit facility with a consortium of lenders for a principal amount of \$205.0 million and principal committed amount of \$15.0 million, respectively. During the three months ended March 31, 2020, affiliates of Vista were paid \$0.8 million in interest on the portion of the Prior Term Loan held by them.

#### Note 13. Subsequent event

On May 11, 2021, the Company announced that it has signed a definitive agreement (the "Merger Agreement") to acquire Wandera, Inc. ("Wandera"), a leader in zero trust cloud security and access for mobile devices. Under the terms of the Merger Agreement, Jamf will acquire Wandera for total cash consideration of \$400.0 million subject to customary adjustments, including working capital, indebtedness, closing cash on hand and transaction expenses. The total consideration consists of an initial payment of \$350.0 million at close and deferred consideration of \$50.0 million to be paid in \$25.0 million increments on October 1, 2021 and December 15, 2021. The transaction is expected to close during the third quarter of 2021, subject to the satisfaction of customary closing conditions, including required U.S. regulatory approvals.

#### **Forward-Looking Statements**

This Quarterly Report on Form 10-Q contains forward-looking statements that are subject to risks and uncertainties. All statements other than statements of historical fact included in this Quarterly Report on Form 10-Q are forward-looking statements. Forward-looking statements give our current expectations and projections relating to our financial condition, results of operations, plans, objectives, future performance and business. You can identify forward-looking statements by the fact that they do not relate strictly to historical or current facts. These statements may include words such as "anticipate," "estimate," "expect," "project," "plan," "intend," "believe," "may," "will," "should," "can have," "likely" and other words and terms of similar meaning in connection with any discussion of the timing or nature of future operating or financial performance or other events. For example, all statements we make relating to our estimated and projected costs, expenditures, cash flows, growth rates and financial results or our plans and objectives for future operations, growth initiatives, or strategies are forward-looking statements. All forward-looking statements are subject to risks and uncertainties that may cause actual results to differ materially from those that we expected, including:

- the ability of Jamf and Wandera to close the announced transaction;
- the ability of Jamf to realize the potential benefits of the acquisition of Wandera;
- the possibility that the closing of the transaction may be delayed;
- other risks related to Jamf's integration of Wandera's business, team, and technology;
- the impact on our operations and financial condition from the effects of the current COVID-19 pandemic;
- the potential impact of customer dissatisfaction with Apple or other negative events affecting Apple services and devices, and failure of enterprises to adopt Apple products;
- the potentially adverse impact of changes in features and functionality by Apple on our engineering focus or product development efforts;
- changes in our continued relationship with Apple;
- the fact that we are not party to any exclusive agreements or arrangements with Apple;
- our reliance, in part, on channel partners for the sale and distribution of our products;
- the impact of reputational harm if users perceive our products as the cause of device failure;
- our ability to successfully develop new products or materially enhance current products through our research and development efforts;
- our ability to continue to attract new customers;
- our ability to retain our current customers;
- our ability to sell additional functionality to our current customers;
- our ability to meet service-level commitments under our subscription agreements;
- our ability to correctly estimate market opportunity and forecast market growth;

- risks associated with failing to continue our recent growth rates;
- our dependence on one of our products for a substantial portion of our revenue;
- our ability to scale our business and manage our expenses;
- our ability to change our pricing models, if necessary to compete successfully;
- the impact of delays or outages of our cloud services from any disruptions, capacity limitations or interferences of third-party data centers that host our cloud services, including Amazon Web Services ("AWS");
- our ability to maintain, enhance and protect our brand;
- our ability to maintain our corporate culture;
- the ability of Jamf Nation to thrive and grow as we expand our business;
- the potential impact of inaccurate, incomplete or misleading content that is posted on Jamf Nation;
- our ability to offer high-quality support;
- risks and uncertainties associated with potential acquisitions and divestitures, including, but not limited to,
  disruptions to ongoing operations; diversions of management from day-to-day responsibilities; adverse impacts
  on our financial condition; failure of an acquired business to further our strategy; uncertainty of synergies;
  personnel issues; resulting lawsuits and issues unidentified in diligence processes;
- our ability to predict and respond to rapidly evolving technological trends and our customers' changing needs;
- our ability to compete with existing and new companies;
- the impact of adverse general and industry-specific economic and market conditions;
- the impact of reductions in IT spending;
- our ability to attract and retain highly qualified personnel;
- risks associated with competitive challenges faced by our customers;
- the impact of our often long and unpredictable sales cycle;
- our ability to develop and expand our marketing and sales capabilities;
- the risks associated with sales to new and existing enterprise customers;
- the risks associated with free trials and other inbound, lead-generation sales strategies;
- the risks associated with indemnity provisions in our contracts;
- our management team's limited experience managing a public company;
- the impact of any catastrophic events;

- the impact of global economic conditions;
- risks associated with cyber-security events;
- the impact of real or perceived errors, failures or bugs in our products;
- the impact of interruptions or performance problems associated with our technology or infrastructure;
- the impact of general disruptions to data transmission;
- risks associated with stringent and changing privacy laws, regulations and standards, and information security
  policies and contractual obligations related to data privacy and security;
- the risks associated with intellectual property infringement, misappropriation or other claims;
- our reliance on third-party software and intellectual property licenses;
- our ability to obtain, protect, enforce and maintain our intellectual property and proprietary rights;
- the risks associated with our use of open source software in our products; and
- other factors disclosed in the section entitled "Risk Factors" and elsewhere in our Annual Report on Form 10-K
  and this Quarterly Report on Form 10-Q.

We derive many of our forward-looking statements from our operating budgets and forecasts, which are based on many detailed assumptions. While we believe that our assumptions are reasonable, we caution that it is very difficult to predict the impact of known factors, and it is impossible for us to anticipate all factors that could affect our actual results. Important factors that could cause actual results to differ materially from our expectations, or cautionary statements, are disclosed under "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in our Annual Report on Form 10-K and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Part I, Item 2 of this Quarterly Report on Form 10-Q. All written and oral forward-looking statements attributable to us, or persons acting on our behalf, are expressly qualified in their entirety by these cautionary statements as well as other cautionary statements that are made from time to time in our other SEC filings and public communications. You should evaluate all forward-looking statements in the context of these risks and uncertainties.

We caution you that the important factors referenced above may not contain all of the factors that are important to you. In addition, we cannot assure you that we will realize the results or developments we expect or anticipate or, even if substantially realized, that they will result in the consequences or affect us or our operations in the way we expect. The forward-looking statements included in this Quarterly Report on Form 10-Q are made only as of the date hereof. We undertake no obligation to update or revise any forward-looking statement as a result of new information, future events or otherwise, except as otherwise required by law.

### Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis summarizes the significant factors affecting the consolidated operating results, financial condition, liquidity and cash flows of our company as of and for the periods presented below. The following discussion and analysis should be read in conjunction with our consolidated financial statements and the related notes included elsewhere in this Quarterly Report on Form 10-Q and our consolidated financial statements and the related notes in our Annual Report on Form 10-K. The discussion contains forward-looking statements that are based on the beliefs of management, as well as assumptions made by, and information currently available to, our management. Actual results could differ materially from those discussed in or implied by forward-looking statements as a result of various factors, including those discussed below and in our Annual Report on Form 10-K, particularly in the sections entitled "Risk Factors" and "Forward-Looking Statements."

#### Overview

We are the standard in Apple Enterprise Management, and our cloud software platform is the only vertically-focused Apple infrastructure and security platform of scale in the world. We help organizations, including businesses, hospitals, schools and government agencies, connect, manage and protect Apple products, apps and corporate resources in the cloud without ever having to touch the devices. With Jamf's software, Apple devices can be deployed to employees brand new in the shrink-wrapped box, set up automatically and personalized at first power-on and administered continuously throughout the life of the device.

Jamf was founded in 2002, around the same time that Apple was leading an industry transformation. Apple transformed the way people access and utilize technology through its focus on creating a superior consumer experience. With the release of revolutionary products like the Mac, iPod, iPhone and iPad, Apple built the world's most valuable brand and became ubiquitous in everyday life.

We have built our company through a singular focus on being the primary solution for Apple in the enterprise. Through our long-standing relationship with Apple, we have accumulated significant Apple technical experience and expertise that give us the ability to fully and quickly leverage and extend the capabilities of Apple products, operating systems and services. This expertise enables us to fully support new innovations and operating system releases the moment they are made available by Apple. This focus has allowed us to create a best-in-class user experience for Apple in the enterprise.

We sell our SaaS solutions via a subscription model, through a direct sales force, online and indirectly via our channel partners, including Apple. Our multi-dimensional go-to-market model and cloud-deployed offering enable us to reach all organizations around the world, large and small, with our software solutions. As a result, we continue to see rapid growth and expansion of our customer base as Apple continues to gain momentum in the enterprise.

On May 11, 2021, we announced the acquisition of Wandera, a leader in zero trust cloud security and access for mobile devices, extending our leadership in Apple Enterprise Management. See Note 13 for further information on the acquisition. Until close, the companies will continue to operate independently. We currently intend to finance the acquisition with a combination of cash on hand and debt financing.

#### Response to COVID-19

With social distancing measures having been implemented to curtail the spread of COVID-19, we enacted a robust business continuity plan, including a global work-from-home policy for all of our employees. We believe our internal cloud-first technology platforms have allowed for a seamless transition to a remote working environment without any material impacts to our business, highlighting the resilience of our business model. Our product portfolio and platform has enabled our commercial customers to continue with their efforts to work remotely, our K-12 and higher-education customers to deliver distance learning and our health-care customers to provide quality care via a telehealth model, a solution that was conceptualized and released *during* the current pandemic. We believe that a

business like ours is well-suited to navigate the current environment in which customers are focused on effectively conducting business remotely, while the underlying demand for our core products remains relatively unchanged.

The extent to which the COVID-19 pandemic affects our business will depend on future developments in the United States and around the world, which are highly uncertain and cannot be predicted, including new information which may emerge concerning the severity of the coronavirus and the actions required to contain and treat it, among others. Although the ultimate impact of the COVID-19 pandemic on our business and financial results remains uncertain, a continued and prolonged public health crisis such as the COVID-19 pandemic could have a material negative impact on our business, operating results and financial condition. See "Risk Factors — Risks Relating to Our Business — The COVID-19 pandemic could materially adversely affect our business, operating results, financial condition and prospects" included in Part II, Item 1A in this Quarterly Report on Form 10-Q for additional information.

#### **Key Factors Affecting Our Performance**

Our historical financial performance has been, and we expect our financial performance in the future to be, driven by our ability to:

Attract new customers. Our ability to attract new customers is dependent upon a number of factors, including the effectiveness of our pricing and solutions, the features and pricing of our competitors' offerings, the effectiveness of our marketing efforts, the effectiveness of our channel partners in selling, marketing and deploying our software solutions and the growth of the market for Apple devices and services for SMBs and enterprises. Sustaining our growth requires continued adoption of our platform by new customers. We intend to continue to invest in building brand awareness as we further penetrate our addressable markets. We intend to expand our customer base by continuing to make significant and targeted investments in our direct sales and marketing to attract new customers and to drive broader awareness of our software solutions.

Expand within our customer base. Our ability to increase revenue within our existing customer base is dependent upon a number of factors, including their satisfaction with our software solutions and support, the features and pricing of our competitors' offerings and our ability to effectively enhance our platform by developing new products and features and addressing additional use cases. Often our customers will begin with a small deployment and then later expand their usage more broadly within the enterprise as they realize the benefits of our platform. We believe that our "land and expand" business model allows us to efficiently increase revenue from our existing customer base. We intend to continue to invest in enhancing awareness of our software solutions, creating additional use cases, and developing more products, features, and functionality, which we believe are important factors to expand usage of our software solutions by our existing customer base. We believe our ability to retain and expand usage of our software solutions by our existing customer base is evidenced by our dollar-based net retention rate.

Sustain product innovation and technology leadership. Our success is dependent on our ability to sustain product innovation and technology leadership in order to maintain our competitive advantage. We believe that we have built a highly differentiated platform and we intend to further extend the adoption of our platform through additional innovation. While sales of subscriptions to our Jamf Pro product account for most of our revenue, we intend to continue to invest in building additional products, features and functionality that expand our capabilities and facilitate the extension of our platform to new use cases. Our future success is dependent on our ability to successfully develop, market and sell additional products to both new and existing customers. For example, in 2018, we introduced Jamf Connect to provide users with a seamless connection to corporate resources using a single identity and in 2019 we introduced Jamf Protect to extend Apple's security and privacy model to enterprise teams by creating unprecedented visibility into MacOS fleets through customized remote monitoring and threat detection and prevention.

Continue investment in growth. Our ability to effectively invest for growth is dependent upon a number of factors, including our ability to offset anticipated increases in operating expenses with revenue growth, our ability to spend our research and development budget efficiently or effectively on compelling innovation and technologies, our ability to accurately predict costs and our ability to maintain our corporate culture as our headcount expands. We plan to continue investing in our business so we can capitalize on our market opportunity. We intend to grow our sales team to target expansion within our midmarket and enterprise customers and to attract new customers. We expect to continue to

make focused investments in marketing to drive brand awareness and enhance the effectiveness of our customer acquisition model. We also intend to continue to add headcount to our research and development team to develop new and improved products, features and functionality. Although these investments may increase our operating expenses and, as a result, adversely affect our operating results in the near term, we believe they will contribute to our long-term growth.

Continue international expansion. Our international growth in any region will depend on our ability to effectively implement our business processes and go-to-market strategy, our ability to adapt to market or cultural differences, the general competitive landscape, our ability to invest in our sales and marketing channels, the maturity and growth trajectory of Apple devices and services by region and our brand awareness and perception. We plan to continue making investments in our international sales and marketing channels to take advantage of this market opportunity while refining our go-to-market approach based on local market dynamics. While we believe global demand for our platform will increase as international market awareness of Jamf grows, our ability to conduct our operations internationally will require considerable management attention and resources and is subject to the particular challenges of supporting a growing business in an environment of multiple languages, cultures, customs, legal and regulatory systems (including with respect to data transfer and privacy), alternative dispute systems and commercial markets. In addition, global demand for our platform and the growth of our international operations is dependent upon the rate of market adoption of Apple products in international markets.

Enhance our offerings via our partner network. Our success is dependent not only on our independent efforts to innovate, scale and reach more customers directly but also on the success of our partners to continue to gain share in the enterprise. With a focus on the user and being the bridge between critical technologies — with Apple and Microsoft as two examples — we feel we can help other market participants deliver more to enterprise users with the power of Jamf. We will continue to invest in the relationships with our existing, critical partners, nurture and develop new relationships and do so globally. We will continue to invest in developing "plus one" solutions and workflows that help tie our software solutions together with those delivered by others.

#### **Key Business Metrics**

In addition to our GAAP financial information, we review several operating and financial metrics, including the following key metrics, to evaluate our business, measure our performance, identify trends affecting our business, formulate business plans, and make strategic decisions.

#### **Number of Devices**

We believe our ability to grow the number of devices on our software platform provides a key indicator of the growth of our business and our future business opportunities. We define a device at the end of any particular period as a device owned by a customer, which device has at least one Jamf product pursuant to an active subscription or support and maintenance agreement or that has a reasonable probability of renewal. We define a customer at the end of any particular period as an entity with at least one active subscription or support and maintenance agreement as of the measurement date or that has a reasonable probability of renewal. A single organization with separate subsidiaries, segments or divisions that use our platform may represent multiple customers as we treat each entity, subsidiary, segment or division that is invoiced separately as a single customer. In cases where customers subscribe to our platform through our channel partners, each end customer is counted separately. A single customer may have multiple Jamf products on a single device, but we still would only count that as one device.

The number of devices was 21.8 million and 16.3 million as of March 31, 2021 and 2020, respectively, representing a 34% year-over-year growth rate. We have seen particular strength in the growth rate of devices as COVID-19 has accelerated the demand for organizations to connect remotely, manage, and protect their Apple devices.

#### **Annual Recurring Revenue**

Annual Recurring Revenue ("ARR") represents the annualized value of all subscription and support and maintenance contracts as of the end of the period. ARR mitigates fluctuations due to seasonality, contract term and the

sales mix of subscriptions for term-based licenses and SaaS. ARR does not have any standardized meaning and is therefore unlikely to be comparable to similarly titled measures presented by other companies. ARR should be viewed independently of revenue and deferred revenue and is not intended to be combined with or to replace either of those items. ARR is not a forecast and the active contracts at the end of a reporting period used in calculating ARR may or may not be extended or renewed by our customers.

Our ARR was \$308.0 million and \$224.9 million as of March 31, 2021 and 2020, respectively, which is an increase of 37% year-over-year. The growth in our ARR is primarily driven by our high device expansion rates, our new logo acquisition and the upselling and cross selling of products into our installed base.

#### **Dollar-Based Net Retention Rate**

To further illustrate the "land and expand" economics of our customer relationships, we examine the rate at which our customers increase their subscriptions for our software solutions. Our dollar-based net retention rate measures our ability to increase revenue across our existing customer base through expanded use of our software solutions, offset by customers whose subscription contracts with us are not renewed or renew at a lower amount.

We calculate dollar-based net retention rate as of a period end by starting with the ARR from the cohort of all customers as of 12 months prior to such period end ("Prior Period ARR"). We then calculate the ARR from these same customers as of the current period end ("Current Period ARR"). Current Period ARR includes any expansion and is net of contraction or attrition over the last 12 months but excludes ARR from new customers in the current period. We then divide the total Current Period ARR by the total Prior Period ARR to arrive at the dollar-based net retention rate.

Our dollar-based net retention rates were 117% and 120% for the trailing twelve months ended March 31, 2021 and 2020, respectively. Our high dollar-based net retention rates are primarily attributable to an expansion of devices. We believe our ability to cross-sell our new solutions to our installed base, particularly Jamf Connect and Jamf Protect, will continue to support our high dollar-based net retention rates.

#### **Non-GAAP Financial Measures**

In addition to our results determined in accordance with GAAP, we believe the non-GAAP measures of Non-GAAP Gross Profit, Non-GAAP Gross Profit Margin, Non-GAAP Operating Income, Non-GAAP Operating Income Margin, Non-GAAP Net Income and Adjusted EBITDA are useful in evaluating our operating performance. We believe that non-GAAP financial information, when taken collectively, may be helpful to investors because it provides consistency and comparability with past financial performance and assists in comparisons with other companies, some of which use similar non-GAAP information to supplement their GAAP results. The non-GAAP financial information is presented for supplemental informational purposes only, and should not be considered a substitute for financial information presented in accordance with GAAP, and may be different from similarly-titled non-GAAP measures used by other companies. A reconciliation is provided below for each non-GAAP financial measure to the most directly comparable financial measure stated in accordance with GAAP. Investors are encouraged to review the related GAAP financial measures and the reconciliation of these non-GAAP financial measures to their most directly comparable GAAP financial measures.

#### Non-GAAP Gross Profit

Non-GAAP Gross Profit and Non-GAAP Gross Profit Margin are supplemental measures of operating performance that are not prepared in accordance with GAAP and that do not represent, and should not be considered as, alternatives to gross profit or gross profit margin, as determined in accordance with GAAP. We define Non-GAAP Gross Profit as gross profit, adjusted for stock-based compensation expense and amortization expense. We define Non-GAAP Gross Profit Margin as Non-GAAP Gross Profit as a percentage of total revenue.

We use Non-GAAP Gross Profit and Non-GAAP Gross Profit Margin to understand and evaluate our core operating performance and trends and to prepare and approve our annual budget. We believe Non-GAAP Gross Profit and Non-GAAP Gross Profit Margin are useful measures to us and to our investors to assist in evaluating our core

operating performance because it provides consistency and direct comparability with our past financial performance and between fiscal periods, as the metric eliminates the effects of variability of stock-based compensation expense and amortization of acquired developed technology, which are non-cash expenses that may fluctuate for reasons unrelated to overall operating performance. While the amortization expense of acquired developed technology is excluded from Non-GAAP Gross Profit, the revenue related to acquired developed technology is reflected in Non-GAAP Gross Profit as these assets contribute to our revenue generation.

Non-GAAP Gross Profit and Non-GAAP Gross Profit Margin have limitations as analytical tools, and you should not consider them in isolation, or as substitutes for analysis of our results as reported under GAAP. Because of these limitations, Non-GAAP Gross Profit and Non-GAAP Gross Profit Margin should not be considered as replacements for gross profit or gross profit margin, as determined by GAAP, or as measures of our profitability. We compensate for these limitations by relying primarily on our GAAP results and using non-GAAP measures only for supplemental purposes.

A reconciliation of Non-GAAP Gross Profit to gross profit, the most directly comparable GAAP measure, is as follows:

	Three Months Ended March 31,				
		2021	—	2020	
Cross profit	\$	63.912	(in thousands)		
Gross profit	Ф	,-	Ф	45,379	
Amortization expense		2,777		2,677	
Stock-based compensation		401		38	
Non-GAAP Gross Profit	\$	67,090	\$	48,094	
Non-GAAP Gross Profit Margin		83 9	%	80 %	

#### **Non-GAAP Operating Income**

Non-GAAP Operating Income and Non-GAAP Operating Income Margin are supplemental measures of operating performance that are not prepared in accordance with GAAP and that do not represent, and should not be considered as, alternatives to operating loss or operating loss margin, as determined in accordance with GAAP. We define Non-GAAP Operating Income as operating loss, adjusted for amortization, stock-based compensation, acquisition-related expense, acquisition-related earnout, payroll taxes related to stock-based compensation and costs associated with our secondary offerings. In the first quarter of 2021, we began excluding payroll taxes related to stock-based compensation from our non-GAAP measures as these expenses are tied to the exercise or vesting of underlying equity awards and the price of our common stock at the time of vesting or exercise. As a result, these taxes may vary in any particular period independent of the financial and operating performance of our business. Payroll taxes related to stock-based compensation were not material prior to the first quarter of 2021. We define Non-GAAP Operating Income Margin as Non-GAAP Operating Income as a percentage of total revenue.

We use Non-GAAP Operating Income and Non-GAAP Operating Income Margin to understand and evaluate our core operating performance and trends, to prepare and approve our annual budget, and to develop short-term and long-term operating plans. We believe that Non-GAAP Operating Income and Non-GAAP Operating Income Margin facilitate comparison of our operating performance on a consistent basis between periods, and when viewed in combination with our results prepared in accordance with GAAP, help provide a broader picture of factors and trends affecting our results of operations. While the amortization expense of acquired trademarks, customer relationships, and developed technology is excluded from Non-GAAP Operating Income, the revenue related to acquired trademarks, customer relationships, and developed technology is reflected in Non-GAAP Operating Income as these assets contribute to our revenue generation.

Non-GAAP Operating Income and Non-GAAP Operating Income Margin have limitations as analytical tools, and you should not consider them in isolation, or as substitutes for analysis of our results as reported under GAAP. Because of these limitations, Non-GAAP Operating Income and Non-GAAP Operating Income Margin should not be considered as replacements for operating loss or operating loss margin, as determined by GAAP, or as measures of our

profitability. We compensate for these limitations by relying primarily on our GAAP results and using non-GAAP measures only for supplemental purposes.

A reconciliation of Non-GAAP Operating Income to operating loss, the most directly comparable GAAP measure, is as follows:

	Three Months Ended March 31,			
		2021		2020
		(in thousands)		
Operating loss	\$	(2,778)	\$	(6,483)
Amortization expense		8,404		8,351
Stock-based compensation		2,832		811
Acquisition-related expense		110		1,600
Acquisition-related earnout		300		_
Payroll taxes related to stock-based compensation		395		_
Non-GAAP Operating Income	\$	9,263	\$	4,279
Non-GAAP Operating Income Margin		11 %	ó	7 %

#### Non-GAAP Net Income (Loss)

Non-GAAP Net Income (Loss) is a supplemental measure of operating performance that is not prepared in accordance with GAAP and that does not represent, and should not be considered as, an alternative to net loss, as determined in accordance with GAAP. We define Non-GAAP Net Income (Loss) as net loss, adjusted for amortization, stock-based compensation, foreign currency transaction loss, loss on extinguishment of debt, acquisition-related expense, acquisition-related earnout, costs associated with our secondary offerings, payroll taxes related to stock-based compensation, discrete tax items and provision (benefit) for income taxes.

We use Non-GAAP Net Income (Loss) to understand and evaluate our core operating performance and trends, to prepare and approve our annual budget, and to develop short-term and long-term operating plans. We believe that Non-GAAP Net Income (Loss) facilitates comparison of our operating performance on a consistent basis between periods, and when viewed in combination with our results prepared in accordance with GAAP, helps provide a broader picture of factors and trends affecting our results of operations. While the amortization expense of acquired trademarks, customer relationships, and developed technology is excluded from Non-GAAP Net Income (Loss), the revenue related to acquired trademarks, customer relationships, and developed technology is reflected in Non-GAAP Net Income (Loss) as these assets contribute to our revenue generation.

Non-GAAP Net Income (Loss) has limitations as an analytical tool, and you should not consider it in isolation, or as a substitute for analysis of our results as reported under GAAP. Because of these limitations, Non-GAAP Net Income (Loss) should not be considered as a replacement for net loss, as determined by GAAP, or as a measure of our profitability. We compensate for these limitations by relying primarily on our GAAP results and using non-GAAP measures only for supplemental purposes.

A reconciliation of Non-GAAP Net Income (Loss) to net loss, the most directly comparable GAAP measure, is as follows:

	Thi	Three Months Ended March 31,			
		2021		2020	
		(in th	ousan	ds)	
Net loss	\$	(3,069)	\$	(8,290)	
Amortization expense		8,404		8,351	
Stock-based compensation		2,832		811	
Foreign currency transaction loss		171		304	
Acquisition-related expense		110		1,600	
Acquisition-related earnout		300		_	
Payroll taxes related to stock-based compensation		395		_	
Discrete tax items		49		(318)	
Provision (benefit) for income taxes <sup>(1)</sup>		66		(2,703)	
Non-GAAP Net Income (Loss)	\$	9,258	\$	(245)	

(1) In the first quarter of 2020, the related tax effects of the adjustments to Non-GAAP Net Income (Loss) were calculated using the respective statutory tax rate for applicable jurisdictions, which was not materially different from our annual effective tax rate for full year 2020 of approximately 25%. In the first quarter of 2021, our annual effective tax rate was impacted by changes in the domestic valuation allowance. Therefore, we used the annual effective tax rate of (0.5)% in the first quarter of 2021 as this rate was materially different than our statutory rate.

#### Adjusted EBITDA

Adjusted EBITDA is a supplemental measure of operating performance that is not prepared in accordance with GAAP and that does not represent, and should not be considered as, an alternative to net loss, as determined in accordance with GAAP. We define Adjusted EBITDA as net loss, adjusted for interest expense, net, provision (benefit) for income taxes, depreciation and amortization, stock-based compensation, foreign currency transaction loss, loss on extinguishment of debt, acquisition-related expense, acquisition-related earnout, costs associated with our secondary offerings and payroll taxes related to stock-based compensation.

We use Adjusted EBITDA to understand and evaluate our core operating performance and trends, to prepare and approve our annual budget, and to develop short-term and long-term operating plans. We believe that Adjusted EBITDA facilitates comparison of our operating performance on a consistent basis between periods, and when viewed in combination with our results prepared in accordance with GAAP, helps provide a broader picture of factors and trends affecting our results of operations.

Adjusted EBITDA has limitations as an analytical tool, and you should not consider it in isolation, or as a substitute for analysis of our results as reported under GAAP. Because of these limitations, Adjusted EBITDA should not be considered as a replacement for net loss, as determined by GAAP, or as a measure of our profitability. We compensate for these limitations by relying primarily on our GAAP results and using non-GAAP measures only for supplemental purposes.

A reconciliation of Adjusted EBITDA to net loss, the most directly comparable GAAP measure, is as follows:

	Three Months Ended March 31,			March 31,
		2021		2020
		(in th	ousan	ds)
Net loss	\$	(3,069)	\$	(8,290)
Interest expense, net		55		4,778
Provision (benefit) for income taxes		65		(3,220)
Depreciation expense		1,380		1,235
Amortization expense		8,404		8,351
Stock-based compensation		2,832		811
Foreign currency transaction loss		171		304
Acquisition-related expense		110		1,600
Acquisition-related earnout		300		_
Payroll taxes related to stock-based compensation		395		_
Adjusted EBITDA	\$	10,643	\$	5,569

#### **Components of Results of Operations**

#### Revenues

We recognize revenue under ASC 606 when or as performance obligations are satisfied. We derive revenue primarily from sales of SaaS subscriptions and support and maintenance contracts, and to a lesser extent, sales of onpremise subscriptions and perpetual licenses and services.

Subscription. Subscription revenue consists of sales of SaaS subscriptions and support and maintenance contracts. We sell our software solutions primarily with a one-year contract term. We typically invoice SaaS subscription fees and support and maintenance fees annually in advance and recognize revenue ratably over the term of the applicable agreement, provided that all other revenue recognition criteria have been satisfied. In the fourth quarter of 2020, we reclassified the license portion of on-premise subscription revenue from license revenue to subscription revenue in the consolidated statements of operations on a retroactive basis. See additional information in "Note 1 — Basis of Presentation and Description of Business" to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report on Form 10-Q. The license portion of on-premise subscription revenue is recognized upfront, assuming all revenue recognition criteria are satisfied. See "Management's Discussion and Analysis of Financial Condition and Results of Operations — Critical Accounting Policies" in our Annual Report on Form 10-K for more information. We expect subscription revenue to increase over time as we expand our customer base because sales to new customers are expected to be primarily SaaS subscriptions.

*License*. License revenue consists of revenue from on-premise perpetual licenses of our Jamf Pro product sold primarily to existing customers. We recognize license revenue upfront, assuming all revenue recognition criteria are satisfied. We expect license revenue to decrease because sales to new customers are primarily cloud-based subscription arrangements and therefore reflected in subscription revenue.

*Services*. Services revenues consist primarily of professional services provided to our customers to configure and optimize the use of our software solutions, as well as training services related to the operation of our software solutions. Our services are priced on a fixed fee basis and generally invoiced in advance of the service being delivered. Revenue is recognized as the services are performed. We expect services revenues to decrease as a percentage of total revenue as the demand for our services is not expected to grow at the same rate as the demand for our subscription solutions.

#### Cost of Revenues

*Cost of subscription*. Cost of subscription revenue consists primarily of employee compensation costs for employees associated with supporting our subscription and support and maintenance arrangements, our customer success

function, and third-party hosting fees related to our cloud services. Employee compensation and related costs include cash compensation and benefits to employees and associated overhead costs. We expect cost of subscription revenue to increase in absolute dollars, but to remain relatively consistent as a percentage of subscription revenue, relative to the extent of the growth of our business.

Cost of services. Cost of services revenue consists primarily of employee compensation costs directly associated with delivery of professional services and training, costs of third-party integrators and other associated overhead costs. We expect cost of services revenue to decrease in absolute dollars relative to the decrease of our services business.

#### **Gross Profit**

Gross profit, or revenue less cost of revenue, has been and will continue to be affected by various factors, including the mix of cloud-based subscription customers, the costs associated with supporting our cloud solution, the extent to which we expand our customer support team and the extent to which we can increase the efficiency of our technology and infrastructure though technological improvements. We expect our gross profit to increase in absolute dollars.

#### **Operating Expenses**

Sales and Marketing. Sales and marketing expenses consist primarily of employee compensation costs, sales commissions, costs of general marketing and promotional activities, travel-related expenses and allocated overhead. Sales commissions earned by our sales force are deferred and amortized over the period of benefit, which is estimated to be 5 years. We expect our sales and marketing expenses to increase on an absolute dollar basis as we expand our sales personnel and marketing efforts.

Research and development. Research and development expenses consist primarily of personnel costs and allocated overhead. We will continue to invest in innovation so that we can offer our customers new solutions and enhance our existing solutions. See "Business — Research and Development" in our Annual Report on Form 10-K for more information. We expect such investment to increase on an absolute dollar basis as our business grows.

General and Administrative. General and administrative expenses consist primarily of employee compensation costs for corporate personnel, such as those in our executive, human resource, facilities, accounting and finance, legal and compliance, and information technology departments. In addition, general and administrative expenses include acquisition-related expenses which primarily consist of third-party expenses, such as legal and accounting fees, and adjustments to contingent consideration. General and administrative expenses also include costs incurred in secondary offerings. We expect our general and administrative expenses to increase on a dollar basis as our business grows, particularly as we continue to invest in technology infrastructure and expand our operations globally. Also, we incur additional general and administrative expenses as a result of operating as a public company, including costs to comply with the rules and regulations applicable to companies listed on a national securities exchange, costs related to compliance and reporting obligations pursuant to the rules and regulations of the SEC, and increased expenses for insurance, investor relations and accounting expenses.

*Amortization*. Amortization expense primarily consists of amortization of acquired trademarks, customer relationships and developed technology.

### Interest Expense, Net

Interest expense, net consists primarily of interest payments on our outstanding borrowings under our credit facilities as well as the amortization of associated deferred financing costs. See "Liquidity and Capital Resources — Credit Facilities."

#### Foreign Currency Transaction Gain (Loss)

Our reporting currency is the U.S. dollar. The functional currency of all our international operations is the U.S. dollar. The assets, liabilities, revenues and expenses of our foreign operations are remeasured in accordance with ASC Topic 830, *Foreign Currency Matters*. Remeasurement adjustments are recorded as foreign currency transaction gains (losses) in the consolidated statement of operations.

#### Income Tax (Provision) Benefit

Income tax (provision) benefit consists primarily of income taxes related to U.S. federal and state income taxes and income taxes in foreign jurisdictions in which we conduct business.

#### Other Income

Other income consists primarily of sublease rental income. The sublease was terminated in the second quarter of 2020.

### **Results of Operations**

The following table sets forth our consolidated statements of operations data for the periods indicated:

	Three Months Ended Marc 2021 2020 (in thousands)			2020
Consolidated Statement of Operations Data:		(in tho	usano	is)
Revenue:				
Subscription	\$	74,923	\$	54,618
Services		4,003		4,010
License		2,242		1,762
Total revenue		81,168		60,390
Cost of revenue:				
Cost of subscription <sup>(1)(2)</sup> (exclusive of amortization expense shown below)		12,014		9,248
Cost of services <sup>(1)(2)</sup> (exclusive of amortization expense shown below)		2,465		3,086
Amortization expense		2,777		2,677
Total cost of revenue		17,256		15,011
Gross profit		63,912		45,379
Operating expenses:				
Sales and marketing <sup>(1)(2)</sup>		29,332		22,282
Research and development <sup>(1)(2)</sup>		15,626		12,617
General and administrative <sup>(1)(2)(3)</sup>		16,105		11,289
Amortization expense		5,627		5,674
Total operating expenses		66,690		51,862
Loss from operations		(2,778)		(6,483)
Interest expense, net		(55)		(4,778)
Foreign currency transaction loss		(171)		(304)
Other income, net				55
Loss before income tax (provision) benefit		(3,004)		(11,510)
Income tax (provision) benefit		(65)		3,220
Net loss	\$	(3,069)	\$	(8,290)

### (1) Includes stock-based compensation as follows:

	 Three Months Ended Mare 2021 2022 (in thousands)		
Cost of revenue:	,	Í	
Subscription	\$ 324	\$	38
Services	77		_
Sales and marketing	842		111
Research and development	778		157
General and administrative	811		505
	\$ 2,832	\$	811

### (2) Includes depreciation expense as follows:

	 2021	Ended March 31, 2020 ousands)	
Cost of revenue:		,	
Subscription	\$ 263	\$	238
Services	43		53
Sales and marketing	574		494
Research and development	305		292
General and administrative	195		156
	\$ 1,380	\$	1,233

### (3) Includes acquisition-related expense as follows:

	Three Months I	Ended March 31,	
	2021	2020	
	(in tho	ousands)	
rative	\$ 110	\$ 1,600	

General and administrative also includes a Digita earnout expense of \$0.3 million for the three months ended March 31, 2021.

The following table sets forth our consolidated statements of operations data expressed as a percentage of total revenue for the periods indicated:

	Three Months Ended N	
	2021 (as a percentage of tota	2020 Lrevenue)
Consolidated Statement of Operations Data:	(== = F == =====	
Revenue:		
Subscription	92 %	90 %
Services	5	7
License	3	3
Total revenue	100	100
Cost of revenue:		
Cost of subscription (exclusive of amortization expense shown below)	15	15
Cost of services (exclusive of amortization expense shown below)	3	5
Amortization expense	3	5
Total cost of revenue	21	25
Gross profit	79	75
Operating expenses:		
Sales and marketing	36	37
Research and development	19	21
General and administrative	20	19
Amortization expense	7	9
Total operating expenses	82	86
Loss from operations	(3)	(11)
Interest expense, net	_	(7)
Foreign currency transaction loss	(1)	(1)
Other income, net	<u> </u>	<u> </u>
Loss before income tax (provision) benefit	(4)	(19)
Income tax (provision) benefit	<u> </u>	5
Net loss	(4)%	(14)%

### Comparison of the Three Months Ended March 31, 2021 and 2020

#### Revenue

	Three Months Ended March 31,			Change		
		2021		2020	\$	%
		(in the	ousano	ls, except pe	rcentages)	
SaaS subscription and support and maintenance	\$	66,669	\$	50,078	\$ 16,591	33 %
On-premise subscription		8,254		4,540	3,714	82
Subscription revenue		74,923		54,618	20,305	37
Professional services		4,003		4,010	(7)	(0)
Perpetual licenses		2,242		1,762	480	27
Non-subscription revenue		6,245		5,772	473	8
Total revenue	\$	81,168	\$	60,390	\$ 20,778	34 %

Total revenue increased by \$20.8 million, or 34%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. Overall revenue increased as a result of higher subscription revenue and license revenue. Subscription revenue accounted for 92% of total revenue for the three months ended March 31, 2021 compared to 90% for the three months ended March 31, 2020. The increase in subscription revenue was driven by device expansion, the addition of new customers and cross-selling. License revenue increased due to additional licenses for an existing customer, partially offset by shifting customers to our SaaS model as opposed to on-premise perpetual licenses.

#### Cost of Revenue and Gross Margin

	Three Months Ended March 31,			Chang	Change		
				2020	\$	%	
C+ -f	(in thousands, except percen			rcentages)			
Cost of revenue:							
Cost of subscription (exclusive of amortization expense shown							
below)	\$	12,014	\$	9,248	\$ 2,766	30 %	
Cost of services (exclusive of amortization expense show below)		2,465		3,086	(621)	(20)	
Amortization expense		2,777		2,677	100	4	
Total cost of revenue	\$	17,256	\$	15,011	\$ 2,245	15 %	
Gross margin		79 %	<u> </u>	75 %			

Cost of revenue increased by \$2.2 million, or 15%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020 driven by an increase in cost of subscription revenue, partially offset by lower cost of services revenue. Cost of subscription revenue increased \$2.8 million, or 30%, primarily due to an increase of \$1.5 million in employee compensation costs related to higher headcount to support the growth in our subscription customer base, an increase of \$1.2 million in third party hosting fees as we increased capacity to support our growth and a \$0.3 million increase in stock-based compensation expense. Cost of services revenue decreased \$0.6 million primarily due to lower travel-related costs and costs of third-party integrators.

Total gross margin was 79% and 75% for the three months ended March 31, 2021 and 2020, respectively, as our revenue expanded faster than the costs required to deliver the revenue.

#### **Operating Expenses**

	Three Months Ended March 31,				Change		
	2021		2020			\$	%
	(in thousands, except percentages			tages)			
Operating expenses:							
Sales and marketing		29,332	\$	22,282	\$	7,050	32 %
Research and development		15,626		12,617		3,009	24
General and administrative		16,105		11,289		4,816	43
Amortization expense		5,627		5,674		(47)	(1)
Operating expenses	\$	66,690	\$	51,862	\$	14,828	29 %

Sales and Marketing. Sales and marketing expenses increased by \$7.1 million, or 32%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. The increase was primarily due to an increase of \$5.5 million in employee compensation costs driven by headcount growth, a \$1.4 million increase in marketing costs, an increase of \$0.3 million in computer hardware and software costs to support the growth of the business and a \$0.7 million increase in stock-based compensation expense reflecting the IPO grant, partially offset by a \$1.2 million decrease in travel-related expenses reflecting less travel due to COVID-19. Marketing costs increased primarily due to increases in demand generation programs, advertising, and brand awareness campaigns focused on new customer acquisition.

Research and Development. Research and development expenses increased by \$3.0 million, or 24%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. The increase was primarily due to an increase of \$2.1 million in employee compensation costs due to higher headcount, an increase of \$0.7 million in outside services and a \$0.6 million increase in stock-based compensation expense reflecting the IPO grant, partially offset by a \$0.4 million decrease in travel-related expenses reflecting less travel due to COVID-19.

*General and Administrative.* General and administrative expenses increased by \$4.8 million, or 43%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. The increase was primarily

due to an increase of \$2.8 million in employee compensation costs driven by higher headcount to support our continued growth, \$2.0 million in additional expenses as a result of operating as a public company, an increase of \$0.9 million in computer hardware and software costs to support the growth of the business, a \$0.3 million increase in stock-based compensation expense and a \$0.3 million increase to contingent consideration, partially offset by a \$1.5 million decrease in acquisition-related expenses and a \$1.1 million decrease in travel-related expenses reflecting less travel due to COVID-19. The remainder of the cost increase is primarily related to costs to support the growth in business and headcount.

#### Interest Expense, Net

Thre	e Months l	Ended 1	March 31,	Chang	e	
	2021		2020	\$	%	
	(in th	ousand	ls, except pe	ercentages)		
\$	55	\$	4,778	\$ (4,723)	(99)%	

Interest expense, net decreased by \$4.7 million, or 99%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020 reflecting the repayment of the Prior Term Loan Facility.

#### **Foreign Currency Transaction Loss**

		Three Month	s Ended	March 31,	Chan	ge
	•	2021		2020	\$	%
		(in t	housand	s, except per	rcentages)	
Foreign currency transaction loss		\$ 171	. \$	304	\$ (133)	(44)%

Foreign currency transaction loss decreased by \$0.1 million, or 44%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. The decrease in the loss was driven by a favorable impact from the remeasurement of monetary assets and liabilities denominated in Euros and Australian dollars compared to the prior year period.

#### Other Income, Net

	Three M	Three Months Ended March 31,			Cha	nge	
	202	1	20	020	\$	%	
	·	(in thou	sands, e	xcept per	centages)		
et	\$	_	\$	55	\$ (55)	(100)%	

Other income, net decreased by \$0.1 million, or 100%, for the three months ended March 31, 2021 compared to the three months ended March 31, 2020. The decrease in Other income, net was due to the termination of our sublease in the second quarter of 2020.

#### Income Tax (Provision) Benefit

	Three Mo	onths E	inded 1	March 31,	Change		
	2021			2020	\$	%	
	·	(in the	ousand	s, except pe	rcentages)		
rovision) benefit	\$	(65)	\$	3,220	\$ (3,285)	NM	

Income tax (provision) benefit was \$(0.1) million and \$3.2 million for the three months ended March 31, 2021 and 2020, respectively. The effective tax rates for the three months ended March 31, 2021 and 2020 were (2.2)% and 28.0%, respectively. The effective tax rate for the three months ended March 31, 2021 was lower than the prior year period due to Section 162(m) of the Internal Revenue Code, stock option activity and the domestic valuation allowance.

The Company's annual effective tax rate for the three months ended March 31, 2021 and 2020 was (0.5)% and 25.2%, respectively.

#### **Liquidity and Capital Resources**

#### General

As of March 31, 2021, our principal sources of liquidity were cash and cash equivalents totaling \$196.2 million, which were held for working capital purposes, as well as the available balance of our New Revolving Credit Facility, described in Note 7 to our Consolidated Financial Statements. Our cash equivalents are comprised of money market funds and/or U.S. Treasuries with original or remaining maturities at the time of purchase of three months or less. Our positive cash flows from operations enable us to make continued investments in supporting the growth of our business. We expect that our operating cash flows, in addition to our cash and cash equivalents, will enable us to continue to make such investments in the future.

On May 5, 2021, we entered into a commitment letter (the "Commitment Letter") with JPMorgan Chase Bank, N.A. and other commitment parties thereto (the "Commitment Parties"), pursuant to which, subject to the terms and conditions set forth therein, the Commitment Parties have committed to provide a 364-day term loan facility in an aggregate principal amount of up to \$250.0 million, on substantially the same terms and conditions as the New Credit Agreement. The funding of the term loan facility provided for by the Commitment Letter is contingent on the satisfaction of customary conditions, including the execution and delivery of definitive documentation with respect to the term loan facility in accordance with the terms set forth in the Commitment Letter.

We believe our cash and cash equivalents, our New Revolving Credit Facility and cash provided by sales of our software solutions and services will be sufficient to meet our working capital and capital expenditure needs for at least the next 12 months. Our future capital requirements will depend on many factors including our growth rate, the timing and extent of spending to support development efforts, the expansion of sales and marketing activities, the introduction of new and enhanced products and services offerings, and the continuing market acceptance of our products. In the future, we may use cash to acquire or invest in complementary businesses, services and technologies, including intellectual property rights.

A majority of our customers pay in advance for subscriptions and support and maintenance contracts, a portion of which is recorded as deferred revenue. Deferred revenue consists of the unearned portion of billed fees for our subscriptions, which is later recognized as revenue in accordance with our revenue recognition policy. As of March 31, 2021, we had deferred revenue of \$221.6 million, of which \$167.9 million was recorded as a current liability and is expected to be recorded as revenue in the next 12 months, provided all other revenue recognition criteria have been met.

#### Cash Flows

The following table presents a summary of our consolidated cash flows from operating, investing and financing activities:

	Three Months Ended March 31,			
		2021		2020
		(in tho	usands)	
Net cash provided by (used in) operating activities	\$	4,023	\$	(7,355)
Net cash used in investing activities		(6,319)		(1,039)
Net cash provided by (used in) financing activities		4,019		(1,362)
Effect of exchange rate changes on cash and cash equivalents		(401)		_
Net increase (decrease) in cash and cash equivalents		1,322		(9,756)
Cash and cash equivalents at beginning of period		194,868		32,433
Cash and cash equivalents at end of period	\$	196,190	\$	22,677
Cash paid for interest	\$	3	\$	4,734
Cash paid for purchases of equipment and leasehold improvements		3,290		1,039

#### **Operating Activities**

Our largest source of operating cash is cash collections from our customers for subscriptions. Our primary use of cash from operating activities is for employee-related expenditures, marketing expenses and third-party hosting costs.

For the three months ended March 31, 2021, net cash provided by operating activities was \$4.0 million reflecting our net loss of \$3.1 million, adjusted for non-cash charges of \$17.3 million and net cash outflows of \$10.2 million from changes in our operating assets and liabilities. Non-cash charges primarily consisted of depreciation and amortization of property and equipment and intangible assets, amortization of deferred contract costs, non-cash lease expense, share-based compensation and a \$0.3 million adjustment to our Digita earnout, partially offset by the deferred tax benefit. The primary drivers of net cash outflows from changes in operating assets and liabilities included a \$7.1 million increase in trade accounts receivable due to higher sales and the timing of cash receipts from our customers, a \$6.5 million increase in deferred contract costs due to an increase in capitalized contract costs, a \$3.3 million increase in prepaid expenses and other assets and an \$8.9 million decrease in accounts payable and accrued liabilities due to the timing of cash disbursements. These changes were partially offset by a \$15.5 million increase in deferred revenue due to the upfront billing for a majority of our subscriptions.

For the three months ended March 31, 2020, net cash used in operating activities was \$7.4 million reflecting our net loss of \$8.3 million, adjusted for non-cash charges of \$10.5 million and net cash outflows of \$9.5 million from changes in our operating assets and liabilities. Non-cash charges primarily consisted of depreciation and amortization of property and equipment and intangible assets, amortization of deferred contract costs, amortization of debt issuance costs and share-based compensation, partially offset by deferred taxes. The primary drivers of net cash outflows from changes in operating assets and liabilities included an increase in deferred contract costs of \$4.8 million, an increase in prepaid expenses and other assets of \$2.3 million and a decrease in accounts payable and accrued liabilities of \$6.9 million, partially offset by a \$5.0 million increase in deferred revenue.

#### **Investing Activities**

During the three months ended March 31, 2021, net cash used in investing activities was \$6.3 million driven by purchases of \$3.3 million in equipment and leasehold improvements primarily reflecting updates to office space and hardware and software and the acquisition of cmdReporter of \$3.0 million.

During the three months ended March 31, 2020, net cash used in investing activities was \$1.0 million driven by purchases of equipment and leasehold improvements to support our higher headcount with additional office space and hardware and software.

#### Financing Activities

Net cash provided by financing activities of \$4.0 million during the three months ended March 31, 2021 was due to proceeds from the exercise of stock options.

Net cash used in financing activities of \$1.4 million during the three months ended March 31, 2020 was due to the payment of offering costs of \$1.5 million, partially offset by \$0.1 million of proceeds from the exercise of stock options.

#### **Contractual Obligations and Commitments**

As of March 31, 2021, our principal commitments consist of obligations under operating leases for office space. In "Management's Discussion and Analysis of Financial Conditions and Results of Operations" included in our Annual Report on Form 10-K, we disclosed our total contractual obligations as of December 31, 2020. Outside of the above and routine transactions made in the ordinary course of business, there have been no material changes to the contractual obligations as disclosed in our Annual Report on Form 10-K.

#### **Indemnification Agreements**

In the ordinary course of business, we enter into agreements of varying scope and terms pursuant to which we agree to indemnify customers, channel partners, vendors, lessors, business partners and other parties with respect to certain matters, including, but not limited to, losses arising out of the breach of such agreements, services to be provided by us or from intellectual property infringement, misappropriation or other violation claims made by third parties. See "Risk Factors — We have indemnity provisions under our contracts with our customers, channel partners and other third parties, which could have a material adverse effect on our business" in our Annual Report on Form 10-K. In addition, we have entered into indemnification agreements with our directors and certain executive officers that will require us, among other things, to indemnify them against certain liabilities that may arise by reason of their status or service as directors, officers or employees. No demands have been made upon us to provide indemnification under such agreements and there are no claims that we are aware of that could have a material effect on our consolidated balance sheets, consolidated statements of operations and comprehensive loss, or consolidated statements of cash flows.

#### JOBS Act

We qualify as an "emerging growth company" pursuant to the provisions of the JOBS Act. For as long as we are an "emerging growth company," we may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not "emerging growth companies," including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404(b) of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements, exemptions from the requirements of holding advisory "say-on-pay" votes on executive compensation and shareholder advisory votes on golden parachute compensation.

The JOBS Act also permits an emerging growth company like us to take advantage of an extended transition period to comply with new or revised accounting standards applicable to public companies. We have elected to use the extended transition period for complying with new or revised accounting standards and, therefore, we will not be subject to the same new or revised accounting standards as other public companies that comply with such new or revised accounting standards on a non-delayed basis.

#### **Critical Accounting Policies**

Our discussion and analysis of financial condition and results of operations are based upon our consolidated financial statements. The preparation of our financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses. We base our estimates on experience and other assumptions that we believe are reasonable under the circumstances, and we evaluate these estimates on an ongoing basis. Actual results may differ from those estimates, impacting our reported results of operations and financial condition.

Except for the accounting policies for leases that were updated as a result of adopting the new accounting standard, there have been no material changes to our critical accounting policies and estimates disclosed in our Annual Report on Form 10-K. For more information, refer to "Note 2 — Summary of Significant Accounting Policies" to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report on Form 10-Q.

#### **Recent Accounting Pronouncements**

For a description of our recently adopted accounting pronouncements and recently issued accounting standards not yet adopted, see "Note 2 — Summary of Significant Accounting Policies" to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report on Form 10-Q.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk

Market risk represents the risk of loss that may impact our financial position due to adverse changes in financial market prices and rates. Our market risk exposure is primarily a result of exposure due to potential changes in inflation or interest rates. We do not hold financial instruments for trading purposes.

#### Foreign Currency Exchange Risk

The functional currency of our foreign subsidiaries is the U.S. dollar. Most of our sales are denominated in U.S. dollars, and therefore our revenue is not currently subject to significant foreign currency risk. Our operating expenses are denominated in the currencies of the countries in which our operations are located, which are primarily in the U.S., Poland, and the Netherlands. Our consolidated results of operations and cash flows are, therefore, subject to fluctuations due to changes in foreign currency exchange rates and may be adversely affected in the future due to changes in foreign exchange rates. To date, we have not entered into any hedging arrangements with respect to foreign currency risk or other derivative financial instruments. During the three months ended March 31, 2021 and 2020, a hypothetical 10% change in foreign currency exchange rates applicable to our business would not have had a material impact on our consolidated financial statements.

#### Impact of Inflation

While inflation may impact our net revenue and costs of revenue, we believe the effects of inflation, if any, on our results of operations and financial condition have not been significant. However, there can be no assurance that our results of operations and financial condition will not be materially impacted by inflation in the future.

# **Item 4. Controls and Procedures**

#### **Evaluation of Disclosure Controls and Procedures**

We maintain "disclosure controls and procedures," as defined in Rule 13a–15(e) and Rule 15d–15(e) under the Exchange Act, that are designed to provide reasonable assurance that information required to be disclosed by the Company in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to provide reasonable assurance that information required to be disclosed by the Company in the reports that we file or submit under the Exchange Act is accumulated and communicated to our management, including our principal executive and principal financial officers, as appropriate to allow timely decisions regarding required disclosure. Based on such evaluation, our principal executive officer and principal financial officer have concluded that as of March 31, 2021, our disclosure controls and procedures were effective at the reasonable assurance level.

# **Changes in Internal Control**

There have been no changes in internal control over financial reporting during the quarter ended March 31, 2021 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### PART II

#### OTHER INFORMATION

#### **Item 1. Legal Proceedings**

We are not presently a party to any litigation the outcome of which, we believe, if determined adversely to us, would individually or taken together have a material adverse effect on our business, operating results, cash flows or financial condition.

#### **Item 1A. Risk Factors**

This quarterly report should be read in conjunction with the risk factors included in our Annual Report on Form 10-K. Except for the risk factors set forth below that are new or contain changes to the similarly titled risk factors included in our Annual Report on Form 10-K, there have been no material changes to the risk factors disclosed in Part 1, Item 1A "Risk Factors" in our Annual Report on Form 10-K.

#### Risks Related to the Proposed Acquisition of Wandera

Our ability to complete the acquisition of Wandera is subject to various closing conditions, approvals from governmental authorities, which may impose conditions that could adversely affect us or cause the acquisition not to be completed.

On May 11, 2021, we entered into the Merger Agreement for the acquisition of Wandera, a leader in zero trust cloud security and access for mobile devices. The acquisition is subject to a number of customary conditions to closing as specified in the Merger Agreement. These closing conditions include, among others, the receipt of required approvals under U.S. competition laws, and the absence of governmental or other legal restraints or prohibitions preventing the consummation of the acquisition. No assurance can be given that the required regulatory approvals will be obtained or that the required conditions to closing will be satisfied, and, if all required consents and approvals are obtained and the required conditions are satisfied, no assurance can be given as to the terms, conditions and timing of such consents and approvals. Any delay in completing the acquisition could cause the combined company not to realize, or to be delayed in realizing, some or all of the benefits that we and Wandera expect to achieve if the acquisition is successfully completed within its expected time frame. Additionally, either we or Wandera may terminate the Merger Agreement under certain circumstances, including, among other reasons, if the acquisition is not completed by August 3, 2021.

We can provide no assurance that the various closing conditions will be satisfied and that the necessary approvals will be obtained, or that any required conditions will not materially adversely affect the combined company following the acquisition. In addition, we can provide no assurance that these conditions will not result in the abandonment or delay of the acquisition. The occurrence of any of these events individually or in combination could have a material adverse effect on our results of operations and the trading price of our common stock.

#### The termination of the Merger Agreement could negatively impact our business.

If the acquisition is not completed for any reason, our ongoing business may be adversely affected and, without realizing any of the expected benefits of having completed the acquisition, we would be subject to a number of risks, including the following:

- we may experience negative reactions from the financial markets, including negative impacts on our stock price;
- · we may experience negative reactions from our customers, suppliers, distributors and employees;
- we will be required to pay our costs relating to the acquisition, such as legal, advisory, and accounting costs and associated fees and expenses, whether or not the acquisition is completed;

- · we could be subject to time-consuming and costly litigation related to the acquisition; and
- matters relating to the acquisition (including integration planning) require substantial commitments of time
  and resources by our management, which could otherwise have been devoted to day-to-day operations or to
  other opportunities that may have been beneficial to our business.

Whether or not the acquisition is completed, the announcement and pendency of the acquisition could cause disruptions in our and Wandera's business, which could have an adverse effect on our business and financial results.

Whether or not the acquisition is completed, the announcement and pendency of the acquisition could cause disruptions in our and Wandera's business. Specifically:

- disruptions to ongoing operations;
- retaining and motivating key Wandera personnel;
- declining employee morale and retention issues affecting Wandera employees, which may result from changes in management, reporting relationships, or other changes to employment relationships;
- maintaining good relationships with customers or business partners of Wandera or our own customers as a
  result of the acquisition announcement, any uncertainty regarding completion of the acquisition, and any
  integration of operations; and
- the attention of our and Wandera's management may be directed toward the completion of the acquisition.

We have diverted significant management resources in an effort to complete the acquisition. If the acquisition is not completed, we will have incurred significant costs, including the diversion of management resources, for which we will have received little or no benefit.

We will incur significant acquisition-related costs in connection with the Wandera acquisition, and we could incur substantial expenses related to the integration of Wandera.

We have incurred and expect to incur a number of non-recurring costs associated with combining the operations of the two companies, as well as transaction fees and other costs related to the acquisition. These costs and expenses include fees paid to financial, legal and accounting advisors, potential employment-related costs, filing fees, and other related charges. We will need to pay some of these costs regardless of whether the acquisition is completed.

We will also incur integration costs in connection with the acquisition. There are a large number of processes, policies, procedures, operations, technologies and systems that must be integrated in connection with the acquisition and the integration of Wandera's business. Although we expect that the elimination of duplicative costs, strategic benefits, and additional income, as well as the realization of other efficiencies related to the integration of the businesses, may offset incremental transaction and acquisition-related costs over time, any net benefit may not be achieved in the near term or at all. We will bear many of these costs even if the acquisition is not completed. While we have assumed that certain expenses would be incurred in connection with the acquisition and the other transactions contemplated by the Merger Agreement, there are many factors beyond our control that could affect the total amount or the timing of the integration and implementation expenses.

Integrating Wandera's business may be more difficult, costly or time-consuming than expected and we may fail to realize the anticipated benefits of the acquisition, which may adversely affect our business results and negatively affect the value of our common stock.

The success of the acquisition will depend on, among other things our ability to integrate Wandera's businesses in a manner that facilitates growth opportunities and realizes expected cost savings. We may encounter difficulties in integrating our and Wandera's businesses and realizing the anticipated benefits of the acquisition. We must achieve

anticipated post-closing growth and cost savings without adversely affecting current revenues and investments in future growth. If we are not able to successfully achieve these objectives, the anticipated benefits of the acquisition may not be realized fully, or at all, or may take longer to realize than expected.

The acquisition involves the combination of two companies which currently operate, and until the completion of the acquisition will continue to operate, as independent companies. There can be no assurances that our respective businesses can be integrated successfully. It is possible that the integration process could result in the loss of key employees from both companies, the loss of customers, the disruption of our, Wandera's or both companies' ongoing businesses, inconsistencies in standards, controls, procedures and policies, unexpected integration issues, higher than expected integration costs and an overall post-completion integration process that takes longer than originally anticipated. We will be required to devote management attention and resources to business integration efforts, and prior to the acquisition, management attention and resources will be required to plan for such integration. Potential difficulties we may encounter in the integration process include the following:

- lost sales and customers as a result of certain of our and/or Wandera's customers deciding not to do business with us post-close;
- differences in customer and vendor composition, sales cycles, and other business dissimilarities;
- integrating personnel from the two companies while maintaining focus on providing consistent, high-quality products and services, especially in the COVID-19 environment;
- potential unknown liabilities and unforeseen increased expenses, delays or regulatory conditions associated with the acquisition; and
- our and/or Wandera's performance shortfalls as a result of the diversion of management's attention caused by completing the acquisition and integrating the companies' operations.

Any of these factors could result in us failing to realize the anticipated benefits of the acquisition, on the expected timeline or at all. An inability to realize the full extent of the anticipated benefits of the acquisition and the other transactions contemplated by the Merger Agreement, as well as any delays encountered in the integration process, could have an adverse effect upon the revenues, level of expenses and our operating results, which may adversely affect the value of our common stock.

In addition, the actual integration may result in additional and unforeseen expenses, and the anticipated benefits of the integration plan may not be realized. Actual growth and cost savings, if achieved, may be lower than what we expect and may take longer to achieve than anticipated. If we are not able to adequately address integration challenges, we may be unable to successfully integrate their operations or realize the anticipated benefits of such integration.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

**Item 3. Defaults Upon Senior Securities** 

None.

**Item 4. Mine Safety Disclosures** 

Not applicable.

**Item 5. Other Information** 

None.

# Item 6. Exhibits

The following is a list of all exhibits filed or furnished as part of this report:

Exhibit Number	Description
2.1	Agreement and Plan of Merger, dated as of May 5, 2021, by and among JAMF Holding Corp., JAMF
	Software, LLC, White Wolf Merger Sub, Inc., Wandera, Inc., and Shareholder Representative Services LLC, filed herewith.
10.1	First Amendment, dated as of April 22, 2021, to the Letter Agreement between JAMF Holdings, Inc. and Dean Hager (incorporated by reference to the Company's Current Report on Form 8-K filed with the SEC on April 26, 2021).
10.2	First Amendment, dated as of April 22, 2021, to the Letter Agreement between JAMF Holdings, Inc. and Jill Putman (incorporated by reference to the Company's Current Report on Form 8-K filed with the SEC on April 26, 2021).
10.3	First Amendment, dated as of April 22, 2021, to the Letter Agreement between JAMF Holdings, Inc. and John Strosahl (incorporated by reference to the Company's Current Report on Form 8-K filed with the SEC on April 26, 2021).
31.1	Certification of the Chief Executive Officer pursuant to Exchange Act Rules Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, filed herewith.
31.2	Certification of the Chief Financial Officer pursuant to Exchange Act Rules Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, filed herewith.
32.1*	Certification of the Chief Executive Officer pursuant to 18 U.S.C. Section 1350, filed herewith.
32.2*	Certification of the Chief Financial Officer pursuant to 18 U.S.C. Section 1350, filed herewith.
101.INS	Inline XBRL Instance Document
101.SCH	Inline XBRL Taxonomy Extension Schema Document
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document
104	Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibit 101)

<sup>\*</sup> The certifications furnished in Exhibit 32.1 and Exhibit 32.2 hereto are deemed to accompany this Quarterly Report on Form 10-Q and will not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, except to the extent that the registrant specifically incorporates it by reference.

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# **Jamf Holding Corp. (Registrant)**

Date: May 11, 2021 By: /s/ Ian Goodkind

Ian Goodkind Chief Accounting Officer (Principal Accounting Officer) Confidential Execution Version

# AGREEMENT AND PLAN OF MERGER

**AMONG** 

JAMF HOLDING CORP.,

JAMF SOFTWARE, LLC,

WHITE WOLF MERGER SUB, INC.,

WANDERA, INC.

**AND** 

# SHAREHOLDER REPRESENTATIVE SERVICES LLC, AS REPRESENTATIVE OF THE COMPANY EQUITYHOLDERS

MAY 5, 2021

# TABLE OF CONTENTS

		Page
ARTICLE :	I DEFINITIONS	2
1.1	Definitions	2
1.2	Interpretation	20
ARTICLE	II THE MERGER	21
2.1	The Merger	21
2.2	Effective Time	21
2.3	Effect of the Merger	21
2.4	Certificate of Incorporation; Bylaws	21
2.5	Directors and Officers	21
ARTICLE	III MERGER CONSIDERATION; CONVERSION OF SECURITIES; SURRENDER OF	22
	ANY SHARES	
3.1	Calculation and Payment of the Merger Consideration	22
3.2	Effect of Merger on the Capital Stock and Warrants	25
3.3	Options	28
3.4	Post-Closing Adjustments	29
3.5	Effect of Merger on the Capital Stock of Merger Sub	32
3.6	Appraisal Rights	32
3.7	Withholding Rights	33
3.8	Payments Administrator	33
ARTICLE	IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY	33
4.1	Organization, Qualification, Power and Authority	33
4.2	Authorization; Enforceability	34
4.3	Capitalization	34
4.4	Non-contravention	36
4.5	Brokers' Fees	36
4.6	Financial Statements	37
4.7	Subsequent Events	38
4.8	Legal Compliance	40
4.9	Tax Matters	41
4.10	Real Property	43
4.11	Personal Property	44
4.12	Intellectual Property	44
4.13	Contracts	47
4.14	Litigation	49
4.15	Employee Benefits	49
4.16	Environmental Matters	51
4.17	Labor Matters	52
4.18	Insurance Policies	54
4.19	Affiliated Transactions	54
4.20	Material Business Relationships	55
4.21	Trade Laws	55

# TABLE OF CONTENTS (continued)

		Page
4.22	Anti-Money Laundering	56
4.23	Anti-Bribery and Anti-Corruption	56
ARTICI F	V REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	57
5.1	Organization of Parent and Merger Sub	57
5.2	Ownership of Merger Sub; No Prior Activities	57
5.3	Authorization; Enforceability	57
5.4	Non-contravention	58
5.5	Brokers' Fees	58
5.6	Litigation	58
5.7	Financing	58
5.8	Solvency	58
5.0	Solvency	50
	VI PRE-CLOSING COVENANTS RELATING TO THE COMPANY	59
6.1	Conduct of Business	59
6.2	Consents; Filings; Etc.	61
6.3	Publicity	61
6.4	Access	62
6.5	Requisite Stockholder Approval	62
6.6	Information Statement	62
6.7	Section 280G	62
6.8	Financing Cooperation	63
6.9	Exclusivity	65
6.10	R&W Insurance Policy	65
6.11	Transition.	65
6.12	Non-Public Information.	66
ARTICLE	VII COVENANTS RELATING TO PARENT AND MERGER SUB	66
7.1	Filings; Consents; Etc.	66
7.2	Publicity	67
ARTICLE	VIII CONDITIONS PRECEDENT TO THE CLOSING	67
8.1	Conditions Precedent to Each Party's Obligations	67
8.2	Conditions Precedent to Obligations of Parent and Merger Sub	68
8.3	Conditions Precedent to Obligations of the Company	68
8.4	Frustration of Closing Conditions	69
ARTICI F	IX CLOSING	69
9.1	Deliveries by the Company	69
9.2	Deliveries by Parent and Merger Sub	70
J. <u>_</u>	Zentenes of Latent and Merger out	, 0
	X POST-CLOSING COVENANTS	70
10.1	Tax Covenants	70

# TABLE OF CONTENTS (continued)

		Page
10.2	Director and Officer Liability and Indemnification.	71
10.3	Employee Matters.	72
ARTICLE	XI TERMINATION	73
11.1	Termination of Agreement	73
11.2	Effect of Termination	75
ARTICLE	XII MISCELLANEOUS	75
12.1	Expenses	75
12.2	No Third Party Beneficiaries	75
12.3	Entire Agreement	75
12.4	Succession and Assignment	75
12.5	Counterparts	76
12.6	Headings	76
12.7	Notices	76
12.8	Governing Law; Waiver of Jury Trial	77
12.9	Survival; Acknowledgment; Waiver	78
12.10	Amendments and Waivers	80
12.11	Severability	80
12.12	Construction; Disclosure	80
12.13	Incorporation of Exhibits and Schedules	81
12.14	Specific Performance	81
12.15	Representative	81
12.16	Waiver of Conflict; Privilege	84
12.17	Non-Recourse	85
Evhibit A	- Working Capital Schedule	
	- Working Capital Schedule - Form of Escrow Agreement	
-	I OIIII OI LIBEIOW / IĞIECIIICII	

Exhibit C – Form of Stockholder Letter of Transmittal

Exhibit D-1 – Form of Warrant Cancellation Agreement

Exhibit D-2 – Form of Option Cancellation Agreement

Schedule 1.1 – Principal Stockholders

Schedule 1.2 – Non-Compete Parties

Schedule 1.3 – Non-Solicit Parties

Company Disclosure Schedules

Exhibits and schedules have been omitted pursuant to Item 601(a)(5) of Regulation S-K and will be furnished on a supplemental basis to the Securities and Exchange Commission upon request.

# AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this "Agreement") is made and entered into as of May 5, 2021 by and among JAMF Holding Corp. ("Ultimate Parent"), a Delaware corporation, solely for the purposes of its obligations set forth in Section 10.3(c), JAMF Software, LLC, a Minnesota limited liability company ("Parent"), White Wolf Merger Sub, Inc., a Delaware corporation and wholly-owned Subsidiary of Parent ("Merger Sub"), Wandera, Inc., a Delaware corporation (the "Company"), and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the representative, agent and attorney-in-fact of the Company Equityholders (the "Representative").

#### **RECITALS**

WHEREAS, the board of directors of each of Ultimate Parent, Parent, Merger Sub and the Company has unanimously approved and declared advisable this Agreement and the merger of Merger Sub with and into the Company (the "Merger"), pursuant to which the Company shall continue as the surviving corporation and as a wholly-owned Subsidiary of Parent, upon the terms and subject to the conditions of this Agreement and in accordance with the General Corporation Law of the State of Delaware (the "DGCL");

WHEREAS, the board of directors of each of Parent and the Company has (a) determined that this Agreement, the Merger and the transactions contemplated by this Agreement are in furtherance of and consistent with their respective business strategies and are in the best interest of Parent, the Company or Merger Sub, as the case may be, and the respective stockholders of Parent and the Company and the sole stockholder of Merger Sub and (b) approved the Merger upon the terms and subject to the conditions set forth in this Agreement pursuant to the DGCL:

WHEREAS, the board of directors of Merger Sub, has taken all actions required for the execution of this Agreement by Merger Sub, to authorize, adopt and approve this Agreement and to approve the consummation by Merger Sub of the Merger and the other transactions contemplated by this Agreement;

WHEREAS, pursuant to the Merger and at the Effective Time, among other things, all of the issued and outstanding Company Shares, outstanding Vested Options and outstanding Company Warrants shall be converted into the right to receive a portion of the Merger Consideration as provided herein;

WHEREAS, as a condition and material inducement to the willingness of Parent and Merger Sub to enter into this Agreement, concurrently with the execution and delivery of this Agreement, each Principal Stockholder is entering into a Support Agreement in favor of Parent (each, a "Support Agreement");

WHEREAS, as a condition and material inducement to the willingness of Parent and Merger Sub to enter into this Agreement, concurrently with the execution and delivery of this Agreement, each of (a) the Persons set forth on Schedule 1.2 is entering into a form of Restrictive Covenant Agreement (each, a "Non-Competition Agreement"), and (b) the Persons set forth on Schedule 1.3 is entering into a form of Restrictive Covenant Agreement (each, a "Non-Solicitation Agreement"), in each case, in a form mutually agreed upon by Parent and the Company;

WHEREAS, as a condition and material inducement to the willingness of Parent and Merger Sub to enter into this Agreement, concurrently with the execution and delivery of this Agreement, the Company is delivering evidence to Parent of the Requisite Stockholder Approval (as defined herein); and

WHEREAS, to induce the other parties hereto to enter into this Agreement and consummate the transactions contemplated hereby, the parties to this Agreement desire to make certain representations, warranties, covenants and other agreements in connection with the Merger.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth in this Agreement, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto, intending to be legally bound hereby, agree as follows:

# ARTICLE I DEFINITIONS

- 1.1 <u>Definitions</u>. As used herein, the following terms shall have the meanings ascribed to them in this <u>Section 1.1</u>.
  - "Acquired Companies" means, collectively, the Company and each of the Company Subsidiaries.
  - "Acquired Company Intellectual Property" has the meaning set forth in Section 4.12(a).
- "<u>Acquired Company Product Data</u>" means all data and information, whether in electronic or any other form or medium, that is accessed, collected, used, processed, stored, shared, distributed, transferred, disclosed, destroyed, or disposed of by any of the Acquired Company Products.
- "Acquired Company Products" means all Software and other products and services from which the Acquired Companies have derived within the three (3) years preceding the date hereof or are currently deriving revenue from the sale, license, maintenance or other provision thereof.
- "Affiliate" of any particular Person means any other Person controlling, controlled by or under common control with such particular Person. For the purposes of this definition, "controlling," "controlled" and "control" means the possession, directly or indirectly, of the power to direct the management and policies of a Person whether through the ownership of voting securities, by contract or otherwise.
- "Aggregate Merger Consideration" means an amount equal to (i) the Closing Merger Consideration, plus (ii) an amount equal to the aggregate exercise price of all of the outstanding Vested Options (excluding, for clarity, the EMI Options to the extent exercised as of immediately prior to the Effective Time) as of immediately prior to the Effective Time, plus (iii) an amount equal to the aggregate exercise price of all Company Warrants outstanding as of immediately prior to the Effective Time, plus (iv) the Deferred Merger Consideration.

- "Agreement" has the meaning set forth in the introductory paragraph above.
- "Audited Financial Statements" has the meaning set forth in Section 4.6(a).
- "Balance Sheet" has the meaning set forth in Section 4.6(b).
- "Business" means (a) the Acquired Companies' business of providing an enterprise software service that secures mobile data and delivers controls including cost management, compliance enforcement and threat prevention and (b) any other business engaged in by the Acquired Companies as of the date hereof.
- "<u>Business Data</u>" means all Business information and all Personal Information (whether of employees, contractors, consultants, customers, consumers, or other persons and whether in electronic or any other form or medium) that is accessed, collected, used, processed, stored, shared, distributed, transferred, disclosed, destroyed, or disposed of by or on behalf of any of the Acquired Companies, including by any of the Acquired Companies' IT Systems.
- "Business Day" means any day other than a Saturday or Sunday, or a day on which banking institutions in New York or London are authorized or obligated by Law to close.
- "CARES Act" means the Coronavirus Aid, Relief, and Economic Security Act (Pub. L. 116-136) and any administrative or other guidance published with respect thereto by any Governmental Authority, or any other Law intended to address the consequences of COVID-19.
- "Cash on Hand" means all cash and cash equivalents of the Acquired Companies, as determined in accordance with FRS 102. For the avoidance of doubt, Cash on Hand shall (a) be calculated net of issued but uncleared checks and drafts, (b) include (i) checks and wire transfers and drafts deposited or available for deposit for the account of the Acquired Companies (ii) any security deposits in respect of real estate leases or deposits securing credit cards, and (iii) the aggregate exercise price of the EMI Options to the extent exercised as of immediately prior to the Effective Time and (c) exclude any amounts relating to credit card receivables or Restricted Cash.
  - "Certificate of Merger" has the meaning set forth in Section 2.2.
  - "Closing" has the meaning set forth in Section 2.2.
- "Closing Cash on Hand" means the Cash on Hand of the Acquired Companies as of 11:59 PM ET on the day immediately preceding the Closing Date.
  - "Closing Date" has the meaning set forth in Section 2.2.
- "<u>Closing Indebtedness</u>" means the Indebtedness of the Acquired Companies as of immediately prior to the Closing.
  - "Closing Merger Consideration" has the meaning set forth in Section 3.1(a).
  - "Closing Transaction Expenses" means the Transaction Expenses that are unpaid as of the Closing.

"Closing Working Capital" means, as of 11:59 PM EST on the day immediately preceding the Closing Date, an amount equal to Current Assets minus Current Liabilities calculated in accordance with FRS 102. Exhibit A represents an illustrative calculation of Closing Working Capital as of the date hereof.

"Closing Working Capital Target" means negative six million two hundred fifty thousand dollars (\$(6,250,000)). In the event that the exchange rate from the British Pound sterling to U.S. dollars, calculated using the applicable exchange rate set forth in *The Wall Street Journal*, Eastern Edition, on the Closing Date is either 10% higher or 10% lower than such exchange rate on the date of this Agreement, then the Closing Working Capital Target will be adjusted (upward or downward, as applicable) to account for the change in the exchange rate.

"Collar Amount" means \$625,000.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Common Stock" means the Common Stock of the Company, par value \$0.0001 per share.

"Company" has the meaning set forth in the introductory paragraph above.

"Company Disclosure Schedules" has the meaning set forth in Article IV.

"Company Employee" means any employee, officer, director or individual independent contractor of the Acquired Companies.

"Company Equityholders" means the Company Stockholders, Vested Option holders and the Warrantholders.

"Company Escrow Equityholders" means the Company Stockholders and the Warrantholders (excluding the EMI Stockholders).

"Company Representatives" has the meaning set forth in Section 6.12.

"Company Share" means each share of Preferred Stock and Common Stock that is issued and outstanding immediately prior to the Effective Time.

"Company's Knowledge" means the actual knowledge of Eldar Tuvey, Roy Tuvey, Ben Oxnam and Mark Flanders, none of whom will have any personal liability regarding such knowledge.

"Company Stockholders" means the holders of all issued and outstanding Company Shares immediately prior to the Effective Time.

"Company Subsidiaries" has the meaning set forth in Section 4.3(d).

"Company Transaction" means any (i) reorganization, liquidation, dissolution or recapitalization involving the Acquired Companies, (ii) merger or consolidation involving the

Acquired Companies, (iii) sale of all or any material assets of any of the Acquired Companies (other than sales of inventory in the ordinary course of business and sales of assets that are obsolete or no longer useful to the business of the applicable Acquired Company) or all or any Company Shares (including any rights to acquire, or securities convertible into or exchangeable for, any such Company Shares but excluding (x) any exercises of Options set forth in Schedule 4.3(b), (y) any exercises of Company Warrants set forth in Schedule 4.3(c) or (z) conversion of Preferred Stock, in each case in accordance with the terms and conditions of such Option, Company Warrant or Preferred Stock) of the Acquired Companies or (iv) similar transaction or business combination involving the Acquired Companies or their respective businesses or assets.

"Company Transactions Legal Advice" has the meaning set forth in Section 12.16.

"Company Warrants" means each warrant to purchase shares of Series B Preferred Stock that is issued and outstanding as of immediately prior to the Effective Time.

"Confidentiality Agreement" means that certain Mutual Non-Disclosure Agreement, dated August 31, 2020, by and between Parent and the Company.

"Consideration Schedule" has the meaning set forth in Section 3.1(b).

"Continuing Employees" has the meaning set forth in Section 10.3(a).

"Contracts" has the meaning set forth in Section 4.13.

"COVID-19" means SARS-CoV-2 or COVID-19, and any evolutions thereof or related or associated epidemics, pandemic or disease outbreaks.

"Current Assets" means the aggregate amount of the consolidated current assets of the Acquired Companies as of 11:59 PM EST on the day immediately preceding the Closing Date, calculated using only those general ledger accounts for current assets included in Exhibit A. For the avoidance of doubt, (i) Current Assets shall include only amounts that are expected to be received or consumed within 12 months of the Closing Date, and (ii) Current Assets shall exclude any Cash on Hand, Restricted Cash, deferred Tax assets, income Tax assets, and acquisition related escrow assets. Exhibit A includes an illustrative calculation of Current Assets as of the date hereof.

"Current Liabilities" means the aggregate amount of the consolidated current liabilities of the Acquired Companies as of 11:59 PM EST on the day immediately preceding the Closing Date, calculated using only those general ledger accounts for current liabilities included in Exhibit A, including both the short term and long term portion of deferred revenue, any liability related to payment of earned and accrued bonuses or commissions (provided that, to the extent that the Company does not accrue for any applicable bonus or commission in the normal course of business, then such bonus or commission shall be included in the definition of Current Liabilities to the extent earned, without regard to such lack of accrual), and all unpaid Pre-Closing Taxes that are income Taxes, but excluding (i) Transaction Expenses and (ii) Indebtedness balances included in Current Liabilities. For the avoidance of doubt, apart from the foregoing exceptions, no liability shall be classified as a Current Liability unless such liability has been consistently classified as a current liability on the Financial Statements.

"Czech Company" has the meaning set forth in Section 4.1.

"D&O Indemnified Person" has the meaning set forth in Section 10.2(a).

"Data Security Requirements" means, collectively, all of the following, as applicable, to the extent relating to Data Treatment or otherwise relating to privacy, security, or security breach notification requirements and applicable to the Acquired Companies or to the conduct of the Business, including any Business Data: (i) each Acquired Company's own published rules, policies, and procedures (whether physical or technical in nature, or otherwise); (ii) all applicable laws, rules and regulations (including, as applicable, the California Online Privacy Protection Act of 2003 (CalOPPA), General Data Protection Regulation (EU) 2016/679) (GDPR), the UK Data Protection Act 2018, the GDPR as it forms part of the laws of England and Wales, Scotland and Northern Ireland by virtue of section 3 of the European Union (Withdrawal) Act 2018, the Privacy and Electronic Communications (EC Directive) Regulations 2003, and the e-Privacy Directive (2002/58/EC), including any predecessor, successor or implementing legislation in respect of the foregoing, any amendments or re-enactments of the foregoing, and California Consumer Privacy Act (CCPA); (iii) binding industry principles (including, if applicable, the Payment Card Industry Data Security Standard (PCI DSS)); (iv) any applicable decisions, guidelines and codes of conduct applicable to processing of personal data and, more in general, to privacy matters issued by the relevant data protection authority or other Governmental Authority; and (v) contracts into which any of the Acquired Companies have entered or by which any Acquired Company is otherwise bound.

"<u>Data Treatment</u>" means the receipt, access, collection, compilation, use, processing, protection, privacy, storage, sharing, safeguarding, distribution, transfer, disclosure, security, destruction, or disposal of any personal, sensitive, or confidential information or data (whether in electronic or any other form or medium), including Personal Information, payment card data or other protected information relating to individuals.

"Debt Financing" has the meaning set forth in Section 6.8.

"Decree 231" has the meaning set forth in Section 4.23(b).

"Deferred Merger Consideration" has the meaning set forth in Section 3.1(e).

"Deferred Pro Rata Share" means with respect to a particular Company Equityholder, a fraction, (x) the numerator of which is the number of Company Shares (including shares of EMI Stock), Company Shares issuable in respect of Vested Options (other than EMI Options that are exercised for Company Shares as of immediately prior to the Effective Time), and Company Shares issuable in respect of Company Warrants that such Company Equityholder held as of immediately prior to the Effective Time, and (y) the denominator of which is the aggregate number of Company Shares (including shares of EMI Stock), Vested Options (other than EMI Options that are exercised as of immediately prior to the Effective Time), and Company Warrants that all Company Equityholders held as of immediately prior to the Effective Time.

"Designated Contacts" has the meaning set forth in Section 6.4.

"DGCL" has the meaning set forth in the recitals above.

- "Dispute Notice" has the meaning set forth in Section 3.4(b).
- "<u>Dispute Period</u>" has the meaning set forth in <u>Section 3.4(b)</u>.
- "Dissenting Shares" has the meaning set forth in Section 3.6.
- "<u>Dissenting Stockholders</u>" has the meaning set forth in <u>Section 3.6</u>.
- "Effect" has the meaning set forth in the definition of Material Adverse Effect.
- "Effective Time" has the meaning set forth in Section 2.2.
- "EMI Options" means each Option that was intended to be granted under Schedule 5 to the Income Tax (Earnings and Pensions Act (2003).
  - "EMI Stock" has the meaning set forth in Section 3.2(d).
  - "EMI Stockholders" has the meaning set forth in Section 3.2(d).
- "Employee Benefit Plan" means any "employee benefit plan" (as such term is defined in ERISA Section 3(3), whether or not subject to ERISA) and each other employment, individual consulting, retention, change in control, severance, equity, equity-based, phantom equity, stock option, stock purchase, bonus, incentive, retirement, welfare, post-employment welfare, deferred compensation or other employee benefit or compensation plan, program, policy agreement or arrangement that is maintained, sponsored or contributed to by any Acquired Company, to which any Acquired Company has any obligation to make any contributions or with respect to which any Acquired Company has any liability, provided, however, that the term "Employee Benefit Plan" shall exclude any (a) statutory employee benefit plan required pursuant to applicable Law that is maintained by a Governmental Authority and (b) any PEO Plan.
  - "Employee Benefits" has the meaning set forth in Section 10.3(a).
- "<u>Employee Option Holder</u>" means each holder of Options that is a current or former employee of the Company or its Subsidiaries.
  - "Employment Agreements" has the meaning set forth in the recitals above.
  - "End Date" has the meaning set forth in Section 11.1(b)(i).
- "Environmental Laws" means all Laws concerning pollution, human health or safety or protection of the environment, as such Laws are enacted and in effect on or prior to the Closing Date.
- "<u>Equity Incentive Plans</u>" means the Wandera, Inc. 2013 Equity Incentive Plan and the 2012 Wandera, Inc. Share Option Scheme.
  - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

- "Escrow Account" has the meaning set forth in Section 3.1(c)(i).
- "Escrow Agent" has the meaning set forth in Section 3.1(c)(i).
- "Escrow Agreement" has the meaning set forth in Section 3.1(c)(i).
- "Escrow Amount" means an amount equal to \$500,000.
- "Escrow Funds" has the meaning set forth in Section 3.1(c)(i).
- "Escrow Pro Rata Share" means with respect to a particular Company Escrow Equityholder, a fraction, (x) the numerator of which is the portion of Merger Consideration that such holder is entitled to be paid pursuant to this Agreement with respect to its Company Shares and Company Warrants, and (y) the denominator of which is the aggregate Merger Consideration that all Company Escrow Equityholders are entitled to be paid pursuant to this Agreement with respect to their Company Shares and Company Warrants, in each case after giving effect to the Reallocation.
  - "Estimated Closing Cash on Hand" has the meaning set forth in Section 3.1(b).
  - "Estimated Closing Indebtedness" has the meaning set forth in Section 3.1(b).
  - "Estimated Closing Transaction Expenses" has the meaning set forth in Section 3.1(b).
  - "Estimated Closing Working Capital" has the meaning set forth in Section 3.1(b).
  - "ET Affiliate" means The Gaskaar Trust.
  - "FCPA" has the meaning set forth in Section 4.23(a)(vi).
- "FFCRA" means the Families First Coronavirus Response Act (Public Law 116-127) and any administrative or other guidance published with respect thereto by any Governmental Authority.
  - "Financial Statements" has the meaning set forth in Section 4.6(b).
- "<u>Fraud</u>" means a claim for Delaware common law fraud with a specific intent to deceive with respect to the making by the applicable Party of the representations in <u>Article IV</u> or <u>Article V</u> of this Agreement (as applicable) brought by a Party against another Party.
- "FRS 102" means FRS 102 The Financial Reporting Standard applicable in the UK and Republic of Ireland.
- "<u>Fully-Diluted Number</u>" means an amount equal to the sum of (a) the total number of shares of Common Stock outstanding as of immediately prior to the Effective Time (including, for the avoidance of doubt, the Common Stock into which the outstanding Preferred Stock will be convertible as of immediately prior to the Effective Time and the EMI Stock), <u>plus</u> (b) the total number of shares of Common Stock that would be issued assuming the exercise of all outstanding

Vested Options (excluding any Vested Options that are EMI Options and that are exercised as of immediately prior to the Effective Time), <u>plus</u> (c) the total number of shares of Common Stock that would be issued assuming the exercise of all outstanding Company Warrants, assuming such exercise or conversion immediately prior to the Effective Time.

"Fundamental Representations" has the meaning set forth in Section 8.2(a).

"Government Official" means any officer or employee of a Governmental Authority or any department, agency, or instrumentality thereof, or of a public international organization, or any person acting in an official capacity for or on behalf of any such government or department, agency, or instrumentality, or for or on behalf of any such public international organization, or any political party, party official, or candidate thereof, excluding officials related to the government of the United States.

"Governmental Authority" means any governmental, regulatory or administrative body, agency or authority, including any supervisory authority, any court or judicial authority, any arbitrator or arbitral tribunal (public or private) or any other public authority, whether domestic or foreign, federal, state, local or supranational.

"HSR Act" means the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" means, without duplication, any liability (i) in respect of borrowed money or evidenced by bonds, notes, debentures or similar instruments, (ii) for the deferred purchase price of goods and services, including any earn-out obligations or similar contingent payables associated with past acquisitions, but excluding any trade payables in the ordinary course, (iii) in respect of guaranties, direct or indirect, in any manner, of all or any part of any Indebtedness of any Person, (iv) in respect of accrued and unpaid severance for any current or former employee, officer or director of any Acquired Company that arises before or on the Closing Date, together with the employer portion of any payroll Taxes due on the foregoing amounts (including, for the avoidance of doubt, any such Taxes which may be deferred pursuant to the CARES Act), computed as though all such amounts were payable as of the Closing Date, (v) benefits and any vested obligations arising from any nonqualified deferred compensation plans, unfunded defined benefit plans, or underfunded pension obligations, together with the employer portion of any payroll Taxes due on the foregoing amounts (including, for the avoidance of doubt, any such Taxes which may be deferred pursuant to the CARES Act), computed as though all such amounts were payable as of the Closing Date, (vi) in respect of interest, fees, prepayment premiums, penalties and other fees and expenses owed with respect to the Indebtedness referred to above assuming the repayment in full of such Indebtedness as of such time, (vii) in respect of delinquent rent, (viii) required to be paid to terminate any interest rate swaps, hedging or similar arrangements, (ix) in respect of any reimbursement obligations under or pursuant to letters of credit or other similar instruments or arrangements by which such Person assures a creditor against loss, to the extent any such letters of credit or similar instruments or arrangements have been drawn, (x) in respect of any payroll Taxes deferred pursuant to the CARES Act, (xi) in respect of deferred UK VAT under HM Revenue and Customs' COVID-19 VAT deferral scheme, (xii) in respect of any management, monitoring, advisory or other similar fees or expenses and (xiii) PTO liability accrued as of December 31, 2020 and that was not taken or utilized as of the Closing. Notwithstanding the

foregoing, Indebtedness does not include (A) any operating or lease obligations (other than capital leases), (B) any intercompany obligations between or among the Company or any Company Subsidiary, (C) any Current Liabilities, (D) any Transaction Expenses or (E) any deferred revenue.

"Indemnity Pro Rata Share" means with respect to a particular Company Equityholder, a fraction, (x) the numerator of which is the portion of Merger Consideration that such Company Equityholder is entitled to be paid pursuant to this Agreement with respect to its Company Shares, Vested Options and Company Warrants (which, for EMI Stock, shall be calculated net of the exercise price of the EMI Options paid therefor), and (y) the denominator of which is the aggregate Merger Consideration that all Company Equityholders are entitled to be paid pursuant to this Agreement with respect to their Company Shares, Vested Options and Company Warrants (which, for EMI Stock, shall be calculated net of the exercise price of the EMI Options paid therefor), in each case after giving effect to the Reallocation.

"Independent Accountant" has the meaning set forth in Section 3.4(b).

"Information Statement" has the meaning set forth in Section 6.6.

"Insurance Policies" has the meaning set forth in Section 4.18.

"Intellectual Property" means all intellectual property and proprietary rights throughout the world, including (a) patents, patent applications, and inventions and all improvements thereto (whether or not patentable or reduced to practice), (b) trademarks, service marks, logos, trade names, trade dress and domain names, and other indicia of source, together with the goodwill associated therewith, (c) copyrights, including copyrights in computer software and all works of authorship (whether or not copyrightable), (d) registrations and applications for registration of any of the foregoing, (e) intellectual property rights in Software, (f) intellectual property rights in confidential and proprietary information, trade secrets, technologies, data, databases, and documentation thereof, utility models, processes, techniques, protocols, methods, formulae, algorithms, layouts, designs, specifications and know-how, and (g) moral rights.

"Interim Financial Statements" has the meaning set forth in Section 4.6(b).

"IT Systems" means all Software, computer systems, servers, hardware, firmware, information technology systems or infrastructure, electronic data processing systems, communication networks, interfaces, platforms, peripherals, networks, network equipment, databases and data or information contained therein or transmitted thereby, and other information technology equipment, in each case, whether owned, controlled, used, held for use, outsourced, leased or licensed by the Acquired Companies in connection with the Business as currently conducted.

"<u>Labor Agreement</u>" means any collective bargaining agreement or other Contract or agreement with any labor union, labor organization or works council.

"<u>Law</u>" means any law, act, code, ruling, statute, regulation, rule, judgment, injunction, ruling, award, decree, writ, ordinance, order, legally binding written guidance, or other binding action or requirement of a Governmental Authority.

"Leased Real Property" has the meaning set forth in Section 4.10(b).

"<u>Leases</u>" means all leases and subleases, licenses, concessions and other agreement (written or oral) including all amendments, extensions, renewals and guaranties with respect thereto, pursuant to which any of the Acquired Companies holds any Leased Real Property.

"<u>Legal Proceeding</u>" means any judicial, administrative, arbitral and any other actions, suits, claims, charges, complaints, grievances, audits, investigations, inquiries, or proceedings (public or private) by or before a Governmental Authority.

"<u>Lien</u>" means any mortgage, pledge, lien, license, restriction on transfer, encumbrance, charge or other security interest.

"<u>Malicious Code</u>" means any Software code or other mechanism that (a) contains any "back door," virus, malware, Trojan horse or similar device, (b) may disrupt, disable, erase or harm the operation of Software, or cause any Software to damage or corrupt any data, hardware, storage media, programs, equipment or communications, or (c) permits any Person to access any Software, data, hardware, storage media, programs, equipment or communications without authorization.

"Material Adverse Effect" means any effect, change, event, occurrence, development or circumstance, either alone or in combination (any such item, an "Effect"), that has or would reasonably be expected to have a material adverse effect on (a) the condition (financial or otherwise), business or results of operations of the Acquired Companies, taken as a whole or (b) the ability of the Company to enter into this Agreement or consummate the transactions contemplated hereby in a timely manner; provided, however, that no Effect caused by or resulting from any of the following, either alone or in combination, shall constitute or be taken into account in determining whether there has been or would reasonably be expected to be a Material Adverse Effect: (i) any Effect affecting the economy of the United States generally or political or regulatory conditions or the economy of any country in which any Acquired Company conducts business; (ii) any Effect affecting the industries in which the Acquired Companies conduct business; (iii) any Effect arising in connection with acts of God, disasters, emergencies, calamities, pandemics or similar events including the escalation or material worsening of any such acts of God, disasters, emergencies, calamities, pandemics or similar events whether commenced before or after the date hereof, (iv) any Law, directive, pronouncement or guideline issued by a Governmental Authority, the Centers for Disease Control and Prevention, the World Health Organization or industry group providing for business closures, "sheltering-in-place," curfews or other restrictions that relate to, or arise out of, a pandemic, epidemic or disease outbreak (including the COVID-19 pandemic) or any change in such Law, directive, pronouncement or guideline or interpretation thereof following the date of this Agreement or any material worsening of such conditions existing as of the date of this Agreement, (v) any Effect arising in connection with global, national or regional political or social changes, actions or conditions, including hostilities, military actions, political instability, political elections, acts of terrorism or war or any escalation or material worsening of any such hostilities, military actions, political instability, acts of terrorism or war whether commenced before or after the date hereof; (vi) any failure, in and of itself, by any Acquired Company to meet any internal or published projections, forecasts or revenue or earnings predictions for any period (it being understood that the facts or occurrences giving rise to such failure may be taken into

account in determining whether there has been or will be, a Material Adverse Effect); (vii) compliance with, or any action expressly required to be taken by any Acquired Company under the terms of this Agreement; (viii) any Effect that results from any action taken at the express request of Parent; (ix) the announcement or pendency of the transactions contemplated by this Agreement, or (x) any change in Law or FRS 102 or interpretation thereof solely to the extent occurring after the date hereof, unless, in the cases of clauses (i), (ii), (iii), (iv) or (v) above, such changes would reasonably be expected to have a disproportionate impact on the financial condition, business or results of operations of the Acquired Companies, taken as a whole, relative to other affected participants in the industries in which any Acquired Company conducts business (in which case, only the incremental disproportionate impact shall be taken into account in determining whether there has been a Material Adverse Effect).

- "Material Contracts" has the meaning set forth in Section 4.13.
- "Material Customer" has the meaning set forth in Section 4.20(a).
- "Material Supplier" has the meaning set forth in Section 4.20(b).
- "Merger" has the meaning set forth in the recitals above.
- "Merger Consideration" has the meaning set forth in Section 3.1(a).
- "Merger Sub" has the meaning set forth in the introductory paragraph above.
- "Money Laundering Laws" has the meaning set forth in Section 4.21(a).
- "Non-Competition Agreement" has the meaning set forth in the recitals above.
- "Non-Employee Option Holder" means each holder of Options that is not an Employee Option

Holder.

- "Non-Party Affiliates" has the meaning set forth in Section 12.16.
- "Non-Solicitation Agreement" has the meaning set forth in the recitals above.
- "OFAC" has the meaning set forth in the definition of Trade Laws.

"Open Source Software" means any Software that is licensed pursuant to: (a) any license that is a license approved by the Open Source Initiative and listed at http://www.opensource.org/licenses, which licenses include all versions of the GNU General Public License (GPL), the GNU Lesser General Public License (LGPL), the GNU Affero GPL, the MIT license, the Eclipse Public License, the Common Public License, the CDDL, the Mozilla Public License (MPL), the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL), and the Sun Industry Standards License (SISL); (b) any license to Software that is considered "free" or "open source software" by the Open Source Foundation or the Free Software Foundation; or (c) any Reciprocal License, in each case whether or not source code is available or included in such license.

- "Option" has the meaning set forth in Section 3.3(a).
- "Option Cancellation Agreement" has the meaning set forth in Section 3.3(c).
- "Option Holders" means, collectively, the Employee Option Holders and the Non-Employee Option Holders.
  - "Option Payments" has the meaning set forth in Section 3.3(b).
- "Ordinary Course of Business" means, with respect to the Acquired Companies, the ordinary course of business of the Acquired Companies consistent in all material respects with past practice through the date hereof.
- "Organizational Documents" means, as applicable, (a) the certificate of incorporation; (b) the articles of incorporation; (c) the articles of organization; (d) any charter or similar document adopted or filed in connection with the creation, formation or organization of a Person; (e) the bylaws or any similar document adopted in connection with the creation, formation or organization of a Person; and (f) any amendment to any of the foregoing.
- "Owned Intellectual Property" means all Intellectual Property that is owned or purported to be owned by the Acquired Companies, including the Scheduled Intellectual Property.
  - "Parent" has the meaning set forth in the introductory paragraph above.
- "<u>Parent RSU</u>" means a restricted stock unit in respect of the common stock, par value \$0.001, of Ultimate Parent.
  - "Party" or "Parties" means the Company, the Representative, Merger Sub and Parent.
  - "Payments Administrator" means Acquiom Financial LLC.
- "<u>Payments Agreement</u>" means that certain payments administration agreement, dated as of the date hereof, by and among Parent, the Representative, and the Payments Administrator, providing for the payment of certain funds in accordance with the terms thereof.
- "<u>Payroll Tax Executive Order</u>" means the Presidential Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, as issued on August 8, 2020 and including any administrative or other guidance published with respect thereto by any Taxing authority (including IRS Notice 2020-65 and IRS Notice 2021-11).
- "<u>PEO Plan</u>" means any benefit or compensation plan, program, policy agreement or arrangement maintained by a third party professional employer organization for the benefit of employees of any Acquired Company.
- "<u>Per Share Merger Consideration</u>" means an amount equal to the quotient of (x) the Aggregate Merger Consideration, <u>divided by</u> (y) the Fully-Diluted Number; <u>provided</u> that "Per Share Merger Consideration" for (a) any Specified Investors shall be reduced by such Specified

Investor's Specified Investor Pro Rata Share of the Reallocation Amount divided by such Specified Investor's Company Shares, (b) the ET Affiliate shall be increased by 50% of the Reallocation Amount divided by the sum of the ET Affiliate's Company Shares, and (c) for the RT Affiliate shall be increased by 50% of the Reallocation Amount divided by the sum of such RT Affiliate's Company Shares.

"Per Warrant Merger Consideration" has the meaning set forth in Section 3.2(f).

"Permits" has the meaning set forth in Section 4.8.

"Permitted Liens" means each of the following: (a) Liens incurred in the Ordinary Course of Business in connection with workers' compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government Contracts, performance and return of money bonds and similar obligations; (b) mechanics', carriers', workers', repairers', materialmen's, warehousemen's and other similar Liens which have arisen in the Ordinary Course of Business for amounts which are not due and payable or which are being contested in good faith by appropriate filings; (c) Liens for Taxes not yet delinquent for which adequate reserves have been established in accordance with FRS 102; (d) non-exclusive licenses of Intellectual Property granted to customers in the Ordinary Course of Business; (e) all leases, subleases, concessions or service contracts to which any Person or any of its Subsidiaries is a party, (f) all Liens identified on title policies or preliminary title reports or other documents or writings included in the public reports, and (g) Liens which are not reasonably likely to materially impair the continued use of the asset or property to which they relate.

"<u>Person</u>" means an individual, a sole proprietorship, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, any other business entity or any Governmental Authority.

"Personal Information" means, in addition to any definition for any similar term (e.g., "personally identifiable information" or "PII") provided by applicable Law, any information relating to an identified or identifiable natural person, including personal data and personally identifiable information that identifies, can be used to identify or is otherwise associated with an individual person or device, whether or not such information is associated with an identified individual, including: (a) names, addresses, telephone numbers, email address, financial information, financial account number, personal health information, drivers' license numbers and government-issued identification numbers; and (b) Internet Protocol addresses, device identifiers or other persistent identifiers.

"<u>Pre-Closing Tax Period</u>" means all taxable periods ending on or before the Closing Date and the portion through the end of the Closing Date for any taxable period that includes (but does not end on) the Closing Date.

"Pre-Closing Taxes" means all U.S. federal, state, local and non-U.S. Taxes of the Acquired Companies for Pre-Closing Tax Periods, including (A) all liabilities for Taxes under Code Sections 951, 951A, or 965 (including any installments of Tax payable after the Closing Date with respect to any election under Code Section 965(h) (or any applicable guidance issued with

respect thereto) made prior to the Closing) to the extent the relevant inclusion is attributable to the income of a foreign entity arising in any such period and calculated by assuming that the taxable period of each Acquired Company ended as of the Closing Date, (B) any payroll Taxes deferred under the CARES Act or the Payroll Tax Executive Order that are attributable to a Pre-Closing Tax Period, (C) any Taxes of any members of an affiliated, consolidated, combined, or unitary group of which any of any of the Acquired Companies (or any predecessor) is or was a member on or prior to the Closing Date, including pursuant to Treasury Regulations Section 1.1502-6 (or any analogous or similar state, local, or foreign Law), (D) any Taxes of any Person imposed on any of the Acquired Companies as a result of any express or implied obligation to indemnify or otherwise assume or succeed to the liability of any other Person, including as a transferee or successor, by contract or operation of Law or otherwise, which Taxes relate to an event or transaction occurring before the Closing, and (E) any Transfer Taxes borne by the Company Equityholders pursuant to Section 10.1(a). For any taxable period that includes but does not end on the Closing Date, Pre-Closing Taxes shall be calculated as if such taxable periods ended on the Closing Date using (i) a closing-of-the-books method for Taxes based on income, payroll or receipts (and for such purpose, the Taxable period of any "controlled foreign corporation" as defined in Section 957 in which any Acquired Company holds a beneficial interest will be deemed to terminate at such time), except that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions), other than with respect to property placed in service after the Closing, shall be allocated on a daily pro-ration method and (ii) a daily pro-ration method for all other Taxes; provided, that, the amount of such unpaid Taxes shall not be less than zero (0). Notwithstanding the foregoing, Pre-Closing Taxes shall not include the employer portion of any payroll Taxes due in connection with any amounts included in the definition of Indebtedness or Transaction Expenses (including, for the avoidance of doubt, any such Taxes which may be deferred pursuant to the CARES Act). For the avoidance of doubt, Pre-Closing Taxes shall include the amount of any Taxes for Pre-Closing Tax Periods arising from prepaid amounts (if any) that are required to be accelerated into Pre-Closing Tax Periods as a result of the transactions contemplated by this Agreement.

"<u>Preferred Stock</u>" means the Series Seed Preferred Stock, Series A Preferred Stock, Series A-1 Preferred Stock, Series B Preferred Stock and Series C Preferred Stock.

"Principal Stockholder" means each holder of Preferred Stock set forth on Schedule 1.1.

"R&W Insurance Policy" has the meaning set forth in Section 6.10.

"Reallocation Amount" means an amount in cash equal to \$6,400,000.

"Reciprocal License" means a license of an item of Software that requires or that conditions any rights granted in such license upon: (i) the disclosure, distribution or licensing of any other Software (other than such item of Software as provided by a third party in its unmodified form); (ii) a requirement that any disclosure, distribution or licensing of any other Software (other than such item of Software in its unmodified form) be at no charge; (iii) a requirement that any other licensee of the Software be permitted to access the source code of, modify, make derivative works of, or reverse-engineer any such other Software; (iv) a requirement that such other Software be redistributable by other licensees; or (v) the grant of any patent rights (other than patent rights

in such item of Software), including non-assertion or patent license obligations (other than patent obligations relating to the use of such item of Software).

"Representative" has the meaning set forth in the introductory paragraph above.

"Representative Holdback Amount" means an amount equal \$150,000.

"Representative Holdback Distribution Date" has the meaning set forth in Section 12.15(d)(iii).

"Representative Losses" has the meaning set forth in Section 12.15(d)(ii).

"Requisite Stockholder Approval" means, with respect to the Company, the affirmative vote or consent to approve of the transactions contemplated by this Agreement and the Agreement itself of (A) at least two (2) of the three (3) following constituencies: (i) the holders of at least the majority of the shares of Common Stock then issued or issuable upon conversion of the Series C Preferred Stock then outstanding (voting as a single series), (ii) the holders of at least the majority of the shares of Common Stock then issued or issuable upon conversion of the Series B Preferred Stock then outstanding (voting as a single series), (iii) the holders of at least the majority of the shares of Common Stock then issued or issuable upon conversion of the Series A Preferred Stock and Series A-1 Preferred Stock then outstanding (voting as a single series and on an as-converted basis), and (B) the holders of a majority of the outstanding shares of Common Stock and Preferred Stock (on an asconverted to Common Stock basis), in each case, which Shares are outstanding on the record date chosen for purposes of determining stockholders of the Company entitled to vote on the adoption of this Agreement, voting as a single class on an as-converted basis in accordance with the DGCL and the Organizational Documents of the Company.

"Restated Certificate" means the Amended and Restated Certificate of Incorporation of the Company, dated as of May 12, 2017.

"Restricted Cash" shall mean any cash or cash equivalents of the Acquired Companies that (a) is subject to a restriction or limitation of any kind, including any cash or cash equivalents securing any Indebtedness, vendor, customer or other security deposits, or similar obligations, (b) is held as collateral or held in trust for or is otherwise to be remitted to business relations of the Acquired Companies or (c) the declaration, repatriation or payment of dividends or similar distributions by the Acquired Companies of which is not permitted, directly or indirectly, without any prior approval of any governmental entity or by its Organizational Documents or by any contract, law, order or lien applicable to the Acquired Companies or such cash and cash equivalents (including, for the avoidance of doubt, reserves of profits which are required to be set aside); provided that no security deposits in respect of real estate leases or deposits securing credit cards shall be deemed to be Restricted Cash. Any monetary conversion from the currency of a foreign country to United States dollars will be calculated using the applicable exchange rates set forth in *The Wall Street Journal*, Eastern Edition, on the Closing Date.

"RSU Agreement" has the meaning set forth in Section 10.3(c).

"RT Affiliate" means The Twegg Trust.

"Sanctioned Country" means any country or region that is, or has been in the last five years, the subject or target of a comprehensive embargo under Trade Controls, including Crimea, Cuba, Iran, North Korea, Sudan, Venezuela, and Syria.

"Sanctioned Person" means (i) any Person located, organized, or resident in a Sanctioned Country, (ii) any Person named on any U.S. sanctions or export controls list, including OFAC's Specially Designated Nationals List, the Sectoral Sanctions Identifications List, and the Foreign Sanctions Evaders List, (iii) any Person listed on any applicable non-U.S. sanctions- or export-related restricted party list, including the EU Consolidated List, and (iv) any entity that is, in the aggregate, 50 percent or greater owned, directly or indirectly, or otherwise controlled by a Person or Persons described in clauses (i) through (iii).

"Scheduled Intellectual Property" has the meaning set forth in Section 4.12(a).

"Security Incident" means any (a) successful breach of security, phishing incident, ransomware or malware attack affecting any IT Systems that processes Business Data, or (b) incident in which Business Data was accessed, disclosed, destroyed, processed, used or exfiltrated in an unauthorized manner (whether any of the foregoing was possessed or controlled by or on behalf of an Acquired Company).

"<u>Series A Preferred Stock</u>" means the Series A Preferred Stock of the Company, par value \$0.0001 per share.

"Series A-1 Preferred Stock" means the Series A-1 Preferred Stock of the Company, par value \$0.0001 per share.

"Series B Preferred Stock" means the Series B Preferred Stock of the Company, par value \$0.0001 per share.

"Series C Preferred Stock" means the Series C Preferred Stock of the Company, par value \$0.0001 per share.

"Series Seed Preferred Stock" means the Series Seed Preferred Stock of the Company, par value \$0.0001 per share.

"Software" means any and all (a) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code, object code, human readable form or other form, (b) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons and (c) all documentation including user manuals and other training documentation relating to any of the foregoing.

"<u>Specified Investors</u>" means Bessemer Venture Partners VIII Institutional L.P., Bessemer Venture Partners VIII L.P., Sapphire Ventures Fund II, L.P., SAPPHIRE SAP HANA FUND OF FUNDS, L.P., 83 North II Limited Partnership and 83 North III Limited Partnership.

"Specified Investor Pro Rata Share" means, with respect to a particular Specified Investor, a fraction, (x) the numerator of which is the portion of Merger Consideration that such Specified Investor is entitled to be paid pursuant to this Agreement with respect to its Company Shares, and (y) the denominator of which is the aggregate Merger Consideration that all Specified Investors are entitled to be paid pursuant to this Agreement with respect to their Company Shares, in each case without giving effect to the Reallocation.

- "Statement of Closing Cash on Hand" has the meaning set forth in Section 3.4(a).
- "Statement of Closing Indebtedness" has the meaning set forth in Section 3.4(a).
- "Statement of Closing Transaction Expenses" has the meaning set forth in Section 3.4(a).
- "Statement of Closing Working Capital" has the meaning set forth in Section 3.4(a).
- "Statements" has the meaning set forth in Section 3.4(a).
- "Stockholder Letter of Transmittal" has the meaning set forth in Section 3.2(g).
- "Subsidiary" means, with respect to any Person, any entity of which securities or other ownership interests having ordinary voting power to elect a majority of the board of directors or other persons performing similar functions are at any time directly or indirectly owned by such Person.
  - "Support Agreement" has the meaning set forth in the recitals above.
  - "Surviving Corporation" has the meaning set forth in Section 2.1.

"Tax" or "Taxes" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, escheat, unclaimed or abandoned property, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add on minimum, estimated or any other tax of any kind whatsoever, including any interest, penalty or additions thereto, whether disputed or not and including any obligations to indemnify or otherwise assume or succeed to the tax liability of any other Person (other than obligations under leases, credit agreements or other commercial agreements entered into in the Ordinary Course of Business the primary purpose of which does not relate to Taxes).

"<u>Tax Return</u>" means any return, election, form, declaration, report, claim for refund or information return, statement or any other document of any nature whatsoever relating to Taxes filed or required to be filed with any Governmental Authority and relating to Taxes, including any schedule or attachment thereto and any amendment thereof.

"<u>Trade Laws</u>" means all applicable customs, import, export, and sanctions laws and regulations in jurisdictions in which the Acquired Companies do business or are otherwise subject

to jurisdiction, including Title 19 of the Code of Federal Regulations and the associated statutes; the U.S. International Traffic in Arms Regulations; the Export Administration Regulations; U.S. sanctions laws and regulations administered by the U.S. Department of the Treasury's Office of Foreign Assets Control ("OFAC"), including OFAC's Specially Designated Nationals List, the Sectoral Sanctions Identifications List, and the Foreign Sanctions Evaders List; and economic or financial sanctions or trade embargoes imposed, administered, or enforced from time to time by the United Nations Security Council, the European Union, or her Majesty's Treasury of the United Kingdom.

"Transaction Expenses" means, without duplication, (a) the third party fees and expenses payable by the Acquired Companies incurred in connection with the negotiation and execution of this Agreement and the consummation of the transactions contemplated hereby (for the avoidance of doubt, excluding any fees and expenses related to indebtedness incurred by Parent or Merger Sub in connection with the transactions contemplated hereby), (b) the amount of sale, transaction, change of control, retention or similar bonus and/or any other incentive compensation, severance or termination payments or benefits due or arising solely as a result of the transactions contemplated hereby, excluding any such payments under any agreement or arrangement entered into with Parent or any of its Affiliates or with the Company or any Company Subsidiary at the request of Parent or any of its Affiliates, (c) 50% of the fees and expenses payable to the Payments Administrator at Closing in accordance with and subject to the terms of the Payments Agreement, (d) 50% of the employer's share of any employment, payroll, social security, unemployment, withholding or other similar Taxes payable as a result of or in connection with the consummation of the transactions contemplated hereby (including any payroll Taxes that the Company or any Company Subsidiary elects to defer pursuant to Section 2302 of the CARES Act), including, without duplication, any amounts described in clause (b) to the extent paid or accrued on or prior to the Closing and the Option Payments (provided that Parent's liability for the remaining 50% of such Taxes shall not exceed \$600,000), (e) any fees or costs to obtain the release and termination of any Liens, other than Permitted Liens, (f) any cost, fee or expense to obtain waivers, consents or approvals of any Person on behalf of the Company or any of its Subsidiaries in connection with the transactions contemplated by this Agreement (excluding, for the avoidance of doubt, the costs contemplated by Section 7.1(a)) (g) any dissenters' costs or liabilities arising out of the approval of the Merger or the allocation of the Merger Consideration, (h) 50% of any Transfer Taxes, (i) 50% of any costs, fees and expenses of the Escrow Agent, and (j) any costs, fees or expenses to obtain the tail policy contemplated by Section 10.2. Notwithstanding the foregoing, Transaction Expenses does not include (A) any Indebtedness, (B) any Current Liabilities, or (C) the Parent RSUs.

"Treasury Regulations" means the Treasury Regulations promulgated under the Code.

"UK Act" has the meaning set forth in Section 4.23(a)(vi).

"<u>Ultimate Parent</u>" has the meaning set forth in the introductory paragraph above.

"<u>Vested Option</u>" means an Option (or any portion thereof) that, as of immediately prior to the Effective Time, is vested, outstanding and exercisable, including any such Option (or portion thereof) that vests in connection with the consummation of the transaction contemplated

by this Agreement, either in accordance with the terms of the award agreement under which such Option was issued or as a result of the board of directors (or committee thereof) of the Company taking action, prior to the Effective Time, to accelerate the vesting thereof.

"Waived 280G Benefits" has the meaning set forth in Section 6.7.

"WARN Act" has the meaning set forth in Section 4.17(b).

"Warrant Cancellation Agreement" has the meaning set forth in Section 3.2(g).

"<u>Warrantholder</u>" means each holder, as of immediately prior to the Effective Time, of a Company Warrant.

# 1.2 <u>Interpretation</u>.

- Unless otherwise expressly provided or unless the context requires otherwise: (i) all (a) references in this Agreement to Articles, Sections, Schedules and Exhibits shall mean and refer to Articles, Sections, Schedules and Exhibits of this Agreement; (ii) all references to statutes and related regulations shall include all amendments of the same and any successor or replacement statutes and regulations; (iii) words using the singular or plural number also shall include the plural or singular number, respectively; (iv) references to "hereof," "herein," "hereby" and similar terms shall refer to this entire Agreement (including the Schedules and Exhibits hereto); (v) references to any Person shall be deemed to mean and include the successors and permitted assigns of such Person (or, in the case of a Governmental Authority, Persons succeeding to the relevant functions of such Person); (vi) the term "including" shall be deemed to mean "including, without limitation;" (vii) words of any gender shall include each other gender; (viii) whenever this Agreement refers to a number of days, such number shall refer to calendar days, unless such reference is specifically to "Business Days;" (ix) the terms "year" and "years" mean and refer to calendar year(s); (x) the term "dollars" and all uses of "\$" shall mean United States dollars and references in this Agreement to dollar amount thresholds or baskets shall not be deemed to be evidence of materiality; (xi) the terms "material" or "material to the Business" and the concept of the "material" nature of an effect upon the Business, the Company, the Acquired Companies or the Company and its Subsidiaries shall be measured relative to the entire Business, taken as a whole, and the Acquired Companies, taken as a whole, as the Business of the Acquired Companies are currently being conducted; (xii) the term "or" has the inclusive meaning represented by the phrase "and/or;" (xiii) references to notices from a third party to an Acquired Company prior to the Closing shall be deemed to mean in writing and delivered to a person included in the definition of Company's Knowledge. When calculating the period of time before which, within which or following which, any act is to be done or step taken pursuant to this Agreement, the date that is the reference date from which the period is to be calculated shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day. The phrase "made available," "provided to" or similar phrases when used in reference to a document includes any document made available for viewing in the Company's electronic data room at least twenty-four (24) hours prior to the date hereof.
- (b) The Article and Section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties and shall not in any way affect

the meaning or interpretation of this Agreement. This Agreement is the product of negotiations among the Parties, each of which is represented by legal counsel, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement. Rules of construction relating to interpretation against the drafter of an agreement shall not apply to this Agreement and are expressly waived by each Party.

# ARTICLE II THE MERGER

- 2.1 <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL, at the Effective Time, Merger Sub shall be merged with and into the Company. As a result of the Merger, the separate corporate existence of Merger Sub shall cease, and the Company shall continue as the surviving corporation and a wholly-owned Subsidiary of Parent (sometimes referred to, in such capacity, as the "<u>Surviving Corporation</u>").
- 2.2 <u>Effective Time</u>. Unless this Agreement is earlier terminated pursuant to <u>Section 11.1</u>, the closing of the Merger (the "<u>Closing</u>") will take place the later of (i) July 1, 2021 and (ii) within two Business Days after the satisfaction or, if permissible, waiver of the conditions set forth in <u>Article VIII</u> (other than those conditions) via electronic exchange of signatures. The date upon which the Closing actually occurs is referred to herein as the "<u>Closing Date</u>." On the Closing Date, the Parties shall cause the Merger to be consummated by filing a certificate of merger (the "<u>Certificate of Merger</u>") with the Secretary of State of the State of Delaware, in such form as required by, and executed in accordance with, the relevant provisions of the DGCL (the date and time of acceptance by the Secretary of State of the State of Delaware of such filing, or, if another date and time is specified in such filing, such specified date and time, being the "<u>Effective Time</u>").
- 2.3 <u>Effect of the Merger</u>. At the Effective Time, the effect of the Merger shall be as provided in the applicable provisions of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, except as otherwise provided herein, all the property, assets, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities, duties and obligations of the Company and Merger Sub shall become the debts, liabilities, duties and obligations of the Surviving Corporation.
- 2.4 <u>Certificate of Incorporation; Bylaws</u>. At the Effective Time, the Organizational Documents of the Surviving Corporation shall by operation of law be amended and restated in their entirety, but reflecting any necessary amendments to reflect the provisions of <u>Section 10.2</u>, to contain the provisions set forth in the Organizational Documents of Merger Sub as in effect immediately prior to the Effective Time.
- 2.5 <u>Directors and Officers</u>. At the Effective Time, (a) the directors of Merger Sub immediately prior to the Effective Time shall be the initial directors of the Surviving Corporation, each to serve in accordance with the Organizational Documents of the Surviving Corporation, and (b) the officers of the Company immediately prior to the Effective Time shall be

the initial officers of the Surviving Corporation, each to hold office in accordance with the Organizational Documents of the Surviving Corporation.

# ARTICLE III MERGER CONSIDERATION; CONVERSION OF SECURITIES; SURRENDER OF COMPANY SHARES

- 3.1 Calculation and Payment of the Merger Consideration.
- (a) <u>Calculation of Merger Consideration</u>. The aggregate amount to be paid by Parent at and following the Closing with respect to the Company Shares, Options and Company Warrants shall equal the following:
  - (i) \$400,000,000 (FOUR HUNDRED MILLION UNITED STATES DOLLARS);
  - (ii) <u>minus</u> the amount, if any, by which the Estimated Closing Working Capital is less than the Closing Working Capital Target (<u>provided</u> that if the difference is less than the Collar Amount, then such amount shall equal zero) (the "<u>Estimated Working Capital Shortfall</u>");
  - (iii) <u>plus</u> the amount, if any, by which the Estimated Closing Working Capital is greater than the Closing Working Capital Target (provided that if the difference is less than the Collar Amount, then such amount shall equal zero) (the "<u>Estimated Working Capital Surplus</u>");
  - (iv) <u>plus</u> the Estimated Closing Cash on Hand (as determined in accordance with <u>Section 3.1(b)</u> below);
  - (v)  $\underline{\text{minus}}$  the Estimated Closing Indebtedness (as determined in accordance with Section 3.1(b) below);
  - (vi)  $\underline{\text{minus}}$  the Estimated Closing Transaction Expenses (as determined in accordance with Section 3.1(b) below).

The aggregate amount of items (i) through (vi) immediately above, less the Deferred Merger Consideration, is referred to herein as the "Closing Merger Consideration"; provided that, after the Effective Time, the Closing Merger Consideration shall be subject to the adjustments set forth in Section 3.4 (the Closing Merger Consideration as reduced at the Closing by the Escrow Amount and the Representative Holdback Amount, and subject to adjustment as provided in Section 3.4, the "Merger Consideration").

(b) <u>Estimated Merger Consideration; Consideration Schedule</u>. At least five (5) Business Days prior to the Closing Date, the Company shall deliver to Parent (i) a statement setting forth in reasonable detail the Company's good faith estimate of the Closing Working Capital (the "<u>Estimated Closing Working Capital</u>"), the Closing Cash on Hand (the "<u>Estimated Closing Cash on Hand</u>"), the Closing Indebtedness (the "<u>Estimated Closing Indebtedness</u>"), the Closing Transaction Expenses (the "<u>Estimated Closing Transaction Expenses</u>") and the calculation of the

Closing Merger Consideration resulting therefrom, in each case determined in accordance with the definitions of Closing Working Capital, Closing Cash on Hand, Closing Indebtedness, and Closing Transaction Expenses, as applicable, in each case, as set forth in Section 1.1, together with reasonable supporting documentation, and (ii) a spreadsheet setting forth (A) a list of the Company Stockholders and holders of Company Warrants as of immediately prior to the Effective Time and their e-mail addresses (to extent available), (B) the class and number of shares of Company Shares owned by such Company Stockholder, (C) the number of Options or Company Warrants held by each Option Holder or holder of Company Warrants as of immediately prior to the Effective Time and the applicable exercise or conversion price per share of Common Stock subject to such Options or Company Warrants, as applicable, (D) the Per Share Merger Consideration, (E) each Company Equityholder's Indemnity Pro Rata Share and Deferred Pro Rata Share, (F) each Company Escrow Equityholder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, (G) each Company Escrow Equityholder's Escrow Pro Rata Share, and (H) the portion of the Aggregate Merger Consideration payable to each Company Equityholder at the Closing in accordance with the Restated Certificate and this Article III, which for the avoidance of doubt, shall be reduced (1) for each Company Escrow Equityholder by such Company Escrow Equityholder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount and (2) for each Company Equityholder by such Company Equityholder's Deferred Pro Rata Share of the Deferred Merger Consideration (the "Consideration Schedule"). The Estimated Closing Working Capital, the Estimated Closing Cash on Hand, the Estimated Closing Indebtedness, and the Estimated Closing Transaction Expenses will be included in the determination of the Closing Merger Consideration as set forth in Section 3.1(a). above. During the period after the delivery of the statement pursuant to this Section 3.1(b) and prior to the Closing, Parent shall have the opportunity to review such statement, and the Company shall cooperate with Parent in good faith to respond to any questions regarding such statement raised by Parent and review and consider in good faith any comments proposed by Parent with respect to the amounts and calculations in such statement; provided, that the Company shall have no obligation to modify the Estimated Closing Working Capital, Estimated Closing Cash on Hand, Estimated Closing Indebtedness, and the Estimated Closing Transaction Expenses, and the resulting calculation of Merger Consideration, and no dispute with respect to the statement delivered by the Company pursuant to this Section 3.1(b) shall be grounds for any failure of any closing condition to be satisfied or for the Closing to be delayed.

- (c) <u>Payment and Allocation of Merger Consideration</u>. At and upon the Effective Time, Parent shall remit the Merger Consideration by wire transfer of immediately available funds as follows:
  - (i) a portion of the Merger Consideration in the amount of the Escrow Amount shall be deposited by Parent in a non-interest-bearing account (the "Escrow Account"), with Citibank, N.A. (the "Escrow Agent") to be held pursuant to an escrow agreement (the "Escrow Agreement"), substantially in the form of Exhibit B hereto, to be entered into at the Closing by Parent, the Representative and the Escrow Agent. Such amounts delivered to the Escrow Agent and subject to any reduction for distributions made pursuant to the terms of the Escrow Agreement, are referred to herein, collectively, as the "Escrow Funds" and will secure Parent's right to receive payments in connection with adjustments to the Merger Consideration pursuant to Section 3.4(c);

- (ii) a portion of the Merger Consideration in the amount of the Representative Holdback Amount shall be deposited by Parent with the Representative to satisfy potential future obligations of the Company Escrow Equityholders, which Representative Holdback Amount will be retained by the Representative in accordance with this Agreement;
- (iii) subject to Section 3.1(e), (A) a portion of the Merger Consideration payable to the Non-Employee Option Holders pursuant to Section 3.3 or the holders of Company Warrants shall be remitted by Parent to the Payments Administrator for distribution to the Non-Employee Option Holders or holders of Company Warrants, as applicable, in accordance with the Consideration Schedule and (B) a portion of the Merger Consideration payable to the Employee Option Holders pursuant to Section 3.3 shall be remitted by Parent to the Payments Administrator for further distribution to the Company (which amount the Company (or one of the Company Subsidiaries) shall pay to the Employee Option Holders in accordance with the Consideration Schedule through the Company's (or one of the Company Subsidiaries') payroll system on or prior to the next regularly scheduled payroll date of the Company (or the applicable Company Subsidiary) that occurs after the Closing, but in no event sooner than three (3) Business Days after the Closing); and
- (iv) subject to Section 3.1(e), the remaining amount of the Merger Consideration, after subtraction of (1) the amounts remitted in accordance with Section 3.1(c)(i) through Section 3.1(c)(iii), and (2) the portion thereof otherwise allocable in accordance with Section 3.6 to Dissenting Shares, shall be remitted by Parent to the Payments Administrator for the benefit of and distribution to the Company Equityholders (other than the Option Holders) as provided in this Article III and in accordance with the Consideration Schedule.
- (d) <u>Payment of Transaction Expenses</u>. At and upon the Effective Time, Parent shall pay, or cause to be paid, on behalf of the Acquired Companies and the Company Equityholders, the Estimated Closing Transaction Expenses by wire transfer of immediately available funds to the Persons or bank accounts and in the amounts specified a schedule to be provided by the Company to Parent at least two (2) Business Days prior to the Closing Date and which shall include the identity of each payee, dollar amounts owed, wire transfer instructions and any other information necessary to effect the final payment in full of the Estimated Closing Transaction Expenses specified thereon; provided that the amounts described in clauses (b) and (d) of the definition of Transaction Expenses shall be paid by Parent to the Company and the Company (or one of the Company Subsidiaries) shall pay such amounts to the recipients thereof through the Company's (or the applicable Company Subsidiary's) payroll as soon as reasonably practicable after such amount becomes due to such recipient (such payment to be net of withholdings, if any, and without interest), or to the applicable Governmental Authority with respect to any applicable Tax withholdings.
- (e) <u>Deferred Payment</u>. Following the Closing, Parent and the Company hereby agree that (i) on October 1, 2021, Parent shall pay or cause to be paid by wire transfer of immediately available funds and pursuant to wire instructions provided by the Representative at least three Business Days prior to such date, to the Payments Administrator, cash in an aggregate

amount equal to \$25,000,000 (the "Initial Deferred Payment"), and (ii) on December 15, 2021, Parent shall pay or cause to be paid by wire transfer of immediately available funds and pursuant to wire instructions provided by the Representative at least three Business Days prior to such date, to the Payments Administrator, cash in an aggregate amount equal to \$25,000,000 (the "Second Deferred Payment" and, together with the Initial Deferred Payment, the "Deferred Merger Consideration"). The Payments Administrator shall distribute the Initial Deferred Payment and the Second Deferred Payment, in each case, within five Business Days after receipt thereof, (A) to Company Equityholders (other than former holders of Options) in the amounts set forth for such Company Equityholder in the Consideration Schedule in accordance with their Deferred Pro Rata Shares, which amounts shall be paid by the Payments Administrator by check or wire transfer in accordance with the instructions provided by such Company Equityholder; and (B) to the Surviving Corporation the aggregate amount payable to former holders of Options as set forth in the Consideration Schedule (for further distribution to such former holders). For the avoidance of doubt, payment of the Deferred Merger Consideration when due is an unconditional joint obligation of Parent and the Company and the Deferred Merger Consideration is not subject to set-off or reduction in any manner whatsoever. Parent and the Company hereby waive any requirement of presentment, protest or demand for performance or any notice of nonperformance or other notices of any kind by Company Equityholders or Representative with respect to payment of the Initial Deferred Payment or Second Deferred Payment when due. The parties hereto agree that should Parent not make the full payment of any Deferred Merger Consideration within five (5) Business Days the date such Deferred Merger Consideration becomes payable, any amount payable shall accrue interest from the date such Deferred Merger Consideration becomes payable to the date such Deferred Merger Consideration has been paid at an interest rate equal to three percent (3%).

# 3.2 <u>Effect of Merger on the Capital Stock and Warrants.</u>

Settlement of Preferred Stock. Subject to Section 3.2(c), each share of Preferred Stock (other than Dissenting Shares and Company Shares to be cancelled pursuant to <u>Section 3.2(c)</u>) shall, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the Company Stockholders, automatically be cancelled and extinguished, and each share of Preferred Stock (other than Dissenting Shares and Company Shares to be cancelled pursuant to Section 3.2(c)) shall be converted into the right to receive the Per Share Merger Consideration (as reduced at the Closing by the applicable holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, the applicable holder's Deferred Pro Rata Share of the Deferred Merger Consideration, and subject to adjustment as provided in Section 3.4, in each case in respect of such holder's shares of Preferred Stock) with respect to such Preferred Stock in cash, payable in accordance with Section 3.2(b) as if each such Preferred Stock had converted into Common Stock in accordance with the Restated Certificate immediately prior to the Effective Time. From and after the Effective Time, each holder of outstanding shares of Preferred Stock shall cease to have any rights with respect to such shares of Preferred Stock, except (i) Dissenting Shares shall have appraisal rights under the DGCL, and (ii) each other share of Preferred Stock shall have the right to receive the Per Share Merger Consideration (as reduced at the Closing by the applicable holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, the applicable holder's Deferred Pro Rata Share of the Deferred Merger Consideration, and subject to adjustment as provided in Section 3.4, in each case in respect of such holder's shares of Preferred Stock) as if each such

Preferred Stock had converted into Common Stock in accordance with the Restated Certificate immediately prior to the Effective Time subject to the conditions provided in this <u>Article III</u>.

- Conversion and Cancellation of Common Stock. Subject to Section 3.2(c) and Section 3.2(d) as the case may be, at the Effective Time, each share of Common Stock (other than Dissenting Shares and Company Shares to be cancelled pursuant to <u>Section 3.2(c)</u>) shall, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the Company Stockholders, be cancelled and extinguished, and each share of Common Stock (other than Dissenting Shares and Company Shares to be cancelled pursuant to Section 3.2(c)) shall be converted into the right to receive the Per Share Merger Consideration (as reduced at the Closing by the applicable holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, the applicable holder's Deferred Pro Rata Share of the Deferred Merger Consideration, and subject to adjustment as provided in Section 3.4, in each case in respect of such holder's shares of Common Stock) in cash, payable in accordance with and subject to the conditions provided in this Article III. From and after the Effective Time, each holder of outstanding shares of Common Stock shall cease to have any rights with respect to such shares of Common Stock, except (i) Dissenting Shares shall have appraisal rights under the DGCL, and (ii) each other share of Common Stock shall have the right to receive the Per Share Merger Consideration (as reduced at the Closing by the applicable holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, the applicable holder's Deferred Pro Rata Share of the Deferred Merger Consideration, and subject to adjustment as provided in Section 3.4, in each case in respect of such holder's shares of Common Stock) subject to the conditions provided in this Article III.
- (c) <u>Reallocation among the RT Affiliate, ET Affiliate, and Specified Investors.</u> Notwithstanding anything to the contrary herein, at the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company, or the Company Stockholders, the aggregate Merger Consideration otherwise payable to the Specified Investors shall be reduced by the Reallocation Amount (on a pro rata basis among the Specified Investors), and the aggregate Merger Consideration otherwise payable to the RT Affiliate and the ET Affiliate shall be increased by the Reallocation Amount (with respect to each such Person, by 50% of the Reallocation Amount) (the "<u>Reallocation</u>"). For the avoidance of doubt, the Reallocation shall have no effect on the Per Share Merger Consideration payable to any other Company Equityholders other than the Specified Investors, the RT Affiliate and the ET Affiliate.
- (d) <u>Common Stock Underlying EMI Options</u>. Notwithstanding anything to the contrary herein, Parent will withhold from the aggregate Merger Consideration otherwise payable to each holder of Common Stock who acquired such Common Stock pursuant to the exercise of EMI Options on or after the date of this Agreement (such Common Stock, the "<u>EMI Stock</u>", and such holders, the "<u>EMI Stockholders</u>") an amount equal to the aggregate exercise price of such EMI Options that the EMI Stockholder has irrevocably undertaken to pay to the Company upon the exercise of such EMI Options. Following the Closing, Parent will pay each such withheld sum to the Company in full satisfaction of each EMI Stockholder's irrevocable undertaking to pay the aggregate exercise price for the exercise of their EMI Options to the Company.

- (e) <u>Treasury Stock</u>. At the Effective Time, each Company Share held in the treasury of the Company immediately prior to the Effective Time shall be cancelled and extinguished without any conversion thereof, and no payment shall be made with respect thereto.
- Company Warrants. Prior to the Effective Time, the Company shall deliver to the holders of Company Warrants notices consistent with the terms and conditions of the applicable Company Warrant. No Company Warrants, whether vested or unvested, shall be assumed by Parent in the Merger, and at the Effective Time, each Company Warrant will, by virtue of the Merger, and without any further action on the part of any holder thereof, be cancelled at the Effective Time, and in exchange therefor, each holder of a Company Warrant shall be entitled to receive, in consideration of the cancellation of such Company Warrant an amount in cash (without interest) equal to the product of (x) the number of shares of Series B Preferred Stock that were issuable upon exercise of such Company Warrant immediately prior to the Effective Time, multiplied by (y) the excess, if any, of the Per Share Merger Consideration (as reduced at the Closing by such holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, such holder's Deferred Pro Rata Share of the Deferred Merger Consideration, and subject to adjustment as provided in <u>Section 3.4</u>, in each case in respect of the Company Shares underlying such holder's Company Warrants) over the per share exercise price of such Company Warrant (the product of clause (x) and (y) with respect to any Company Warrant, the "Per Warrant Merger Consideration"); provided, however, that if the Per Warrant Merger Consideration would be a negative amount (such Company Warrants, the "Out of the Money Company Warrants" and all Company Warrants that are not Out of the Money Company Warrants, the "In the Money Company Warrants"), such Company Warrant shall be cancelled and extinguished at the Effective Time without any present or future right to receive any portion of the Merger Consideration.
- Surrender of Company Shares and Company Warrants. As soon as practicable after the date hereof, but in any event within ten (10) days, the Payments Administrator shall mail to each (i) Company Stockholder a letter of transmittal, substantially in the form of Exhibit C hereto (the "Stockholder Letter of Transmittal") and (ii) Warrantholder, a Warrant Cancellation Agreement, in substantially the form attached hereto as Exhibit D-1 (the "Warrant Cancellation Agreement"). As a condition precedent to each holder's receipt of the Per Share Merger Consideration or Per Warrant Merger Consideration, as applicable, (A) each Company Stockholder shall deliver an executed Stockholder Letter of Transmittal, and (B) each Warrantholder shall deliver a duly executed and completed Warrant Cancellation Agreement to the Payments Administrator. Upon receipt by the Payments Administrator of the items set forth in the immediately preceding sentence (but in no event earlier than the Effective Time), such Company Stockholder or Warrantholder shall be entitled to receive in exchange therefor, the Per Share Merger Consideration or Per Warrant Merger Consideration, as applicable (as reduced at the Closing by such holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, and subject to adjustment as provided in Section 3.4), for each Company Share or Company Warrant, which amount shall be paid by the Payments Administrator on the Closing Date if deliveries from the applicable Company Stockholder or Warrantholder are received prior to the Closing Date, and shall otherwise be paid within five (5) Business Days after the date of receipt by the Payments Administrator of such deliveries from such Company Stockholder or Warrantholder, in each case by delivery of a certified or bank cashier's check or by wire transfer. No interest will be paid or accrued on any portion of the Merger Consideration

payable to Company Stockholders or Warrantholders. In the event of a transfer of ownership of Company Shares or Company Warrants that is not registered in the transfer records of the Company, payment may be made to a transferee if all documents (as determined by the Payments Administrator in its sole discretion) required to evidence and effect such transfer and to evidence that any applicable stock transfer taxes have been paid are presented to the Payments Administrator.

(h) Retention of Amounts. Notwithstanding any contrary provision set forth in this Agreement, no Company Escrow Equityholder shall be entitled to receive that portion of the Merger Consideration represented by such Company Equityholder's Escrow Pro Rata Share of the Escrow Amount or the Representative Holdback Amount until such time as such amount (or any portion thereof), if any, is distributed to such Company Escrow Equityholder pursuant to the terms and conditions of the Escrow Agreement or this Agreement, as applicable. The adoption of this Agreement and the approval of the Merger by the Requisite Stockholder Approval shall constitute approval by the Company Stockholders and the Warrantholders of the Escrow Agreement and of all of the transactions and arrangements relating thereto, including the placement of the Escrow Funds in escrow.

#### 3.3 Options.

- (a) <u>Cancellation of Unvested Options</u>. Effective at the Effective Time, each outstanding and unexercised stock option to purchase Common Stock granted under the Equity Incentive Plans (each, an "<u>Option</u>") that is not a Vested Option shall be forfeited and cancelled for no consideration.
- (b) <u>Settlement of Vested Options</u>. Effective at the Effective Time, each outstanding and unexercised Vested Option shall be cancelled and converted into the right to receive (subject to the immediately following sentence and except as otherwise provided in this <u>Section 3.3</u>, and subject to reduction at the Closing by the applicable holder's Deferred Pro Rata Share of the Deferred Merger Consideration, in respect of Company Shares underlying such holder's Vested Options) a lump sum cash payment, payable by the Payments Administrator to Non-Employee Option Holders on the Closing Date or by the Company or one of the Company Subsidiaries through the Company's or the Company Subsidiary's payroll to Employee Option Holders, in each case, in an amount equal to (i) the excess, if any, of (A) the Per Share Merger Consideration (as reduced at the Closing by the Per Share Portion of such Vested Option of the Deferred Merger Consideration) over (B) the exercise price per share of Common Stock subject to such Vested Option, <u>multiplied by</u> (ii) the total number of shares of Common Stock for which such Vested Option has not been exercised (such payment to be net of withholdings, if any, and without interest) (such amounts payable hereunder being referred to, collectively, as the "Option Payments"). Notwithstanding anything to the contrary in this Agreement, each Vested Option with an exercise price per share of Common Stock that is equal to or in excess of the Per Share Merger Consideration shall be forfeited and cancelled for no consideration as of the Effective Time.
- (c) <u>Surrender of Options; Payment for Options</u>. As soon as practicable after the date hereof, but in any event within ten (10) days of the date hereof, the Payments Administrator shall mail to each Option Holder (i) an Option Cancellation Agreement, substantially in the form attached hereto as <u>Exhibit D-2</u> (the "<u>Option Cancellation Agreement</u>"),

and (ii) instructions for use in effecting the cancellation and termination of the Options in exchange for the portion of Option Payments contemplated to be paid to such Option Holder pursuant to this Section 3.3. As a condition precedent to each Option Holder's right to receive his, her or its Option Payment, if any, such Option Holder shall deliver to the Payments Administrator an executed Option Cancellation Agreement. Upon receipt by the Payments Administrator of the items set forth in the immediately preceding sentence (but in no event earlier than the Effective Time), the Payments Administrator shall pay, subject to Section 3.1(e) with respect to the applicable holder's Deferred Pro Rata Share of Deferred Merger Consideration, to the Non-Employee Option Holders or to the Company for further distribution to the Employee Option Holders, subject to any applicable withholdings and subject to adjustment as provided in Section 3.4, the portion of Option Payments due under this Section 3.3 with respect to such Option, which payment shall be made with respect to Non-Employee Option Holders on the Closing Date and with respect to Employee Option Holders as soon as reasonably practicable on or after the Closing Date, and in any event not later than five (5) days following the Closing Date, if all deliveries from the applicable Option Holder are received prior to the Closing Date, and shall otherwise be paid as soon as reasonably practicable after the date of receipt by the Payments Administrator of such deliveries from such Option Holder.

(d) <u>Company Actions</u>. At or prior to the Closing Date, the Company shall take all actions as are necessary, including adopting resolutions of the board of directors of the Company or any committee thereof, to effectuate the treatment of the Options pursuant to this <u>Section 3.3</u>.

#### 3.4 Post-Closing Adjustments.

- Within ninety (90) days after the Closing Date, Parent shall prepare and deliver to the (a) Representative statements setting forth Parent's good faith determination of each of (i) the Closing Working Capital (the "Statement of Closing Working Capital"), (ii) the Closing Cash on Hand (the "Statement of Closing Cash on Hand"), (iii) the Closing Indebtedness (the "Statement of Closing Indebtedness"), and (iv) the Closing Transaction Expenses (the "Statement of Closing Transaction Expenses"), together with reasonable supporting documentation. The Statement of Closing Working Capital, the Statement of Closing Cash on Hand, the Statement of Closing Indebtedness, and the Statement of Closing Transaction Expenses are referred to herein, collectively, as the "Statements." The Statements shall be prepared in accordance with the definitions of Closing Working Capital, Closing Cash on Hand, Closing Indebtedness, and Closing Transaction Expenses, as applicable, in each case, as set forth in Section 1.1. Until the Statements are finally determined pursuant to this Section 3.4. Parent shall afford the Representative (and its representatives and advisors), upon prior written notice, reasonable access, during normal business hours and under reasonable circumstances, and in a manner so as not to interfere with the normal business operations of the Parent or the Surviving Corporation, to all books and records and all personnel used to prepare, or otherwise relevant to the calculation of, the Statements that are under control of or in the possession of Parent or the Surviving Corporation (or their respective representatives and advisors).
- (b) Each Statement shall be final and binding on the Parties unless the Representative shall, within thirty (30) days following the delivery of the Statements (the "<u>Dispute Period</u>"), deliver to Parent written notice of disagreement with such Statement (a "<u>Dispute</u>

Notice"), which Dispute Notice shall describe the nature of any such disagreement in reasonable detail and identify the specific items involved and the dollar amount of such disagreement. For the avoidance of doubt, amounts set forth in a Statement that are not subject to dispute in a Dispute Notice shall be deemed final, and Parent and the Representative shall jointly instruct the Escrow Agent to deliver from the Escrow Account to an account designated by the Representative and held by the Payments Administrator for further distribution in accordance with <u>Section 3.4(d)(i)–(ii)</u> the excess, if any, of the Escrow Funds over the greatest aggregate value of the items in the Statements in dispute as set forth in the Dispute Notice; provided, however, that neither Representative or Parent shall be deemed to have agreed with any items which are directly or indirectly impacted by the resolution of the items disputed in the Dispute Notice. If the Representative and Parent are unable to resolve all disagreements with respect to a Statement within thirty (30) days of receipt by Parent of a Dispute Notice with respect to such Statement, or such longer period as may be agreed by Parent and the Representative, then the Representative and Parent shall jointly select and retain KPMG LLP, or if KPMG LLP is unwilling or unable to serve in such role, an arbiter from a nationally recognized independent public accounting firm that is not the independent auditor of any of Parent, the Company or the Surviving Corporation (the Person so selected, the "Independent Accountant"). The Representative and Parent agree to jointly enter into a customary indemnity agreement with the Independent Accountant in connection with the retention of such Independent Accountant. The Independent Accountant will consider only those items and amounts set forth in the applicable Statement as to which Parent and the Representative have disagreed within the time periods and on the terms specified above and must resolve such matters in accordance with the terms and provisions of this Agreement. Each Party may furnish to the Independent Accountant such information and documents as it deems relevant, with copies of such submission and all such documents and information being concurrently given to the other Party. The Independent Accountant shall resolve each item of disagreement based solely on the supporting material provided by the Parties and not pursuant to any independent review and may not assign a value to any particular item greater than the greatest value for such item claimed by either Party or less than the lowest value for such item claimed by either Party, in each case as presented to the Independent Accountant. The Independent Accountant shall issue a detailed written report that sets forth the resolution of all items in dispute and that contains, as applicable, a final Statement of Closing Working Capital, a final Statement of Closing Cash on Hand, a final Statement of Closing Indebtedness, and/or a final Statement of Transaction Expenses. Such report shall be final and binding upon Parent and the Representative and the Company Equityholders absent manifest error. The fees and expenses of the Independent Accountant shall be borne on a proportionate basis by the Representative, on behalf of the Company Escrow Equityholders, from the Representative Holdback Amount, on the one hand, and Parent, on the other hand, based on the percentage which the portion of the contested amount not awarded to each such Person bears to the amount actually contested by such Person. Parent and the Representative shall, and Parent shall cause the Surviving Corporation to, cooperate fully with the Independent Accountant and respond on a timely basis to all requests for information or access to documents or personnel made by the Independent Accountant or by other Parties hereto, all with the intent to fairly and in good faith resolve all disputes relating to any Statement as promptly as reasonably practicable. The Independent Accountant shall act as an expert, not as an arbitrator, in resolving the disputed matters under the Dispute Notice.

(c) If the amount representing the Closing Working Capital, the Closing Cash on Hand, the Closing Indebtedness, or the Closing Transaction Expenses, in each case as reflected

in the applicable Statement as finally determined in accordance with this <u>Section 3.4</u>, differs from the Estimated Closing Working Capital, the Estimated Closing Cash on Hand, the Estimated Closing Indebtedness, or the Estimated Closing Transaction Expenses, respectively, as the case may be, the Closing Merger Consideration shall be adjusted on a dollar for dollar basis by the amount of such difference such that:

- (i) if the Closing Working Capital as reflected in the Statement of Closing Working Capital differs from the Estimated Closing Working Capital, then the Closing Merger Consideration shall be increased by the difference between calculating the Estimated Working Capital Surplus based on the Closing Working Capital instead of the Estimated Closing Working Capital, and shall be decreased by the difference between calculating the Estimated Working Capital Shortfall based on the closing Working Capital instead of the Estimated Closing Working Capital (provided, in each of the foregoing cases, that if the difference between the Closing Working Capital and the Closing Working Capital Target is less than the Collar Amount, then there shall be no adjustment therefrom);
- (ii) if the Closing Cash on Hand as reflected in the Statement of Closing Cash on Hand is greater than the Estimated Closing Cash on Hand, then the Closing Merger Consideration shall be increased by the amount of such excess, and if the Closing Cash on Hand as reflected in the Statement of Closing Cash on Hand is less than the Estimated Closing Cash on Hand, then the Closing Merger Consideration shall be decreased by the amount of such shortfall;
- (iii) if the Closing Indebtedness as reflected in the Statement of Closing Indebtedness is greater than the Estimated Closing Indebtedness, then the Closing Merger Consideration shall be decreased by the amount of such excess, and if the Closing Indebtedness as reflected in the Statement of Closing Indebtedness is less than the Estimated Closing Indebtedness, then the Closing Merger Consideration shall be increased by the amount of such shortfall; and
- (iv) if the Closing Transaction Expenses as reflected in the Statement of Closing Transaction Expenses are greater than the Estimated Closing Transaction Expenses, then the Closing Merger Consideration shall be decreased by the amount of such excess, and if the Closing Transaction Expenses as reflected in the Statement of Closing Transaction Expenses are less than the Estimated Closing Transaction Expenses, then the Closing Merger Consideration shall be increased by the amount of such shortfall.

If the adjustments, if any, under this <u>Section 3.4</u> result in an aggregate reduction in the Closing Merger Consideration, the Representative, acting on behalf of the Company Equityholders, and Parent shall deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to release to Parent (by wire transfer to an account designated in writing by Parent) from the Escrow Funds the amount of such reduction within five (5) Business Days after the final determination of the adjustments. Conversely, if such adjustments result in an aggregate increase in the Closing Merger Consideration, Parent shall make a payment equal to the lesser of (x) the amount of such increase and (y) \$500,000 to an account designated by the Representative and held by the Payments

Administrator for further distribution in accordance with  $\underline{Section\ 3.4(\underline{d})(\underline{i})\underline{-(\underline{i}\underline{i})}}$  within five (5) Business Days after the final determination of the adjustments. Notwithstanding anything to the contrary contained in this  $\underline{Section\ 3.4}$ , the Parties acknowledge and agree that in no event will the Representative (or any other Person) be obligated to make any payment to Parent in respect of any adjustments pursuant to this  $\underline{Section\ 3.4}$  other than from, or in an aggregate amount in excess of, the Escrow Funds.

- (d) <u>Escrow Release</u>. On the date that is five (5) Business Days following the final determination of Closing Working Capital, the Closing Cash on Hand, the Closing Indebtedness, and the Closing Transaction Expenses pursuant to this <u>Section 3.4</u>, to the extent funds are available in the Escrow Account after the final adjustment payment(s) in accordance with <u>Section 3.4(c)</u>, Parent and the Representative shall deliver a joint written instruction to the Escrow Agent instructing the Escrow Agent to release such funds to an account designated by the Representative and held by the Payments Administrator for further distribution to:
  - (i) each Company Stockholder (other than any EMI Stockholder) in accordance with such holder's Escrow Pro Rata Share; and
    - (ii) each Warrantholder in accordance with such holder's Escrow Pro Rata Share.
- 3.5 <u>Effect of Merger on the Capital Stock of Merger Sub</u>. At the Effective Time, each share of Merger Sub's common stock, par value \$0.001 per share, that is issued and outstanding immediately prior to the Effective Time, shall, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the Company Stockholders, be converted automatically into and exchanged for one share of common stock, par value \$0.001 per share, of the Surviving Corporation.
- Appraisal Rights. Notwithstanding anything in this Agreement to the contrary, Company Shares that are issued and outstanding immediately prior to the Effective Time and are held by Company Stockholders who have not voted in favor of the Merger, consented thereto in writing or otherwise contractually waived their rights to appraisal and who have complied with all of the relevant provisions of the DGCL (the "Dissenting Shares," and the holders thereof the "Dissenting Stockholders") shall not be converted into or be exchangeable for the right to receive the Per Share Merger Consideration (as reduced at the Closing by the applicable holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, and subject to adjustment as provided in <u>Section 3.4</u>) in accordance with <u>Sections 3.1</u> and <u>3.2</u>, unless and until such Company Stockholders shall have failed to perfect or shall have effectively withdrawn or lost their rights to appraisal under the DGCL. The Company shall give Parent (i) prompt notice of any written demands for appraisal of any Company Shares, any objection to the Merger, attempted withdrawals of such demands and any other instruments served pursuant to the DGCL and received by the Company relating to stockholders' rights of appraisal, and (ii) the opportunity to participate in all negotiations and proceedings with respect to demands for appraisal under the DGCL. Neither the Company nor the Surviving Corporation shall, except with the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed), voluntarily make any payment with respect to, or settle or offer to settle, any such demand for payment. If any Dissenting Stockholder shall fail to perfect or shall have effectively withdrawn or lost the right

to dissent, then (1) as of the occurrence of such event, such holder's Dissenting Shares shall cease to be Dissenting Shares and shall be converted into and represent the right to receive the Per Share Merger Consideration (as reduced at the Closing by such holder's Escrow Pro Rata Share of each of the Escrow Amount and the Representative Holdback Amount, and subject to adjustment as provided in <u>Section 3.4</u>) in accordance with <u>Sections 3.1</u> and <u>3.2</u>, and (2) promptly following the occurrence of such event, Parent shall remit to the Payments Administrator, the portion of the Merger Consideration to which such holder is entitled. Any payments to be made in respect of Dissenting Shares will be made by Parent and/or the Surviving Corporation.

- 3.7 <u>Withholding Rights</u>. Parent, the Company, the Payments Administrator and their respective Affiliates shall be entitled to deduct and withhold from any amount otherwise payable to any person pursuant to this Agreement or any ancillary agreements hereto, such amounts that are required to be deducted and withheld with respect to the making of any such payment under any provisions of federal, state, local or foreign Law, including as a result of (i) the compensatory nature of such payments (such as payments to Employee Option Holders), (ii) a change in Law after the date hereof or (iii) the Company's failure to provide the certificate in accordance with Section 9.1(e). To the extent that such amounts are so withheld and paid over to the proper Governmental Authority, such withheld and deducted amounts will be treated for all purposes of this Agreement as having been paid to the recipient in respect of which such deduction or withholding was made.
- 3.8 <u>Payments Administrator</u>. The Payments Administrator shall act as payments administrator hereunder for the payment and delivery of amounts payable in exchange for Company Shares, Company Warrants and Options due hereunder. The Payments Administrator shall hold the amounts received from Parent and payable to the Company Equityholders pursuant to (i) this Agreement and (ii) the Payments Agreement, in trust for the benefit of such Company Equityholders. Any funds received by the Payments Administrator pursuant to this Agreement will not be used for any purpose except as expressly provided in the Payments Agreement.

# ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Contemporaneously with the execution and delivery of this Agreement, the Company is delivering to Parent and Merger Sub a disclosure schedule with numbered sections corresponding to the relevant sections in this Agreement (the "Company Disclosure Schedules"). Except as disclosed in the Company Disclosure Schedules, the Company makes the following representations and warranties to Parent and Merger Sub on the date hereof and as of the Closing Date as follows (except where a representation or warranty is made as of a specified date, then as of such date):

4.1 <u>Organization, Qualification, Power and Authority</u>. Each of the Acquired Companies is an entity validly existing and in good standing under the Laws of the jurisdiction of its incorporation or formation, in each jurisdiction in which the character of the properties owned, leased or operated by it makes such licensing or qualification necessary. Each of the Acquired Companies is qualified to conduct business and in good standing under the Laws of each jurisdiction where such qualification is required, except where the lack of such qualification or the

failure to be in good standing would not have a Material Adverse Effect. Each of the Acquired Companies has the requisite power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it, except as would not have a Material Adverse Effect. The Company has heretofore made available to Parent true and complete copies of the Company's Organizational Documents and the Organizational Documents of each of the Company Subsidiaries as currently in effect (including all amendments made thereto at any time on or before the date hereof) and none of the Acquired Companies is in default under or in violation of any provision thereunder. With respect to Wandera CZ s.r.o. (the "Czech Company"), all documents required to be deposited in the Collection of Deeds kept by the Czech registry court have been duly submitted (including, without limitation, any missing financial statements or reports on relations).

Authorization; Enforceability. The Company has requisite corporate power and authority to execute and deliver this Agreement and each other agreement contemplated hereby to which the Company is a party and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement by the Company and each other agreement contemplated hereby to which the Company is a party and the consummation by the Company of the Merger and the other transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action, and no other corporate proceedings on the part of the Company and no stockholder votes are necessary to authorize this Agreement or the other agreements contemplated hereby to which the Company is a party or to consummate the Merger and the other transactions contemplated hereby and thereby, other than the Requisite Stockholder Approval, and the filing and recordation of the Certificate of Merger. The Requisite Stockholder Approval is the only vote of the holders of any class or series of Company Shares that is necessary pursuant to applicable law, the certificate of incorporation of the Company or the Bylaws or otherwise to adopt this Agreement and consummate the Merger. This Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by the other Parties hereto, constitutes the valid and legally binding obligation of the Company, enforceable in accordance with its terms and conditions, subject to Laws of general application relating to public policy, bankruptcy, insolvency and the relief of debtors and rules of Law governing specific performance, injunctive relief and other equitable remedies.

## 4.3 <u>Capitalization</u>.

- (a) The entire authorized capital stock of the Company consists, as of the date hereof, of:
- (i) 34,000,000 shares of Common Stock, \$0.0001 par value per share, and 9,246,000 of which are issued and outstanding. All of the outstanding shares of Common Stock have been duly authorized, are fully paid and nonassessable and were issued in compliance with all applicable federal and state securities laws.
- (ii) 19,060,886 shares of Preferred Stock, \$0.0001 par value per share, (a) of which 1,466,658 shares have been designated Series Seed Stock and 1,466,658 of which are issued and outstanding, (b) of which 6,343,350 shares have been designated Series A Preferred Stock and 6,343,350 of which are issued and outstanding, (c) of which 1,415,094 shares have been designated Series A-1

Preferred Stock and 1,415,094 of which are issued and outstanding, (d) of which 3,835,784 shares have been designated Series B Preferred Stock and 3,774,510 of which are issued and outstanding, and (e) 6,000,000 shares have been designated Series C Preferred Stock and 5,814,120 of which are issued and outstanding. The rights, privileges and preferences of the Preferred Stock are as stated in the Restated Certificate and as provided by the DGCL.

- (iii) 61,274 shares of Series B Preferred Stock are issuable under the Company Warrants.
- (iv) 4,421,709 shares of Common Stock are issuable under the Options.
- (v) The Company Shares and Company Warrants are held of record and beneficially by the Persons in the amounts set forth on <u>Schedule 4.3(a)(v)</u>. No dividends or other distributions with respect to any shares of Company Shares or any equity securities of any of the Company Subsidiaries have ever been made or declared, and none have accrued.
- The Company has reserved 5,230,000 shares of Common Stock for issuance to directors, employees, consultants and other service providers of the Company pursuant to the Equity Incentive Plans, of which 62,291 shares of Common Stock remain available for issuance. Schedule 4.3(b) sets forth, with respect to each Option, (i) the name of the Option Holder, (ii) the maximum number of shares of Common Stock that may be issued upon exercise of such Option, (iii) the grant date, (iv) the exercise price, (v) the vesting schedule and vesting conditions, to the extent such Option is unvested as of the date of this Agreement, (vi) the vested status as of the date of this Agreement, (vii) the expiration date, and (viii) whether such Option is intended to constitute an "incentive stock option" (within the meaning of Section 422 of the Code). The Company has furnished to Parent complete copies of the Equity Incentive Plans and all forms of agreements used thereunder. Except as set forth on Schedule 4.3(b), there are no outstanding or authorized options, restricted stock, restricted stock units, warrants, purchase rights, subscription rights, conversion rights, exchange rights or other contracts or commitments that would require the Company to issue, sell or otherwise cause to become outstanding any of the Company Shares. Except as set forth on Schedule 4.3(b), there are no outstanding or authorized equity appreciation, phantom interest, profit participation, equity-based rights or other similar rights with respect to the Company. Each Option is in compliance in all material respects with all applicable Laws, and the per share exercise price of each Option is equal to or greater than the fair market value of the underlying Common Stock on the date of grant within the meaning of Section 409A of the Code and the Treasury Regulations and official guidance promulgated thereunder. Each of the Company Shares is uncertificated.
- (c) <u>Schedule 4.3(c)</u> sets forth a complete and accurate list of (i) each Warrantholder, (ii) each Company Warrant, (iii) the exercise price, (iv) the date of grant, (v) the vesting schedule, (vi) the current vesting status, and (vii) the number of shares and class and series of Company Shares subject to each such Company Warrant.
- (d) The Company's Subsidiaries, as of the date hereof, are listed on <u>Schedule 4.3(d)</u> (the "<u>Company Subsidiaries</u>"). All of the outstanding shares of capital stock (or equivalent

equity interests of entities other than corporations) of each of the Company Subsidiaries are owned of record and beneficially, directly or indirectly, by the Company free and clear of all Liens other than Permitted Liens. There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights or other contracts or commitments that would require any Company Subsidiary to issue, sell or otherwise cause to become outstanding any of its capital stock (or equivalent equity interests of entities other than corporations). There are no outstanding or authorized stock appreciation, phantom stock, profit participation, silent partnership or similar rights with respect to any Company Subsidiary. All outstanding equity securities of the Company Subsidiaries have been duly authorized and are fully paid and non-assessable.

- (e) There are no outstanding obligations for the provision of any payment into the Czech Company's equity, and the Czech Company's general meeting has not adopted any decision on the increase or decrease of the Czech Company's registered capital. The Czech Company is not obliged to return any payment into the Czech Company's equity. Any and all shares in profit and/or other Czech Company's funds have been paid to its shareholder in accordance with applicable Law. There are no unsettled advance payments in share in profit and/or other funds paid by the Czech Company.
- Non-contravention. Neither the execution and delivery of this Agreement or any other agreement contemplated hereby to which the Company is a party, nor the consummation of the Merger or the other transactions contemplated hereby or thereby, will (a) violate any Laws to which any of the Acquired Companies is subject, (b) assuming the Requisite Stockholder Approval is obtained, violate, conflict with, or require any notice under any provision of the Organizational Documents of any of the Acquired Companies, or (c) except as set forth on Schedule 4.4, conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any Person the right to accelerate, terminate, modify or cancel, require any notice or consent under, create in any party the right to payment (including, without limitation, any change of control, severance, or similar payment), or result in the imposition of any Lien (other than Permitted Liens) upon any of the assets of the Acquired Companies under, any Material Contract, except, in the case of clauses (a) and (c) of this <u>Section 4.4</u>, as would not have, individually or in the aggregate, reasonably be expected to be material to the Acquired Companies, taken as a whole. Except for (i) the filing of the Certificate of Merger under the DGCL, (ii) the pre-merger notification requirements of the HSR Act, (iii) filings that may be required as a result of facts specific to Parent or Merger Sub, and (iv) where the failure to obtain such consent, waiver, approval, authorization, order or permit, or to make such declaration, filing, registrations or notification would not, individually or in the aggregate, reasonably be expected to be material to the Acquired Companies, taken as a whole, none of the Acquired Companies needs to give any notice to, make any filing with or obtain any authorization, consent or approval of any Governmental Authority in order for the Company to consummate the transactions contemplated by this Agreement or the other agreements contemplated hereby to which the Company is a party.
- 4.5 <u>Brokers' Fees</u>. <u>Schedule 4.5</u> sets forth any liability or obligation of the Acquired Companies to pay any fees or commissions to any broker, finder or agent or similar compensation arrangements with a third party with respect to the transactions contemplated by this Agreement.

#### 4.6 Financial Statements.

- (a) The audited consolidated financial statements and the notes thereto of the Acquired Companies (the "Audited Financial Statements"), copies of which have been made available to Parent, present fairly, in all material respects, the audited consolidated balance sheet of the Acquired Companies as at December 31, 2018, and December 31, 2019, and the related audited consolidated statements of operations and comprehensive income, changes in stockholders' equity and cash flows, in conformity with FRS 102 applied on a consistent basis during the periods involved, subject to such exceptions as may be indicated in the notes thereto. The Acquired Companies' accounts are accurate, complete and conclusive, and the Audited Financial Statements provide a true and fair view of the state of the Company's affairs and of its revenue, operating loss and cash flows for the years presented.
- (b) The unaudited financial statements and the notes thereto of the Acquired Companies (the "Interim Financial Statements" and together with the Audited Financial Statements the "Financial Statements"), copies of which have been made available to Parent, present fairly, in all material respects, the unaudited consolidated balance sheet of the Acquired Companies as at December 31, 2020, and March 31, 2021 (such 2021 balance sheet, the "Balance Sheet") and the related unaudited consolidated statements of operations and comprehensive income, changes in stockholders' equity and cash flows, in conformity with FRS 102 applied on a consistent basis during the periods involved, subject to the absence of footnote disclosure, none of which would, individually or in the aggregate, be material, and normal and recurring year-end adjustments, which if presented would not differ materially from those presented in the Audited Financial Statements.
- (c) The Acquired Companies have no liabilities, except (i) those which are reflected or reserved against in the Balance Sheet, (ii) to perform in accordance with their terms, any Contract to which any Acquired Company is a party other than liabilities arising from any material breach of any such Contract, (iii) with respect to any Transaction Expenses or liabilities arising under this Agreement, (iv) those which have been incurred in the Ordinary Course of Business since the date of the Balance Sheet and that do not arise from any material breach of a Contract, infringement, misappropriation, or violation of Law, (v) those that have been discharged or paid off since the date of the Balance Sheet or (vi) those that would not be material to the Acquired Companies, taken as a whole.
- (d) The Acquired Companies maintain, and have maintained for periods reflected in the Financial Statements, a system of internal accounting controls sufficient to provide commercially reasonable assurance that: (i) transactions are executed in accordance with management's general specific authorizations; (ii) transactions are recorded as necessary to permit preparation of financial statements in accordance with FRS 102 and to maintain asset accountability; (iii) access to assets is permitted only in accordance with management's general or specific authorization; (iv) there has not been any significant deficiency or weakness in the system of internal accounting controls used by the Acquired Companies; and (v) there has not been any Fraud or wrongdoing by any employee of the Acquired Companies who has or had a role in the preparation of financial statements or the internal accounting controls used by the Acquired Companies.

- (e) <u>Schedule 4.6(e)</u> accurately lists all Indebtedness of the Acquired Companies for borrowed money as of the date hereof, including, for each such item of Indebtedness the agreement(s) governing the Indebtedness and, if applicable, the interest rate, maturity date and any assets or properties securing such Indebtedness.
- (f) The Financial Statements are consistent with, and derived from, the books and records of the Acquired Companies, each of which is true and correct in all material respects. No written notice that any of the books and records of any of the Acquired Companies is inaccurate or incorrect has been received by any of the Acquired Companies.
- 4.7 <u>Subsequent Events</u>. Except as set forth on <u>Schedule 4.7</u>, from the period beginning on November 1, 2020 through the date of this Agreement, the Acquired Companies have conducted their respective businesses in the Ordinary Course of Business and, except as required or contemplated by this Agreement there has not been any:
  - (a) Material Adverse Effect;
- (b) incident of damage, destruction or loss of any property owned by the Acquired Companies or used in the operation of the Business that is not covered by insurance and having a replacement cost or fair market value in excess of \$50,000 per occurrence or \$250,000 in the aggregate (whether or not covered by insurance);
- (c) voluntary or involuntary sale, transfer, surrender, abandonment, waiver, release or other disposition of any kind by any of the Acquired Companies of any right, power, claim, debt, asset or property (having a replacement cost or fair market value in excess of \$50,000 individually or \$250,000 in the aggregate), except the sale of inventory and collection of accounts in the Ordinary Course of Business;
- (d) abandonment or lapse of any Owned Intellectual Property (except for the abandonment or lapse of Owned Intellectual Property in the Ordinary Course of Business);
- (e) sale or license of, or Lien on, any Owned Intellectual Property, except for non-exclusive licenses of Intellectual Property granted to customers in the Ordinary Course of Business;
- (f) loan or advance by any of the Acquired Companies to any Person, other than advances to employees, officers or directors for business or travel expenses to be incurred in the Ordinary Course of Business or transactions with customers on credit in the Ordinary Course of Business;
- (g) declaration, setting aside or payment of any dividend or other distribution in respect of any of the Acquired Companies' equity interests or any direct or indirect redemption, purchase or other acquisition of such equity interests by any Acquired Company, or the payment of principal or interest on any note, bond, debt instrument or debt to any Affiliate of the Acquired Companies, other than repurchases of equity interests from former employees in the Ordinary Course of Business;

- (h) reclassification, combination, split, subdivision or redemption, or purchase or acquisition, directly or indirectly, of any Acquired Company's capital stock or other equity or ownership interest or any other change with respect to its capital structure;
- (i) change in accounting principles, methods or practices (including any change in depreciation or amortization policies or rates) utilized by the Acquired Companies, except insofar as was required by a change in FRS 102;
- (j) capital expenditures or commitments therefor by the Acquired Companies in excess of \$250,000 individually, other than capital expenditures or commitments relating to equipment purchases by any Acquired Company arising in the Ordinary Course of Business;
- (k) except as otherwise required by Law, (i) increase or promises to increase the compensation or benefits payable or provided, or to become payable or provided, whether conditionally or otherwise, to any current or former Company Employee, except (A) in connection with any hiring or engagement of any Company Employee to the extent the annual total compensation opportunity for such employee is less than \$125,000 or (B) for increases in total compensation for any employee to the extent that such increases in total compensation are no greater than 5% for any such employee, (ii) grant or announcement of any cash or equity or equity-based incentive awards, bonuses, transaction, retention, severance or other compensation or benefits to any current or former Company Employee, other than grants of ordinary course cash amounts or Options to newly hired or engaged Company Employees consistent with past practice, (iii) establishment, adoption, amendment, modification or termination of any Employee Benefit Plan or PEO Plan (or any other compensation or benefit plan or arrangement that would be an Employee Benefit Plan or PEO Plan if in effect on the date hereof) other than as permitted under clauses (i) and (ii) of this Section 4.7(k), or (iv) acceleration of the time of payment, vesting or funding of any compensation or increase in the amount of compensation or benefits (whether in cash, property or the vesting of property) provided under any Employee Benefit Plan or otherwise;
- (l) acquisition, reorganization, recapitalization or agreement of any Acquired Company to acquire in any manner (whether by merger or purchase of equity or assets or otherwise) any Person or material assets, other than the acquisition of assets in the Ordinary Course of Business;
- (m) incurrence of any Indebtedness or causing the issuance of any letters of credit or guarantees by any Acquired Company;
- (n) commencement or settlement of any Legal Proceeding in an amount less than \$50,000 (net of amounts covered by insurance) in any single instance or \$100,000 (net of amounts covered by insurance) in the aggregate;
- (o) (i) negotiation, modification, extension, or entry into any Labor Agreement or (ii) recognition or certification of any labor union, labor organization, works council, or group of employees as the bargaining representative for any employees of any Acquired Company;
- (p) termination (other than for "cause"), furloughing, or temporary layoff of any Company Employee or independent contractor with annual base compensation in excess of \$125,000;

- (q) implementation or announcement of any employee layoffs, plant closings, reductions in force, furloughs, temporary layoffs, salary or wage reductions, work schedule changes or other such actions that could implicate the WARN Act;
- (r) waiver or release of any noncompetition, nonsolicitation, nondisclosure, noninterference, nondisparagement or other restrictive covenant obligation of any current or former employee or independent contractor with annual base compensation in excess of \$125,000;
- (s) adoption of a plan or agreement of, or resolutions providing for or authorizing, a complete or partial merger or consolidation with or into any other Person or dissolution, restructuring, recapitalization, complete or partial liquidation or other reorganization;
  - (t) amendment or authorization of the amendment of any of its Organizational Documents;
- (u) conduct of its cash management customs and practices other than in the Ordinary Course of Business (including with respect to collection of accounts receivable, repairs and maintenance, payment of accounts payable, accrued expenses or other liabilities, levels of capital expenditures, pricing and credit practices and operation of cash management practices generally);
- (v) application for or receipt of any government assistance (including, but not limited to, loans, benefits, rights, or amounts) pursuant to the CARES Act, including loans under the Paycheck Protection Program pursuant to Section 1102 of the CARES Act guaranteed by the Small Business Administration;
  - (w) entry, amendment to or termination of any Material Contract;
- (x) filing of any Tax election or change to any Tax election, change to any method of accounting for Tax purposes, filing of any amended Tax Return, entry into any closing agreement related to Taxes, settlement of any Tax claim or assessment, request or consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment (other than pursuant to extensions to file Tax Returns obtained in the Ordinary Course of Business), failure to pay any Tax that becomes due and payable (including any estimated tax payments), or filing of any Tax Return in a manner inconsistent with past practice;
- (y) application for, or conduct of any business that would require the Acquired Companies to hold, any additional Permits not held by the Acquired Companies as at the date hereof;
- (z) lapse of any existing insurance policy relating to the Business or the assets of the Acquired Companies; or
  - (aa) resolution or undertaking to take any of the foregoing actions.
- 4.8 <u>Legal Compliance</u>. As of the date of this Agreement and for a period of three (3) years prior to the date hereof, the Acquired Companies hold all permits, licenses, approvals, certificates, registrations, consents, determinations and other authorizations of and from

all Governmental Authorities necessary for the lawful conduct of the Business (the "Permits"), each of the Permits is in full force and effect and the Acquired Companies are in compliance with the Permits. The Acquired Companies are and have been in compliance in all material respects with all applicable Laws. Each of the Acquired Company's products is in compliance with all Laws applicable to such product (including Laws that such product is designed to address), and no Governmental Authority has questioned or challenged the compliance of the Acquired Company's products with such Laws in writing. The Company has in place reasonable procedures to monitor changes in Laws that the Acquired Company's products are designed to address and to provide product updates to the extent required under applicable customer Contracts. The Czech Company is and has been in compliance with any obligations ensuing from the terms of any public aid it has applied for or been granted, and any and all information or documents it has submitted in connection with any such public aid are and have true, correct, full and not misleading in any respect.

#### 4.9 <u>Tax Matters</u>.

- (a) The Acquired Companies have timely filed (taking into account all valid extensions) all Tax Returns that they were required to file and all such Tax Returns are true, correct and complete in all material respects and were prepared in substantial compliance with applicable Law. All Taxes due and owing by the Acquired Company (whether or not shown as due and owing on a Tax Return) have been timely paid in full. Taxes owed but not payable on the relevant balance sheet date were accounted for as reserves in the accounts.
- (b) There are no pending audits, examinations, proceedings, actions, suits or inquiries of any of the foregoing against or with respect to any of the Acquired Companies regarding Taxes or Tax Returns filed by the Acquired Companies, and no action, suit, proceeding or audit has been threatened in writing against or with respect to any of the Acquired Companies regarding Taxes or Tax Returns.
- (c) None of the Acquired Companies has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to an assessment or deficiency of Taxes (other than pursuant to extensions to file Tax Returns obtained in the Ordinary Course of Business), which waiver or extension is still in effect.
- (d) No Acquired Company (i) is a party to or bound by any Tax sharing, allocation, indemnity or similar agreement or arrangement pursuant to which it will have any obligation to make any payments after the Closing, (ii) has been a member of any consolidated, combined, affiliated or unitary group of corporations for any Tax purposes other than a group of which the Company is the common parent or (iii) has any liability for Taxes of any person other than an Acquired Company under Treasury Regulations Section 1.1502-6 (or any corresponding or similar provision of state, local or foreign Law), as a transferee or successor, by contract, or otherwise, in each case other than (1) leases, credit agreements or other commercial agreements entered into in the Ordinary Course of Business the primary purpose of which does not relate to Taxes or (2) agreements solely between or among the Acquired Companies.

- (e) The Company is not, and has not been during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code, a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code.
- (f) There are no Liens (other than Permitted Liens) for Taxes on any of the assets of the Acquired Companies.
- (g) None of the Acquired Companies have engaged in a trade or business, had a permanent establishment, or otherwise been subject to taxation in any country other than the country of its formation.
- (h) No written claim has ever been made by a taxing authority in a jurisdiction where any Acquired Company does not file Tax Returns that such Person is or may be subject to taxation by such jurisdiction.
- (i) All Taxes required to have been withheld and paid by the Acquired Companies in connection with any amounts paid or owing to any equityholder, employee, creditor, independent contractor, or other third party, have been properly withheld and timely paid in full to the applicable Governmental Authority.
- (j) No Acquired Company (or Parent as a result of the transactions contemplated by this Agreement) will be required to include any item of income in, or exclude any item of deduction from, taxable income for any period (or portion thereof) beginning after the Closing Date as a result of any (i) change in or use of an improper method of accounting for a taxable period (or portion thereof) ending on or prior to the Closing Date; (ii) agreement with any Governmental Authority entered into prior to the Closing; (iii) installment sale or open transaction disposition made on or prior to the Closing Date; (iv) prepaid amount received or deferred revenue accrued on or prior to the Closing Date; (v) intercompany transaction entered into prior to the Closing or excess loss account as in existence immediately prior to the Closing as described in Section 1502 of the Code (or any corresponding or similar provision of state, local or non-U.S. Law), (vi) "global intangible low-taxed income" (within the meaning of Section 951A of the Code (or any corresponding or similar provision of state, local or foreign Law)) attributable to a period (or portion thereof) ending on or prior to the Closing Date or (vii) application of Section 965 of the Code. No Acquired Company has made an election pursuant to Section 965(h) of the Code.
- (k) No Acquired Company has participated in any "listed transaction" or "reportable transaction," within the meaning of Section 6707A(c) of the Code or U.S. Treasury Regulations Section 1.6011-4(b)(2), or any transaction requiring disclosure under a corresponding or similar provision of state, local, or foreign Law.
- (l) Within the last two (2) years, no Acquired Company was a distributing nor a controlled corporation in a transaction intended to be governed in whole or in part by Section 355 or Section 361 of the Code (or any corresponding or similar provision of state, local or foreign Law).
- (m) The Company is and always has been treated as a C corporation for U.S. federal income Tax purposes. Schedule 4.9(m) lists out the Tax classification of each of the Company's Subsidiaries for U.S. federal income Tax purposes.

- (n) The method of allocating income, deductions, expenses and receipts among the offices or permanent establishments of any of the Company and its Subsidiaries materially complies with the principles set forth in Section 482 of the Code and Treasury Regulations promulgated thereunder (or any corresponding or similar provisions of state, local or foreign Tax Law) and any other applicable Laws on transfer pricing, and all arrangements between the offices or permanent establishments of any of the Acquired Subsidiaries materially complies with all requirements (including documentary, retention and filing requirements) of any such Laws.
- (o) No Acquired Company has (i) deferred any amount of the employer's share of any "applicable employment taxes" under Section 2302 of the CARES Act, (ii) received any credits under Sections 7001 through 7005 of the FFCRA or Section 2301 of the CARES Act, or (iii) sought a covered loan under paragraph (36) of Section 7(a) of the Small Business Act (15 U.S.C. 636(a)), as added by Section 1102 of the CARES Act.
- (p) No Acquired Company is party to any joint venture, partnership or other arrangement or contract that would reasonably be treated as a partnership for U.S. federal income Tax purposes.
- (q) No currently effective power of attorney with respect to any Taxes of the Acquired Companies has been executed or filed with any Governmental Authority.
- (r) No Acquired Company is a party to any Tax exemption, Tax holiday or other Tax reduction agreement with any Governmental Authority.
- (s) No Acquired Company (i) has ever been a United States shareholder (within the meaning of Section 951(b) of the Code) of a "controlled foreign corporation" (within the meaning of Section 957 of the Code) or (ii) has ever held an interest in a "passive foreign investment company" (within the meaning of Section 1297 of the Code).
- (t) The Czech Company is not a so-called "unreliable taxpayer" within the meaning of the Act No. 235/2004 Coll., on value added tax, as amended.
  - 4.10 <u>Real Property</u>.
  - (a) None of the Acquired Companies owns any real property.
- (b) <u>Schedule 4.10(b)</u> sets forth the addresses of all real property used in the Business leased or subleased by any Acquired Company, or pursuant to which the Acquired Company has any right to use or occupy any land, buildings, structures, improvements, fixtures or other interest in real property on the date hereof (the "<u>Leased Real Property</u>") and a list of all Leases in effect as of the date hereof. The Company has made available to Parent a copy of each such Lease. With respect to each Lease:
  - (i) such Lease is the valid and binding obligation of the applicable Acquired Company party thereto, enforceable in accordance with its terms subject to proper authorization and execution of such Lease by the other party thereto and the Laws of general application relating to public policy, bankruptcy, insolvency

and the relief of debtors and rules of Law governing specific performance, injunctive relief and other equitable remedies;

- (ii) neither the applicable Acquired Company party thereto nor, to the Company's Knowledge, any other party to such Lease is in breach or default under such Lease as of the date hereof, and no event has occurred or circumstance exists which, with the delivery of notice, the passage of time or both, would constitute such a breach or default, or permit the termination, modification or acceleration of rent under such Lease;
- (iii) the transactions contemplated by this Agreement do not require the consent of any other party to such Lease, will not result in a breach of or default under such Lease, or otherwise cause such Lease to cease to be legal, valid, binding, enforceable and in full force and effect on identical terms following the Closing;
- (iv) the other party to such Lease is not an affiliate of, and otherwise does not have any economic interest in, the Company or any Acquired Company; and
- (v) no Acquired Company has subleased, licensed or otherwise granted any Person the right to use or occupy such property subject to such Lease or any portion thereof.
- 4.11 Personal Property. Except as disposed of in the Ordinary Course of Business, the Acquired Companies have good and marketable title to, or a valid leasehold interest in, all material items of tangible personal property reflected on the Financial Statements as owned or leased by such Acquired Company, free and clear of any Liens other than Permitted Liens. The Acquired Companies' buildings, improvements, fixtures, machinery, equipment and other tangible assets (whether owned or leased) (i) are, except for ordinary wear and tear, in good condition and repair in all material respects, and (ii) (A) constitute all of the property and assets (tangible and intangible) used or held for use in the conduct of the Business by the Acquired Companies as conducted as of the date of this Agreement, and (B) utilized by the Acquired Companies to generate the financial results reflected in the Financial Statements. The Acquired Companies shall be able to use all such properties and assets, and take advantage of all such services and rights, in the same manner after the Closing as such properties and assets were used, or such services or other rights were taken advantage of, by the Acquired Companies before the Closing.

# 4.12 <u>Intellectual Property</u>.

(a) Schedule 4.12(a) sets forth a complete and accurate list of each item of Intellectual Property registered to, applied for by, or issued to any Acquired Company under the authority of any Governmental Authority or domain name registrar as of the date hereof ("Scheduled Intellectual Property"). All Scheduled Intellectual Property is subsisting, and to the Company's Knowledge, valid and enforceable and all necessary registration, maintenance, renewal, and other relevant filing fees due through the date hereof in connection with maintaining Scheduled Intellectual Property in full force and effect have been timely paid. The Acquired Companies (i) exclusively own all right, title, and interest in and to the Owned Intellectual Property

(provided that, with respect to Intellectual Property in Software and other authors' works created by the Czech Company that are not assignable under applicable Law, the Acquired Companies have the right to exercise all authors' proprietary rights to such Intellectual Property), and (ii) validly license or otherwise possess valid rights to use all other Intellectual Property used in or necessary for the conduct of the Business as currently conducted ((i) and (ii), collectively, the "Acquired Company Intellectual Property"), in each case, free and clear of all Liens, other than Permitted Liens. The Acquired Company Intellectual Property shall be available for use by Parent and the Acquired Companies immediately after the Closing Date on identical terms and conditions to those under which the Business and the Acquired Companies owned or used the Acquired Company Intellectual Property immediately prior to the Closing Date.

- (b) <u>Schedule 4.12(b)</u> identifies all of the Acquired Company Products. The Acquired Companies possess all source code and other documentation and materials necessary to compile and operate the Acquired Company Products, and the Acquired Companies have not disclosed, delivered, licensed or otherwise made available, and, except as listed in Schedule 4.13(n), the Acquired Companies do not have a duty or obligation (whether present, contingent or otherwise) to disclose, deliver, license or otherwise make available, any source code for any Acquired Company Products to any Person, other than employees, agents, consultants and contractors engaged to provide services to the Acquired Companies with respect to the Acquired Company Products and pursuant to a written confidentiality agreement.
- (c) None of the Acquired Companies has, during the past six (6) years, infringed, misappropriated, or otherwise violated, or is now infringing misappropriating, or otherwise violating, the Intellectual Property rights of any Person. During the past six (6) years, the Acquired Companies have not received any communication, and there is currently no claim pending or, to the Company's Knowledge, threatened in writing against any of the Acquired Companies, with respect to the alleged infringement, misappropriation, or other violation by any of the Acquired Companies of any Intellectual Property rights of any Person (including any unsolicited demand or request from any Person to license any Intellectual Property) (and none of the Acquired Companies have received any written notices or requests for indemnification related to the foregoing). There is no currently pending claim by any of the Acquired Companies against a third party with respect to the alleged infringement, misappropriation or other violation of Owned Intellectual Property and, to the Company's Knowledge, no third party is infringing, misappropriating, or otherwise violating the Acquired Company Intellectual Property.
- (d) The Acquired Companies do not use and have not used any Open Source Software or any modification or derivative thereof in a manner that (i) would grant any Person any rights to or immunities under any of the Owned Intellectual Property, (ii) requires the Acquired Companies to disclose or distribute the source code to any of the Acquired Company Products, to license or provide the source code to any of the Acquired Company Products for any purpose, including for the purposes of making derivative works, or (iii) imposes any present economic limitations on the Acquired Companies' commercial exploitation of any of the Acquired Company Products.
- (e) All Persons who have contributed, developed or conceived any Owned Intellectual Property or Acquired Company Products have done so pursuant to a valid, written and enforceable agreement that protects the confidential information of the Acquired Companies and

grants the Acquired Companies exclusive ownership of the Person's contribution, development or conception to the Owned Company Intellectual Property (including Intellectual Property in the Acquired Company Products).

- (f) The Acquired Companies are not under any obligation, whether written or otherwise, to develop any customized Intellectual Property (including any elements of any Acquired Company Products) for any Person (including any customer or end user) (except with respect to installation and configuration requirements in customer agreements entered into in the Ordinary Course of Business).
- (g) The Acquired Companies are in compliance with all obligations under any written agreement pursuant to which the Acquired Companies have obtained the right to use any third party Software, including Open Source Software, and in particular the Acquired Companies have purchased a sufficient number of seat licenses for the IT Systems as necessary for the immediate and currently anticipated future operation of the Business for the next six months.
- (h) Each of the Acquired Companies has taken commercially reasonable steps to protect all of the Acquired Company Intellectual Property, including its rights in the material trade secrets owned or used by the Business. The Acquired Companies have not disclosed any material confidential Acquired Company Intellectual Property (including the source code to any Acquired Company Products) to any Person other than pursuant to a written confidentiality agreement pursuant to which such Person agrees to protect such confidential information.
- The Acquired Companies own, validly license or otherwise possess valid rights to access and use all IT Systems, and, the IT Systems (i) are sufficient for, and operate and perform in all material respects as required in connection with, the immediate and currently anticipated future operation of the Business for the next six months, including as to capacity, scalability and ability to process current and anticipated peak volumes in a timely manner, and (ii) are in sufficiently good working condition to effectively perform all information technology operations. The Acquired Companies have implemented and maintain commercially reasonable security, disaster recovery and business continuity plans, procedures and facilities for the continuous operation of the Business in the event of a failure of its IT Systems that are, in the reasonable determination of the Acquired Companies' management team, adequate and in accordance with standard industry practice, and in the past three (3) years, there has not been any material failure with respect to the IT Systems that has not been remedied or replaced in all material respects. No Acquired Company has suffered a Security Incident in the past three (3) vears. The IT Systems and Acquired Company Products are free from Malicious Code, and the Acquired Companies have not received any complaints in writing from any customers related to the foregoing. Acquired Companies take and have taken commercially reasonable actions to protect and maintain the security, confidentiality, continuous operation and integrity of the IT Systems, including by implementing procedures designed to ensure that the IT Systems are free of Malicious Code, and the taking and storing on-site and off-site of back-up copies of critical data. There are, and for the past three (3) years have been, no defects, technical concerns or problems in any of the Acquired Company Products currently offered by the Acquired Companies that would prevent the same from performing substantially in accordance with their user specifications or functionality descriptions.

- (j) The Acquired Companies have complied in all material respects with, and are in compliance in all material respects with, all of the Data Security Requirements. The Acquired Companies have commercially reasonable safeguards in place to protect Business Data against loss, theft, or unauthorized disclosure. The Acquired Companies have not received any written notices or any written claims of, or been charged with, the violation by any Acquired Company of any Data Security Requirements. The transactions contemplated by this Agreement will not result in any liabilities in connection with any Data Security Requirements.
- (k) The Acquired Companies have all rights in and to the Acquired Company Product Data necessary for the operation of the Business, including where applicable the rights to publish, reproduce, distribute, license, sell and create derivative works of the Acquired Company Product Data.
- 4.13 <u>Contracts</u>. Except as listed or described on <u>Schedule 4.13</u>, as of the date hereof, none of the Acquired Companies is a party to or bound by any material agreements or contracts ("<u>Contracts</u>") that are of a type described below (such Contracts listed on <u>Schedule 4.13</u> together with the Leases are referred to herein as the "<u>Material Contracts</u>"):
- (a) any employment or individual service agreement that provides for annual base compensation exceeding \$125,000 per year and cannot be terminated by the Acquired Company party thereto without penalty or notice of thirty (30) days or less;
- (b) any collective bargaining agreement or other Contract with any labor union, works council, or other labor organization or collective bargaining representative;
- (c) any Contract for capital expenditures or the acquisition or construction of fixed assets in excess of \$150,000 individually or \$250,000 in the aggregate, in each case by an Acquired Company;
  - (d) any Contract with a Material Customer (other than non-disclosure agreements);
  - (e) any Contract with a Material Supplier (other than non-disclosure agreements);
- (f) any Contract relating to the borrowing of money, or the guaranty of another Person's borrowing of money or other obligation, including all notes, mortgages, indentures and other obligations, guarantees of performance, agreements and instruments for or relating to any lending or borrowing (other than advances to employees, officers or directors for expenses in the Ordinary Course of Business or transactions with customers on credit in the Ordinary Course of Business), in each case having an outstanding principal amount in excess of \$100,000;
- (g) any Contract granting any Person a Lien on all or any part of the assets of any of the Acquired Companies, other than Liens which will be released at or prior to the Closing and Permitted Liens;
  - (h) any Contract requiring the purchase of any particular product exclusively from a supplier;

- (i) any Contract under which any of the Acquired Companies is (i) a lessee or sublessee of any machinery, equipment, vehicle or other tangible personal property, or (ii) a lessor of any tangible personal property owned by any of the Acquired Companies, in any single lease under (i) or (ii) having an original value in excess of \$50,000;
- (j) any Contract under which any of the Acquired Companies has granted or received a license or under which it is obligated to pay or has the right to receive an annual royalty, license fee or similar payment in an amount in excess of \$50,000, in each case with respect to the use of any Intellectual Property (but excluding Contracts for non-exclusive licenses granted by or to any Acquired Companies in the Ordinary Course of Business, any Contract pursuant to which any Acquired Company grants any right or license to any Acquired Company Products that does not materially deviate from the Company's standard forms (copies of which have been made available to Parent), or "click-wrap" or "shrink-wrap" licenses and other licenses to "off-the-shelf" software or services granted to any Acquired Companies that are widely available through commercial distribution channels on standard terms and conditions with annual royalties, license fees or similar payments in an amount for less than \$150,000);
- (k) any Contract under which any of the Acquired Companies is a licensee of or is otherwise granted by a third party any rights to use any Intellectual Property in connection with any Acquired Company Products (but excluding Contracts for non-exclusive licenses granted to any Acquired Companies in the Ordinary Course of Business or Open Source Software, non-disclosure agreements, employee agreements, or "click-wrap" or "shrink-wrap" licenses and other licenses to "off-the-shelf" software or services granted to any Acquired Companies that are widely available through commercial distribution channels on standard terms and conditions with annual royalties, license fees or similar payments in an amount for less than \$150,000);
- (l) any Contract that relates to the ownership, development or use of any Intellectual Property in connection with the resolution of any claim or dispute related to Intellectual Property or affects any of the Acquired Companies' ability to use, enforce or disclose any Intellectual Property in connection with the resolution of any claim or dispute related to Intellectual Property;
- (m) any Contract relating to the provision of hosting, co-location or related services to any of the Acquired Companies, which services are used by any of the Acquired Companies to fulfill their obligations to provide Software and data hosting services to customers;
- (n) (A) any Contract containing an agreement by any of the Acquired Companies to provide any Person with access to the source code for any Acquired Company Products, or (B) any Contract between any Acquired Company, on the one hand, and an escrow agent, on the other hand, to provide for the source code for any Acquired Company Products to be put in escrow;
  - (o) any Contract involving the operation of any joint venture or partnership entity;
- (p) any Contract containing a covenant of any Acquired Company not to compete in any line of business or with any Person in any geographical area;

- (q) any Contract granting the counterparty exclusive rights;
- (r) any Contract that provides for any "most favored nation" pricing provisions granted by any Acquired Company to any Person from such Acquired Company or similar pricing provisions whereby pricing, discounts or benefits change based on the pricing, discounts or benefits offered to other Persons;
- (s) any Contract prohibiting, limiting or otherwise restricting in any way the Acquired Companies and/or their respective officers, directors, managers, equity owners or employees from soliciting customers or suppliers, or soliciting or hiring employees, of any other Person (but excluding Ordinary Course of Business non-solicitation of employees);
  - (t) any Contract with a Governmental Authority;
  - (u) any Contract that relates to an acquisition, divestiture, merger or similar transaction; or
- (v) any settlement, conciliation or similar Contract the performance of which involves any payment in excess of \$50,000 or is a settlement of any matter related to material Acquired Company Intellectual Property.

The Acquired Companies have made available to the Parent complete and correct copies of each Material Contract (and a true and written description of all oral contracts). Each Material Contract is a valid and binding obligation of the applicable Acquired Company party thereto, enforceable in accordance with its terms and conditions, subject to Laws of general application relating to public policy, bankruptcy, insolvency and the relief of debtors and rules of Law governing specific performance, injunctive relief and other equitable remedies, except for such failure to be valid and binding that would not be material to the Acquired Companies, taken as a whole. Neither the applicable Acquired Company party to such Material Contract nor, to the Company's Knowledge, any other party to such Material Contract is in material breach or default under such Material Contract, and no event has occurred which, with the passage of time or the giving of notice or both, would constitute a default or breach under any Material Contract, in each case, except for such breaches and defaults that would not be material to the Acquired Companies, taken as a whole.

4.14 <u>Litigation</u>. Except as would not be material to the Acquired Companies, taken as a whole, (a) there are no, and for of the past three (3) years there have been no, Legal Proceedings pending or, to the Company's Knowledge, threatened against or affecting any such Acquired Companies and (b) none of the Acquired Companies is, or for the past three (3) years has been, subject to any judgment, order, ruling, award, writ, or decree of any Governmental Authority. There is no action, suit, proceeding or investigation by the Company pending or which the Company intends to initiate.

#### 4.15 Employee Benefits.

(a) <u>Schedule 4.15(a)</u> lists each material Employee Benefit Plan and PEO Plan by jurisdiction. The Company has provided to Parent correct and complete copies of the following, as applicable, with respect to each Employee Benefit Plan: (i) current plan documents and all

amendments thereto and (ii) the most recent summary plan description, and all summaries of material modification thereto, (iii) the most recent determination or opinion letter received from the Internal Revenue Service, and (iv) the most recent financial statements and annual reports on Form 5500 (including attached schedules) required to be filed with the IRS with respect to each such Employee Benefit Plan (if any such report was required). As applicable with respect to each PEO Plan, the Company has provided to Parent correct and complete copies of: (i) the most recent determination or opinion letter issued by the IRS with respect to each PEO Plan that is intended to be qualified under Section 401(a) of the Code; (ii) the two most recent annual reports (IRS Forms 5500 series); (iii) the plan and trust documents, related insurance contracts, and the most recent summary plan description distributed to participants (and any summaries of material modifications thereto); (iv) pending voluntary correction filings and a description of any pending self-correction actions; and (v) any non-routine correspondence with any Governmental Authority.

- (b) Each Employee Benefit Plan and, to the Company's Knowledge, each PEO Plan, has been established, funded, administered and maintained, in form and operation, in accordance with its terms and the requirements of applicable Law, including ERISA and the Code, if applicable. No Acquired Company has incurred any penalty or Tax (whether or not assessed) under Sections 4980H, 4980D, 4980B, 6721, or 6722 of the Code and no circumstances or events have occurred that would reasonably be expected to result in the imposition of any such penalties or Taxes. With respect to each Employee Benefit Plan and PEO Plan, all contributions, premiums or other payments have been paid on a timely basis.
- (c) Each Employee Benefit Plan and PEO Plan that is intended to meet the requirements of a "qualified plan" under Section 401(a) of the Code has received a favorable determination letter from the Internal Revenue Service or is in the form of a prototype document that is the subject of a favorable opinion letter from the Internal Revenue Service and nothing has occurred that would reasonably be expected to adversely affect the qualification thereof.
- (d) No Employee Benefit Plan or PEO Plan is and no Acquired Company sponsors, maintains, contributes to, has any obligation to contribute to or has any liability with respect to (i) any "defined benefit plan" (as defined in Section 3(35) of ERISA) or any other plan that is or was subject to Title IV of ERISA or Section 412 of the Code, (ii) a "multiemployer plan" (as defined in Section 3(37) of ERISA), (iii) any "multiple employer plan" (within the meaning of Section 210 of ERISA or Section 413(c) of the Code), or (iv) any "multiple employer welfare arrangement" (as defined in Section 3(4)) of ERISA). No Acquired Company has any liability by reason of at any time being treated as a single employer under Section 414 of the Code with any other Person. No Employee Benefit Plan provides and no Acquired Company has any liability for the provision of postemployment or post-termination health, life insurance or other welfare-type benefits other than as required by Section 4980B of the Code or similar state Law for which the covered individual pays the full premium cost of such coverage.
- (e) No Legal Proceedings or other actions, claims (other than routine benefit claims) or lawsuits are pending or, to the Company's Knowledge, threatened against or with respect to any Employee Benefit Plan or PEO Plan or related trust, sponsor, administrator or fiduciary with respect to any Employee Benefit Plan or PEO Plan, nor to the Company's Knowledge are there facts that would reasonably be expected to form the basis for any such action, claim or lawsuit. There have been no "prohibited transactions" within the meaning of Section 4975

of the Code or Sections 406 or 407 of ERISA and not otherwise exempt under Section 408 of ERISA and no breaches of fiduciary duty (as determined under ERISA) with respect to any Employee Benefit Plan or, to the Company's Knowledge, any PEO Plan.

- (f) Each Employee Benefit Plan or PEO maintained by an Acquired Company for employees or other service providers who reside or work primarily outside of the United States (each a "Non-U.S. Employee Benefit Plan") required by any applicable Law to be registered or approved by a Governmental Authority has been so registered or approved and has been maintained in good standing with the applicable Governmental Authority in each case. All contributions to and material payments from each Non-U.S. Employee Benefit Plan under the terms of such plan or applicable Law have been made on or before their respective due dates, and all contributions for any period ending on or before the Closing Date that are not yet due have been accrued in accordance with country-specific accounting practices and principles. Each Non-U.S. Employee Benefit Plan required under any applicable Law to be funded or insured has been funded or insured in accordance with such Law and the Transactions will not cause such funding or insurance obligations to be less than such benefit obligations. No Non-U.S. Employee Benefit Plan is a "defined benefit plan" (as defined in ERISA, whether or not subject to ERISA), seniority premium, termination indemnity, provident fund, gratuity or similar plan or arrangement, and no unfunded or underfunded liabilities exist with respect to any Non-U.S. Employee Benefit Plan.
- (g) Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby, whether alone or in connection with any other event, could, directly or indirectly, (A) give rise to any liability under any Employee Benefit Plan, including liability for severance pay, (B) accelerate the timing of vesting, funding or payment, or increase the amount or value, of any compensation or benefits due to any Company Employee (whether current, former or retired), (C) entitle any Company Employee (whether current, former or retired) to any payment or benefit, including any change in control, transaction bonus, retention payment or other similar payment(whether in cash, property or vesting of property) or benefit, or (D) give rise to payments or benefits (whether in cash, property or vesting of property) to any "disqualified individual" (within the meaning of Section 280G of the Code) that could, individually or in combination with any other such payment, constitute an "excess parachute payment" (within the meaning of Section 280G(b)(1) of the Code).
- (h) Each Employee Benefit Plan that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has in all material respects been operated and maintained in compliance with Section 409A of the Code and the regulations and guidance promulgated thereunder.
- (i) None of the Acquired Companies has any indemnity or gross-up obligation for any Taxes, including under Section 409A or 4999 of the Code.

# 4.16 <u>Environmental Matters</u>.

(a) Each of the Acquired Companies is, and for the three (3) years prior to the date hereof has been, in compliance with, and has no material liability (contingent or otherwise) pursuant to, all applicable Environmental Laws, which compliance includes obtaining, maintaining, and complying with all Permits required under Environmental Laws.

- (b) During the three (3) years prior to the date hereof (or earlier to the extent unresolved), none of the Acquired Companies has received or been party to any written notice, report, order or decree regarding any actual or alleged violation of Environmental Laws, or any unresolved liability (contingent or otherwise) arising under Environmental Laws, in each case relating to the Business of the Acquired Companies.
- (c) None of the Acquired Companies, nor any of their predecessors or affiliates, has treated, stored, sold, distributed, manufactured, marketed, disposed of, arranged for or permitted the disposal of, transported, handled, exposed any Person to, or released any hazardous or toxic materials, substances or wastes (including petroleum), or owned or operated any property or facility contaminated by any hazardous or toxic materials, substances or wastes (including petroleum), so as to give rise to any material liabilities (contingent or otherwise) pursuant to Environmental Laws.
- (d) The Acquired Companies have furnished to Parent all environmental, health or safety audits, reports and other material environmental, health or safety documents relating to each Acquired Company's, past or current properties, facilities or operations which are in their possession or under their reasonable control.

#### 4.17 <u>Labor Matters</u>.

- (a) No Acquired Company is party to or bound by any Labor Agreement, and there is not, and during the past three (3) years none of the Acquired Companies has experienced or been affected by, any actual or, to the Company's Knowledge, threatened strike, picketing, handbilling, boycott, work stoppage, lockouts, slowdown, material labor grievance, material labor arbitration or other material labor dispute (including, without limitation, in connection with the transaction contemplated hereunder), or charge or complaint of unfair labor practice or employment discrimination. To the Company's Knowledge, in the past five (5) years, there has been no union or other labor organizational activity with respect to any employees of the Acquired Companies, and no labor union, works council, other labor organization, or group of employees of any Acquired Company has made a demand for recognition or certification, and there are no representation or certification proceedings presently pending or threatened to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority. No employee of any Acquired Company is represented by a labor union, works council, or other labor organization. The Acquired Companies represent that they have no notice or consultation obligations to any labor union, labor organization or works council, which is representing any Company Employee in connection with the execution of this Agreement or consummation of the transactions contemplated by this Agreement.
- (b) There have been no "mass layoffs" or "plant closings" (as defined by the Worker Adjustment and Retraining Notification Act of 1988, as amended, or any similar state or local Law (collectively, the "WARN Act")) affecting employees at the Acquired Companies at any time during the past three (3) years, the liability for which remains unsatisfied. No facility closure or shutdown, reduction-in-force, furlough, layoff, material work schedule change or reduction in hours, or material reduction in salary or wages, or other workforce changes affecting employees or individual independent contractors of the Company has occurred since March 1, 2020 or is currently contemplated, planned or announced but not yet implemented, including as a

result of COVID-19. No Acquired Company has not otherwise experienced any material employment-related liability with respect to COVID-19.

- Each of the Acquired Companies is, and for the past three (3) years has been, in compliance in all material respects with all applicable Laws governing employment and labor including, without limitation, all Laws respecting terms and conditions of employment, health and safety, wages and hours (including the classification of independent contractors and exempt and non-exempt employees), holidays, immigration (including the completion of Forms I-9 for all employees and the proper confirmation of employee visas), employment harassment, discrimination or retaliation, whistleblowing, disability rights or benefits, equal opportunity, plant closures and layoffs (including the WARN Act), employee trainings and notices, workers' compensation, labor relations, employee leave issues, COVID-19, affirmative action and unemployment insurance. Except as would not reasonably be expected to result in material liability for any Acquired Company: (i) each Acquired Company has fully and timely paid, or set aside funds to fully and timely pay, all wages, salaries, wage premiums, commissions, bonuses, severance and termination payments, fees, and other compensation indemnities for work trips, and medical and social security charges and withholding Taxes that have come due and payable to its current or former employees and independent contractors under applicable Laws, Contract or Acquired Company policy; and (ii) each individual who is providing or has provided services to any Acquired Company and is or was classified and treated as an (x) independent contractor, consultant, leased employee, or other non-employee service provider, or (y) exempt employee, in each case, is and has been properly classified and treated as such for all applicable purposes. To the extent required by the Czech Labour Code, the Czech Company has entered into a written employment agreement with all persons carrying out "dependent work" (as defined in the Czech Labour Code) for it and has made available all such contracts to Parent. There are no obligations relating to payments in kind, compensations or other benefits (including purchase options or incentives) with any of the employees. There are no pension arrangements relating to the employees other than those required to be maintained by the Acquired Companies under applicable Laws.
- (d) The Acquired Companies have promptly, thoroughly and impartially investigated all sexual harassment, or other discrimination, retaliation or policy violation allegations of which they are aware. With respect to each such allegation with potential merit, the applicable Acquired Company has taken prompt corrective action that is reasonably calculated to prevent further improper conduct. To the Company's Knowledge, there are not any sexual harassment allegations relating to officers, directors, or senior management of the Company, that could result in material liability or, if known to the public, would bring any Acquired Company into material disrepute. No employee of any Acquired Company has, to the Knowledge of the Company, within the past ten (10) years, been involved in any criminal proceedings relating to the business of any Acquired Company.
- (e) To the Company's Knowledge, no officer, director, or senior management employee of the Company (A) has any plans to terminate employment with any Acquired Company or (B) is a party to any confidentiality, noncompetition, nonsolicitation, proprietary rights or other similar agreement between such employee and any other Person besides the Acquired Companies that would be material to the performance of such employee's employment duties, or the ability of the Acquired Companies to conduct their business.

- (f) To the Company's Knowledge, no current or former employee or independent contractor of the Acquired Companies is in any material respect in violation of any term of any employment agreement, nondisclosure agreement, common law nondisclosure obligation, fiduciary duty, noncompetition agreement, restrictive covenant or other obligation: (i) owed to the Acquired Companies; or (ii) owed to any third party with respect to such person's right to be employed or engaged by the applicable Acquired Company.
- (g) None of the Acquired Companies are or have been party to, during the past three (3) years, any material Legal Proceedings or material disputes involving, or had any liability with respect to, any single employer, joint employer or co-employer claims or causes of action by any individual who was employed or engaged by a third party and providing services to any Acquired Company, and to the Company's Knowledge, each third party providing individuals for the Acquired Companies on a temporary, seasonal or leased basis are in material compliance with all applicable labor and employment Laws.
- (h) To the Company's Knowledge, no current employee of any Acquired Company with annualized compensation at or above \$125,000, intends to terminate his or her employment prior to the one (1) year anniversary of the Closing.
- (i) No Legal Proceedings are pending or threatened between any employees, former employees, or other individuals engaged by any Acquired Companies or social security and labor authorities (including, but not limited to, in relation to health and hygiene at the workplace, occupational health and safety and prevention of work-related accidents), and there are no facts, circumstances or information that would, or could reasonably be expected to, result in any such Legal Proceedings against any of the Acquired Companies.
- 4.18 <u>Insurance Policies</u>. <u>Schedule 4.18</u> sets forth a true and complete list of all insurance policies (including "self-insurance" programs) maintained as of the date hereof by the Acquired Companies, (collectively, the "<u>Insurance Policies</u>"), true and complete copies of which have been made available to Parent. Each such insurance policy is legal, valid, binding and enforceable and in full force and effect, and cover such losses and risks, and in such amounts, as are prudent and customary in the business in which the Acquired Companies are engaged, and none of the Company nor any of the Company's Subsidiaries is in breach or default with respect to its obligations under such insurance policies (including with respect to payment of premiums). The Company has not received any written notice of pending cancellation of, premium increase with respect to, or material alteration of coverage under, any of such Insurance Policies. All such Insurance Policies are fully paid and valid and binding in accordance with their terms.
- 4.19 <u>Affiliated Transactions</u>. Except for (i) employment and equity agreements and arrangements entered into by the Acquired Companies, (ii) advances to employees in the Ordinary Course of Business, (iii) participation by employees, officers and directors in any Employee Benefit Plan or PEO Plan (iv) intercompany agreements and (v) director and officer indemnification agreements approved by the board of directors of the Company, no officer, director or Affiliate of the Company or, to the Company's Knowledge, any individual related by blood, marriage or adoption to any individual or any entity in which any such Person or individual owns any beneficial interest is a party to any material agreement, contract, commitment or transaction with any of the Acquired Companies or, other than such Person's direct or indirect

equity interest in the Company, has any interest in any material property owned or used by any of the Acquired Companies thereof.

### 4.20 <u>Material Business Relationships</u>.

- (a) <u>Schedule 4.20(a)</u> sets forth the top twenty-five (25) customers of the Acquired Companies (measured by aggregate annual contract value of products or services provided by the Acquired Companies to a customer and such customer's Affiliates) for the fiscal year ended December 31, 2020 (each, a "<u>Material Customer</u>" and, collectively, the "<u>Material Customers</u>"). No Material Customer (i) has notified an Acquired Company in writing that it will cease or has threatened in writing to cease to use, or reduce its use of, the services of the Acquired Companies, or (ii) has notified any Acquired Company in writing that it will materially and adversely modify its relationship with the Acquired Companies (including by materially changing the pricing terms or other material terms of such Material Customer's business with the Acquired Companies) or has threatened in writing such material and adverse modification.
- (b) Schedule 4.20(b) sets forth the top ten (10) suppliers of materials, equipment or services to the Acquired Companies (measured by dollar volume of purchases by the Acquired Companies, but excluding professional and utility services provided to the Acquired Companies in the Ordinary Course of Business) for the fiscal year ended December 31, 2020 (each, a "Material Supplier" and, collectively, the "Material Suppliers"). No Material Supplier (i) has notified an Acquired Company in writing that it will cease or reduce or has threatened in writing to cease or reduce provision of materials, equipment or services to the Acquired Companies, or (ii) has notified an Acquired Company in writing that it will materially and adversely modify its relationship with the Acquired Companies (including by materially changing the pricing terms or other material terms of such Material Supplier's business with the Acquired Companies) or has threatened in writing such material and adverse modification.

#### 4.21 <u>Trade Laws</u>.

- (a) For the five (5) years prior to the date hereof, each of the Acquired Companies, any director, officer, and, to the Company's Knowledge, any employee, agent, or representative thereof has been in compliance, and is currently in compliance with all Trade Laws. Each of the Acquired Companies has obtained, and is currently and has been in compliance with the terms of, any export license required for the lawful export of any item, product, technology, or software of such Acquired Company.
- (b) Except as set forth in Schedule 4.21(b), for the five (5) years prior to the date hereof, no Acquired Company, nor any director, officer, nor to the Company's Knowledge, any employee, agent, or representative thereof, has engaged in, and is not now engaging in, directly or, to the Company's Knowledge, indirectly, any dealings or transactions in a Sanctioned Country or with a Sanctioned Person. No Acquired Company nor, any director, officer, nor, to the Company's Knowledge, any employee, agent, or representative thereof, or any other Person authorized to act for or on behalf of any Acquired Company, is, or is owned, fifty percent (50%) or more, directly or to the Company's Knowledge, indirectly, by a Sanctioned Person.

- (c) No Acquired Company or has been in the last five (5) years is involved in any proceeding relating to, or has received a written request for information from, or has made any voluntary or involuntary disclosure to any Governmental Authority regarding, any Trade Law.
- 4.22 <u>Anti-Money Laundering</u>. For the five (5) years prior to the date hereof, the operations of each Acquired Company have been conducted at all times in material compliance with the requirements of applicable anti-money laundering laws, including, but not limited to, the Bank Secrecy Act of 1970, as amended by the USA PATRIOT ACT of 2001, and the rules and regulations promulgated thereunder, and the anti-money laundering laws of the various jurisdictions in which each Acquired Company conducts business (collectively, the "<u>Money Laundering Laws</u>") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving any Acquired Company or any of its subsidiaries with respect to the Money Laundering Laws is pending or, to the Company's Knowledge, threatened in writing.

# 4.23 <u>Anti-Bribery and Anti-Corruption</u>.

- (a) No Acquired Company, or any officer, director or employee, or, to the Company's Knowledge, any agent, representative, or consultant thereof, or any other Person associated with or acting for or on behalf of any Acquired Company, has, directly or indirectly, in connection with the business of any Acquired Company:
  - (i) made, offered or promised to make or offer any payment, loan or transfer of anything of value, including any reward, advantage or benefit of any kind, to or for the benefit of any Government Official, candidate for public office, political party or political campaign, for the purpose of (A) influencing any act or decision of such Government Official, candidate, party or campaign, (B) inducing such Government Official, candidate, party or campaign to do or omit to do any act in violation of a lawful duty, (C) obtaining or retaining business for or with any person, (D) expediting or securing the performance of official acts of a routine nature, or (E) otherwise securing any improper advantage;
  - (ii) paid, offered or promised to make or offer any bribe, payoff, influence payment, kickback, unlawful rebate, or other similar unlawful payment of any nature;
  - (iii) made, offered or promised to make or offer any unlawful contributions, gifts, entertainment or other unlawful expenditures;
  - (iv) established or maintained any unlawful fund of corporate monies or other properties;
  - (v) created or caused the creation of any false or inaccurate books and records of the Acquired Companies related to any of the foregoing; or
  - (vi) otherwise violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, 15 U.S.C. §§78dd-1, *et seq*. ("<u>FCPA</u>"), the UK Bribery Act of 2010 ("<u>UK Act</u>"), or any other applicable anti-corruption or anti-bribery law.

(b) The Acquired Companies have established and currently maintain policies, procedures and internal controls designed to reasonably promote compliance with the FCPA, the UK Act, and all other applicable anti-bribery laws. The Acquired Companies have not committed any of the crimes provided under Legislative Decree no. 231/01 ("Decree 231") and none of them has received any notice from any Governmental Authority of non-compliance with the Decree 231 nor is subject to any litigation, pending or threatened in writing with respect to any alleged non-compliance or violation thereof. No director, officer, employee, representative, or any other Person, duly authorized to act for or on behalf of the Acquired Companies have kept a behavior or committed any action, also by way of omission, that may result in the Acquired Companies being held liable under Decree 231.

# ARTICLE V REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub represent and warrant to the Company and the Representative as of the date hereof and as of the Closing Date as follows:

- 5.1 <u>Organization of Parent and Merger Sub</u>. Parent is a Minnesota limited liability company and Merger Sub is a Delaware corporation, each validly existing and in good standing under the Laws of the jurisdiction of its organization.
- 5.2 <u>Ownership of Merger Sub; No Prior Activities</u>. Merger Sub is a direct, wholly-owned subsidiary of Parent, was formed solely for the purpose of engaging in the transactions contemplated by this Agreement and has engaged in no business activity other than as contemplated by this Agreement. Except for obligations or liabilities incurred in connection with the transactions contemplated by this Agreement and described herein, Merger Sub has not and will not have incurred, directly or indirectly, through any subsidiary or Affiliate, any obligations or liabilities or engaged in any business activities of any type or kind whatsoever or entered into any agreements or arrangements with any Person.
- 5.3 Authorization; Enforceability. Each of Parent and Merger Sub has requisite corporate power and authority to execute and deliver this Agreement and each other agreement contemplated hereby to which Parent or Merger Sub is a party and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement by Parent and Merger Sub and each other agreement contemplated hereby to which Parent or Merger Sub is a party and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action, and no other corporate proceedings on the part of Parent and Merger Sub and no stockholder votes are necessary to authorize this Agreement or the other agreements contemplated hereby to which Parent or Merger Sub is a party or to consummate the Merger and the other transactions contemplated hereby and thereby, other than the filing and recordation of the Certificate of Merger. This Agreement has been duly executed and delivered by Parent and Merger Sub and, assuming the due authorization, execution and delivery by the other Parties hereto, constitutes the valid and legally binding obligation of Parent and Merger Sub, enforceable in accordance with its terms and conditions, subject to Laws of general application relating to public policy, bankruptcy, insolvency and the relief of debtors and rules of Law governing specific performance, injunctive relief and other equitable remedies.

#### 5.4 Non-contravention.

- (a) Neither the execution and delivery of this Agreement or any other agreement contemplated hereby to which Parent or Merger Sub is a party, nor the consummation of the Merger or the other transactions contemplated hereby or thereby, will (i) violate any Laws to which Parent or Merger Sub is subject, (ii) violate any provision of Parent's or Merger Sub's Organizational Documents, or (iii) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any Person the right to accelerate, terminate, modify or cancel or require any notice or consent under, or result in the imposition of any Lien (other than Permitted Liens) upon any of the assets of Parent or Merger Sub under, any agreement, contract, lease, license, instrument or other arrangement to which Parent or Merger Sub is a party or by which any of them is bound or to which any of their assets is subject, except in each case of clause (i) or (iii) where the violation, conflict, breach, default, acceleration, termination, modification, cancellation, failure to give notice or obtain consent or Lien would not adversely affect or delay Parent's or Merger Sub's performance under this Agreement or any other agreement contemplated hereby to which Parent or Merger Sub is a party or the consummation of the transactions contemplated by this Agreement or such other agreements.
- (b) Except for the filing of the Certificate of Merger under the DGCL and the pre-merger notification requirements of the HSR Act, neither Parent nor Merger Sub needs to give any notice to, make any filing with or obtain any authorization, consent or approval of any Governmental Authority in order for the Parties to consummate the transactions contemplated by this Agreement.
- 5.5 <u>Brokers' Fees</u>. Neither Parent nor Merger Sub has any liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by this Agreement.
- 5.6 <u>Litigation</u>. There are no Legal Proceedings pending against, or threatened against, Parent or Merger Sub that would adversely affect or delay Parent's or Merger Sub's performance under this Agreement or the consummation of the transactions contemplated by this Agreement.
- 5.7 <u>Financing</u>. As of the date hereof and as of the Closing, Parent shall have sufficient funds necessary to consummate the transactions contemplated by this Agreement on the terms and subject to the conditions set forth herein.
- 5.8 <u>Solvency</u>. Assuming the accuracy of the representations and warranties in <u>Article IV</u> (and assuming (a) satisfaction of the conditions set forth in <u>Article VIII</u>, (b) that any estimates, projections or forecasts with respect to the Acquired Companies provided by the Company or the Company Stockholders or any of their agents or representatives have been prepared in good faith and are based upon reasonable assumptions, and (c) that all financial information concerning the Acquired Companies provided to Parent by the Company or the Company Stockholders or their agents or representatives (whether before or after the date hereof) present fairly the net assets of the Acquired Companies as of the respective dates thereof and the results of operations and cash flows of the Acquired Companies for the respective periods covered thereby) and that the Company complied in all material respects with the covenants contained in

Section 6.1, immediately following the Effective Time and after giving effect to the Merger, Parent and each of its Subsidiaries (including the Surviving Corporation and the Company Subsidiaries) (i) will be able to pay their respective debts as they become due and will own property that has a fair saleable value greater than the amounts required to pay their respective debts (including a reasonable estimate of the amount of all contingent liabilities) and (ii) will have adequate capital to carry on their respective businesses. No transfer of property is being made and no obligation is being incurred in connection with the transactions contemplated by this Agreement with the intent to hinder, delay or defraud any present or future creditors of Parent, the Surviving Corporation or the Company Subsidiaries.

# ARTICLE VI PRE-CLOSING COVENANTS RELATING TO THE COMPANY

- 6.1 <u>Conduct of Business.</u> Except as contemplated by this Agreement, as required by Law, as set forth on <u>Schedule 6.1</u> or as otherwise consented to in writing by Parent from the date hereof through the Closing or the earlier termination of this Agreement pursuant to <u>Article XI</u> hereof, the Company covenants and agrees that:
- (a) The Company shall cause each of the Acquired Companies to conduct the Business in all material respects according to the Ordinary Course of Business.
- The Company shall not permit any Acquired Company to (i) increase or promise to increase the compensation or benefits payable or provided, or to become payable or provided to, whether conditionally or otherwise, any current or former Company Employee, except (A) in connection with any hiring or engagement of any Company Employee by the Company to the extent the annual total compensation opportunity for such employee is less than \$125,000 or (B) for increases in total compensation for any employee to the extent that such increases in total compensation are no greater than 5% for any such employee, (ii) grant or announce any cash, equity or equity-based incentive awards, bonuses, transaction, retention, severance or other compensation or benefits to any current or former Company Employee, other than grants of ordinary course cash amounts to newly hired or engaged Company Employees consistent with past practice, (iii) adopt, establish, terminate, modify or amend any Employee Benefit Plan or PEO Plan (or any other compensation or benefit plan or arrangement that would be an Employee Benefit Plan or PEO Plan if it was in effect on the date hereof) other than as permitted under clauses (i) and (ii) of this Section 6.1(b), (iv) accelerate the time of payment, vesting or funding of any compensation or benefits or increase in the amount of compensation (whether in cash, property or the vesting of property) provided under any Employee Benefit Plan or otherwise, or (v) hire, engage, terminate (other than for "cause"), furlough or temporarily lay off the employment or service of any current or former Company Employee with an annual base compensation in excess of \$125,000.
- (c) Other than in connection with the exercise of any Options or Company Warrants outstanding as of the date hereof and disclosed in <u>Schedule 4.3(b)</u> and in accordance with the existing terms of the Equity Incentive Plans, the Company shall not permit any Acquired Company to issue, sell, pledge, dispose of or otherwise subject to any Lien any shares of capital stock of any class or other equity interests of such Acquired Company, or any securities convertible

into any such shares or other equity interests, or any rights, warrants, calls, subscriptions or options to acquire any such shares, equity interests or convertible securities.

- (d) The Company shall not permit any Acquired Company to sell, lease, license, encumber or otherwise dispose of, or agree to sell, lease, license, encumber or otherwise dispose of, any of its material assets other than in the Ordinary Course of Business (including non-exclusive licenses granted to customers in the Ordinary Course of Business).
- (e) The Company shall not permit any Acquired Company to incur any indebtedness for borrowed money except in the Ordinary Course of Business, guarantee any such indebtedness, issue or sell any debt securities or warrants or rights to acquire any debt securities of such Person or guarantee any debt securities of others.
- (f) The Company shall not permit any Acquired Company to enter into any Contract that, if entered into prior to the date hereof, would be considered a Material Contract.
- (g) The Company shall not permit any Acquired Company to enter into any contract with a supplier or vendor with a term of greater than one year. In the event any Acquired Company does enter into a contract with a supplier or vendor with a term of greater than one year without Parent's prior written consent, any current or long term asset related to such contract shall be excluded from the definition of Current Assets.
- (h) The Company shall not permit any Acquired Company to take or fail to take any action that, if such action were taken or failed to be taken between November 1, 2020 and the date hereof, would be required to be disclosed on Schedule 4.7 (except as would be required to be disclosed pursuant Section 4.7(a), Section 4.7(b) or Section 4.7(w).
- (i) The Company shall not permit any Acquired Company to publicly announce an intention, enter into any formal or informal agreement, or otherwise make a commitment to do any of the foregoing.
- (j) The Company shall promptly deliver to Parent written notice upon becoming aware of (A) any material failure on the part of the Company to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder or (B) (1) any injunction, writ or order of any nature issued and directing that the transactions provided for herein not be consummated as herein provided or (2) any action pending or threatened before any Governmental Authority with respect to the transactions contemplated hereby; provided that the delivery of any notice pursuant to this clause (j) shall not limit or otherwise affect the remedies available hereunder to the party receiving such notice, or the representations or warranties of, or the conditions to the obligations of, the Parties.
- (k) The Company shall not, and shall not permit any Acquired Company to, file or change any Tax election, change any method of accounting for Tax purposes, file any amended Tax Return, enter into any closing agreement related to Taxes, settle any Tax claim or assessment, request or consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment (other than pursuant to extensions to file Tax Returns obtained in the Ordinary Course of Business), fail to pay any Tax that becomes due and payable (including any estimated Tax payments), or file any Tax Return in a manner inconsistent with past practice.

### 6.2 <u>Consents; Filings; Etc.</u>

- (a) The Company shall use its reasonable best efforts to obtain prior to the Closing such consents of third parties and to give notices to third parties to avoid the breach or termination of any Contract that is listed on Schedule 6.2(a); provided, that in no event shall the Company be required to make any payments to third parties to obtain such waivers or consents, and the failure to obtain such waivers and consents shall not, in and of itself, be a breach of this Section 6.2(a) or give rise to the failure of any condition to Closing set forth in Article VIII. Furthermore, the Company shall use its reasonable best efforts to take or cause to be taken all actions and to do or cause to be done all things (including the satisfaction of the conditions set forth in Section 8.1 and Section 8.1(c)) that are necessary, proper or advisable in compliance with applicable Laws to consummate and make effective, as soon as reasonably practicable, the transactions contemplated hereby. If any suit or other action is threatened or instituted by any Governmental Authority or other Person challenging the validity or legality, or seeking to restrain the consummation of the transactions contemplated by this Agreement, the Company shall use its reasonable best efforts to avoid, resist, resolve or, if necessary, defend such suit or action.
- In addition to and without limiting the foregoing, the Company undertakes and agrees to (b) file as soon as practicable, and in any event not later than the fifth (5th) Business Day after the date hereof, a Notification and Report Form under the HSR Act with the United States Federal Trade Commission and the Antitrust Division of the United States Department of Justice. The Company shall (i) make a reasonable response to any inquiries or requests received from any Governmental Authority for additional information or documentation, and (ii) use its reasonable best efforts to cause the waiting periods or other requirements under the HSR Act, to terminate or expire at the earliest possible date. The Company shall (A) promptly notify Parent of any written communication to the Company or its Affiliates from any Governmental Authority and, subject to applicable Law, permit Parent to review in advance any proposed written communication to any of the foregoing (and consider in good faith the reasonable views of Parent in connection therewith where such views are provided in a timely manner); (B) to the extent possible, consult with Parent in advance and, to the extent permitted by such Governmental Authority, give Parent the opportunity to attend and participate in any substantive meeting or prearranged discussion with any Governmental Authority in respect of any filings, investigation or inquiry concerning this Agreement; and (C) furnish Parent with copies of all correspondence, filings and communications (and memoranda setting forth the substance thereof) between it and its Affiliates and their respective representatives on the one hand, and any Governmental Authority or members of its staff on the other hand, with respect to this Agreement; provided, however, that the Company shall be permitted to designate confidential information as 'Outside Antitrust Counsel Only' and remove or redact any information or contents that are protected by legal privilege and restrict Parent's access to such materials accordingly. The Company shall not be required to share a copy of its filing made under the HSR Act. The Company shall specifically not request early termination of the waiting period under the HSR Act.
- 6.3 <u>Publicity</u>. On and after the date hereof and through the Closing Date, the Company shall, and cause each of the Acquired Companies and their respective Affiliates to, consult with Parent before issuing any press release or making any public announcement relating to the subject matter of this Agreement, and the Company shall not, and shall not permit any of the Acquired Companies or their Affiliates to, issue any press release or make any public

announcement relating to the subject matter of this Agreement without the prior written consent of Parent, provided, however, that no such approval shall be necessary to the extent disclosure may be required by applicable Law, in which case, Parent will have the right to review and comment on such press release or public announcement prior to publication. For the avoidance of doubt, discussions by senior management of the Company of subject matter that was previously publicly announced or otherwise mutually agreed in writing between Parent and the Company in press interviews shall not be deemed to violate this <u>Section 6.3</u>.

- Access. Prior to the first to occur of the Closing Date and the termination of this Agreement pursuant to Article XI hereof, the Company will permit Parent and its representatives (including legal counsel and accountants) to have, upon prior written notice, reasonable access, during normal business hours and under reasonable circumstances, and in a manner so as not to interfere with the normal business operations of the Acquired Companies, to the premises, books, records (including Tax records), contracts and documents of or pertaining to the Acquired Companies. All requests for such access shall be directed to such Persons as the Company may designate in writing from time to time (collectively, the "Designated Contacts"). Other than the Designated Contacts, neither Parent nor any of its Affiliates or any of their respective representatives shall contact any employee, contractor, customer, supplier, landlord, lender or other material business relation of any of the Acquired Companies without the prior written consent of the Company, which consent shall not be unreasonably withheld, conditioned or delayed. Parent shall comply with, and shall cause its representatives to comply with, all of their obligations under the Confidentiality Agreement with respect to the information disclosed pursuant to this Section 6.4, which Confidentiality Agreement will remain in full force and effect.
- 6.5 <u>Requisite Stockholder Approval</u>. The Company shall have obtained the Requisite Stockholder Approval by the date hereof and is delivering the Requisite Stockholder Approval to Parent concurrently with this Agreement.
- 6.6 <u>Information Statement</u>. As promptly as practicable after the date hereof (but in any case, within seven (7) days following the date of this Agreement), the Company, in consultation with Parent, shall prepare an information statement (the "<u>Information Statement</u>") in accordance with Section 228 of the DGCL and shall mail or cause the Information Statement to be mailed to any holder of Company Shares who has not executed in accordance with the DGCL a written consent in favor of adoption of this Agreement and the approval of the Merger and the transactions contemplated hereby and who was otherwise entitled to vote thereon.
- 6.7 Section 280G. To the extent that any "disqualified individual" (within the meaning of Section 280G(c) of the Code and the regulations thereunder) has the right to receive any payments or benefits that constitute "parachute payments" (within the meaning of Section 280G(b)(2)(A) of the Code and the regulations thereunder), then the Company will (a) use reasonable best efforts to obtain from each such "disqualified individual" a waiver of such disqualified individual's rights to some or all of such payments or benefits (the "Waived 280G Benefits") so that any remaining payments and/or benefits shall not be deemed to be "excess parachute payments" (within the meaning of Section 280G of the Code and the regulations thereunder) and, (b) with respect to each individual who agrees to the waiver described in clause (a), submit to a vote of the Company's Stockholders entitled to vote on such matters, the right of any such "disqualified individual" to receive the Waived 280G Benefits. Not less than three (3)

Business Days prior to distribution of any materials to the holders of Common Stock in connection with the waiver and vote described in this Section 6.8, the Company shall provide Parent for its review and comment a copy of all such materials (i.e., the waiver, disclosure statement, equityholder consent, and any other materials to be submitted to the Company's Stockholders in connection with such vote) and a copy of its Section 280G calculations, and the Company shall incorporate all of Parent's reasonable comments to such materials and Section 280G calculations. No later than one (1) Business Day prior to the Closing, the Company shall deliver to Parent evidence reasonably satisfactory to Parent that (A) the vote described in this Section 6.7 was solicited in conformance with Section 280G of the Code and the requisite Company Stockholder approval was obtained with respect to any such Waived 280G Benefits, or (B) no such approval was obtained. Any of the Waived 280G Benefits that fail to be approved as contemplated above shall not be paid or provided to such "disqualified individuals."

### 6.8 <u>Financing Cooperation</u>.

- From the date hereof and ending at the earlier of (i) the Closing Date and (ii) the (a) termination of this Agreement pursuant to Section 11.1, the Company shall use commercially reasonable efforts to, and shall use commercially reasonable efforts to cause its Subsidiaries and their respective Representatives to, reasonably cooperate in connection with the debt financing arrangements of Parent (the "Debt Financing") as may be reasonably requested by Parent at Parent's sole cost and expense (including all documented out-of-pocket costs and expenses (including reasonable and documented attorneys' fees) in accordance with Section 6.8(c)) (other than (x) costs and expenses incurred in connection with the preparation of historical financial statements, (y) any ordinary course amounts payable to employees of Seller, the Company or its Subsidiaries with respect to services provided prior to the Closing Date, and (z) any other ordinary course amounts that would have been incurred in connection with the transactions contemplated hereby regardless of any debt financing established in connection herewith (clauses, (x)-(z), "Seller Costs"), which, in each case, for the avoidance of doubt, shall be at the expense of the Company), including by (A) having appropriate members of senior management of the Company participate in a customary and reasonable number of meetings, presentations, due diligence sessions, drafting sessions and sessions with prospective lenders, (B) furnishing Parent and its lenders with available financial and other pertinent information regarding the Company and its Subsidiaries, (C) assisting Parent and its lenders in the preparation of lender presentations, private placement memoranda, bank information memoranda and similar documents, and (D) reasonably facilitating the taking of all corporate actions by the Company and its Subsidiaries necessary or advisable in connection with entering into definitive financing documents and otherwise necessary to consummate the Debt Financing, provided that no such definitive agreements or corporate actions shall be effective until the Closing unless consented to by the Company and its Subsidiaries (which consent may be withheld in the Company's sole discretion).
- (b) Notwithstanding the foregoing, (i) such requested cooperation shall not unreasonably interfere with the ongoing operations of the Company and its Subsidiaries, or require the Company or its Subsidiaries to agree to pay any fees, reimburse any expenses or otherwise incur any liability or give any indemnities in connection with the Debt Financing prior to the Closing for which it is not subject to reimbursement pursuant to this <u>Section 6.8</u> and (ii) nothing in this <u>Section 6.8</u> shall require cooperation to the extent that it would (A) cause any breach of this

Agreement (B) reasonably be expected to conflict with or violate its organizational documents, any material agreement or any applicable law or (C) waive any attorney-client or other legal privilege of the Company. The Company's obligations set forth in this <u>Section 6.8</u> shall not include any obligation on the part of the Company to obtain or provide Regulation S-X compliant financial statements (historical, pro forma or otherwise) (provided, that the Company shall, to the extent reasonably requested pursuant to this <u>Section 6.8</u>, provide information which may be utilized by Parent in the preparation of such financial statements)

- (c) Each of Parent and Merger Sub shall: (i) promptly upon the written request of the Company reimburse the Company and its Subsidiaries and any of their respective Affiliates and representatives for all of the reasonable and documented out-of-pocket costs and expenses (including reasonable attorneys' fees) incurred by the Company and its Subsidiaries and any of their respective Affiliates and representatives in connection with this Section 6.8, in each case, other than Seller Costs; and (ii) indemnify and hold harmless the Company and its Subsidiaries and any of their respective Affiliates and representatives from and against all claims, losses, damages, injuries, liabilities, judgments, awards, interest, penalties, fines, Taxes, costs (including cost of investigation), expenses (including reasonable and documented out-of-pocket attorneys' fees) or settlement payments suffered or incurred by any of them in connection with the arrangement of the Debt Financing and any information used in connection therewith, except in each case, to the extent suffered or incurred as a result of the bad faith, gross negligence, Fraud or willful misconduct (as determined in a final non-appealable judgment by a court of competent jurisdiction) of the Company or any of its Affiliates or its or its Affiliates' representatives.
- Notwithstanding anything in this Agreement to the contrary (including this Section 6.8), none of the Company, its Subsidiaries or any of its Affiliates or representatives shall: (i) be required to pay any commitment or other fee or reimburse any expenses in connection with the Debt Financing prior to the Closing; (ii) be required to incur or assume any liability or give any indemnity in connection with the Debt Financing prior to the Closing; (iii) be required to take any action (other than customary authorization and representation letters) that would require any director, officer or employee of the Company or any of its Subsidiaries to execute any document, agreement, certificate or instrument in connection with the Debt Financing (other than customary authorization and representation letters) except as may be effective at or after the Closing; (iv) deliver or obtain opinions of internal or external counsel prior to the Closing; (v) provide access to or disclose information where the Company determines that such access or disclosure could jeopardize the attorney-client privilege or contravene any Law or material contract, (vi) waive or amend any terms of this Agreement or any other material contract to which the Company or its Subsidiaries is party or (vii) take any action that would result in the Company or any of its Subsidiaries incurring any liability (other than in respect of customary authorization and representation letters) with respect to the Debt Financing prior to Closing or cause any director, officer or employee of Company or any of its Subsidiaries to incur any personal liability in connection with the Debt Financing. Notwithstanding anything in this Agreement to the contrary, no action, liability or obligation (including any obligation to pay any commitment or other fees or reimburse any expenses, but other than in respect of customary authorization and representation letters) of the Company, its Subsidiaries or any of their respective representatives under any certificate, agreement, arrangement, document or instrument relating to the Debt Financing (other than customary authorization and representation letters) shall be effective until (or that is not contingent upon) the Closing.

- (e) For the avoidance of doubt, the parties hereto acknowledge and agree that the provisions contained in this <u>Section 6.8</u>, represent the sole obligation of the Company, its Subsidiaries and Affiliates and their respective representatives with respect to cooperation in connection with the arrangement of the Debt Financing and no other provision of this Agreement (including the exhibits and schedules hereto) shall be deemed to expand or modify such obligations.
- (f) For the avoidance of doubt, all material non-public information regarding Parent and its Subsidiaries provided to the Company or its representatives shall be kept confidential by them in accordance with the Confidentiality Agreement.
- Exclusivity. The Company agrees that from the date hereof and until the earlier of the 6.9 Closing Date and the date that this Agreement is terminated in accordance with its terms, none of the Acquired Companies or any of the Company Equityholders shall, and the Company shall instruct its representatives not to, directly or indirectly, (i) provide any non-public information to any third party (including via access to any data room or other records) other than Parent and its representatives with respect to any Company Transaction, (ii) solicit, initiate or encourage proposals, offers or inquiries from a third party other than Parent with respect to any Company Transaction, (iii) participate in any negotiations or discussions with any third party other than Parent and its agents and representatives with respect to any Company Transaction or (iv) enter into a letter of intent or other agreement with a third party other than Parent with respect to any Company Transaction. The Company shall notify Parent promptly, but in any event within two Business Days if any Person makes any proposal, offer, inquiry or contact with respect to a Company Transaction, which notification shall include the identity of the Person making such proposal, offer, inquiry or contact and the proposed terms thereof. To the extent the Company is prohibited by a confidentiality obligation from providing the information in the preceding sentence, the Company shall not be required to provide Parent with the identity of the relevant Person but shall nevertheless be required to provide the proposed economic terms.
- 6.10 <u>R&W Insurance Policy</u>. At or before the Closing, Parent shall take all action necessary to obtain and bind a representations and warranties insurance policy with respect to this Agreement (the "<u>R&W Insurance Policy</u>"). The R&W Insurance Policy shall provide that the insurer shall waive and not pursue any subrogation rights against the Representative or the Company Equityholders, other than in the case of claims of, or causes of action arising from, Fraud. For the avoidance of doubt, Parent is responsible for the costs of obtaining the R&W Insurance Policy.
- 6.11 <u>Transition</u>. The Company agrees that from the date hereof and until the earlier of the Closing Date and the date that this Agreement is terminated in accordance with its terms, Eldar Tuvey and Roy Tuvey shall use commercially reasonable efforts to assist Parent in preparing to effectuate an efficient transition of oversight and management from the Company to Parent. Such services include (i) facilitating information sharing sessions between Company and Parent personnel for integration planning, (ii) advising on how best to communicate to the Company's existing employees, partners and customer base following the Closing, (iii) advising Parent on all operational details necessary for the successful transition of the Company's customer, partner and vendor relationship at Closing, and (iv) providing insight into the operation, management and control of the Company and its historical practices in operations, engineering,

technology, finance, tax, human resources, sales/marketing and legal.

Non-Public Information. All non-public information of Parent provided to the Company pursuant to this Agreement shall be treated as confidential pursuant to the terms of this Section 6.12. Except as otherwise provided herein, the Company shall, and shall cause each of its respective Subsidiaries to, hold in confidence, and the Company shall not, and shall cause any of its respective Subsidiaries to not, disclose any nonpublic information of Parent provided hereunder, including exercising the same degree of care as the Company exercises with its own confidential or proprietary information of a similar nature. The Company and its respective Subsidiaries acknowledge and agree that some of the information provided to the Company and its respective Subsidiaries pursuant to this Section 6.12 may be considered "material non-public information" for purposes of securities Laws, and the Company shall, and shall cause its respective Subsidiaries to, abide by all securities Laws relating to the handling of and acting upon material non-public information of or regarding Parent. The Company shall, and shall cause its respective Subsidiaries to, only use any such non-public information for purposes of consummating the transactions contemplated by this Agreement. The Company shall not, and shall cause its respective Subsidiaries to not, disclose any portion of such non-public information to any person other than its or their and its respective Affiliates' managers, members, stockholders, officers, directors, partners, employees, legal counsel, accountants, advisors, agents and consultants (collectively, "Company Representatives") and who "need to know" such non-public information in order to consummate the transactions contemplated by this Agreement. The Company and its respective Subsidiaries shall be responsible for any breach of this Section 6.12 by any of the Company Representatives, and agrees, at its sole expense, to take commercially reasonable measures to restrain the Company Representatives from prohibited or unauthorized disclosure or use of such non-public information.

#### ARTICLE VII COVENANTS RELATING TO PARENT AND MERGER SUB

- 7.1 <u>Filings; Consents; Etc.</u>
- (a) Parent shall use its reasonable best efforts to take or cause to be taken all actions and to do or cause to be done all things (including the satisfaction of the conditions set forth in Section 8.1 and Section 8.3) that are necessary, proper or advisable in compliance with applicable Laws to consummate and make effective, as soon as reasonably practicable, the transactions contemplated hereby. If any suit or other action is threatened or instituted by any Governmental Authority challenging the validity or legality, or seeking to restrain the consummation of the transactions contemplated by this Agreement, Parent shall use its reasonable best efforts to avoid, resist, resolve or, if necessary, defend such suit or action. Parent shall be responsible for one hundred percent (100%) of the filing fees required in order to make any filing under this Section 7.1 or under Section 6.2(b).
- (b) In addition to and without limiting the foregoing, Parent (or its Affiliates, as applicable) undertakes and agrees to file (i) as soon as practicable, and in any event not later than the fifth (5th) Business Day after the date hereof, a Notification and Report Form under the HSR Act with the United States Federal Trade Commission and the Antitrust Division of the United States Department of Justice, and (ii) Parent (or its Affiliates, as applicable) shall (i) make

a reasonable response to any inquiries or requests received from any Governmental Authority for additional information or documentation, and (ii) use its reasonable best efforts to cause the waiting periods or other requirements under the HSR Act to terminate or expire at the earliest possible date. Parent (or its Affiliates, as applicable) shall (A) promptly notify the Company of any written communication to Parent or its Affiliates from any Governmental Authority and, subject to applicable Law, permit the Company to review in advance any proposed written communication to any of the foregoing (and consider in good faith the reasonable views of the Company in connection therewith where such views are provided in a timely manner); (B) not agree to participate, or to permit its Affiliates to participate, in any substantive meeting or substantive pre-arranged discussion with any Governmental Authority in respect of any filings, investigation or inquiry concerning this Agreement unless it consults with the Company in advance and, to the extent permitted by such Governmental Authority, gives the Company the opportunity to attend and participate thereat; and (C) furnish the Company with copies of all correspondence, filings and communications (and memoranda setting forth the substance thereof) between it and its Affiliates and their respective representatives on the one hand, and any Governmental Authority or members of its staff on the other hand, with respect to this Agreement; provided, however, that Parent shall be permitted to designate confidential information as 'Outside Antitrust Counsel Only', and remove or redact any information or contents that are protected by legal privilege and restrict the Company's access to such materials accordingly. Parent shall not be required to share a copy of its filing made under the HSR Act. Parent shall specifically not request early termination of the waiting period under the HSR Act.

7.2 <u>Publicity.</u> On and after the date hereof and through the Closing Date, Parent shall consult with the Company before issuing any press release or making any public announcement relating to the subject matter of this Agreement, and Parent shall not issue any press release or make any public announcement relating to the subject matter of this Agreement without the prior written consent of the Company (which consent shall not be unreasonably withheld, conditioned or delayed); provided, however, that no such approval shall be necessary to the extent disclosure may be required by applicable Law or the rules and regulations of any applicable stock exchange.

# ARTICLE VIII CONDITIONS PRECEDENT TO THE CLOSING

- 8.1 <u>Conditions Precedent to Each Party's Obligations</u>. The respective obligations of each Party to consummate the transactions contemplated hereby will be subject to the satisfaction, at or prior to the Closing, of all of the following conditions, any one or more of which may be waived in writing at the option of the Party whose obligations to consummate the transactions contemplated hereby are subject thereto:
- (a) <u>No Legal Prohibition</u>. No statute, rule, regulation, ruling, consent, decree, judgment, injunction or order shall be enacted, promulgated, entered or enforced by any court or other Governmental Authority of competent jurisdiction which would prohibit the consummation by such Party of the transactions contemplated hereby.

- (b) <u>HSR Act</u>. The applicable waiting period, together with any extensions thereof, under the HSR Act, and any agreement between the Parties and a Governmental Authority not to consummate the transactions, shall have expired or been terminated.
- (c) <u>Stockholder Approval</u>. This Agreement shall have been adopted by the Requisite Stockholder Approval.
- 8.2 <u>Conditions Precedent to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub under this Agreement to consummate the transactions contemplated hereby will be subject to the satisfaction, at or prior to the Closing, of all of the following conditions, any one or more of which may be waived in writing at the option of Parent:
- Accuracy of Representations and Warranties; Performance of Covenants. (i) Each of the representations and warranties of the Company contained in Sections 4.1, 4.2, 4.3 and 4.5 (collectively, the "Fundamental Representations") shall be true and correct in all material respects (in each case, without giving effect to any "materiality" or Material Adverse Effect qualification or exception contained therein), and (ii) each of the representations and warranties of the Company contained in Article IV of this Agreement other than those listed in clause (i) of this Section 8.2(a) (in each case other than Section 4.7(a), without giving effect to any "materiality" or Material Adverse Effect qualification or exception contained therein) shall be true and correct in all respects, in each case as of the Closing with the same force and effect as though made on and as of the Closing Date (other than those representations and warranties that address matters only as of a particular date or only with respect to a specific period of time, which need be accurate only as of such date or with respect to such period), except, in the case of clause (ii) of this Section 8.2(a), where the failure of such representations and warranties to be so true and correct does not have a Material Adverse Effect. The Company shall have performed and complied with, in all material respects, all covenants and agreements required by this Agreement to be performed or complied with by it on or prior to the Closing. Parent shall have received at the Closing a certificate from the Company, dated as of the Closing Date and executed by the Company, certifying its fulfillment of the conditions set forth in this Section 8.2(a).
- (b) <u>Closing Deliverables</u>. The Company shall have delivered or caused to be delivered each of the items required to be delivered at the Closing pursuant to Section 9.1.
- (c) <u>Requisite Stockholder Approval</u>. The Requisite Stockholder Approval will have been obtained by the date hereof, and holders of no more than 5% of the outstanding Company Shares shall have exercised appraisal or dissenters' rights or commenced appraisal proceedings.
- (d) <u>Material Adverse Effect</u>. No Material Adverse Effect shall have occurred and be continuing as of the Closing.
- 8.3 <u>Conditions Precedent to Obligations of the Company.</u> The obligations of the Company under this Agreement to consummate the transactions contemplated hereby will be subject to the satisfaction, at or prior to the Closing, of all the following conditions, any one or more of which may be waived in writing at the option of the Company:

- (a) Accuracy of Representations and Warranties; Performance of Covenants. The representations and warranties of Parent and Merger Sub contained in Article V of this Agreement shall be true and correct in all material respects as of the Closing with the same force and effect as though made on and as of the Closing Date (other than those representations and warranties that address matters only as of a particular date or only with respect to a specific period of time, which need be accurate only as of such date or with respect to such period). Parent and Merger Sub shall have performed and complied with, in all material respects, all covenants and agreements required by this Agreement to be performed or complied with by Parent and Merger Sub on or prior to the Closing. The Company shall have received at the Closing a certificate from Parent and Merger Sub dated as of the Closing Date and executed by Parent and Merger Sub, certifying the fulfillment of the conditions set forth in this Section 8.3(a).
- 8.4 <u>Frustration of Closing Conditions</u>. No Party hereto may rely on the failure of any condition set forth in <u>Section 8.1</u>, <u>8.1(c)</u> or <u>8.3</u>, as the case may be, if such failure was caused by such Party's failure to comply with any provision of this Agreement.

# ARTICLE IX CLOSING

- 9.1 <u>Deliveries by the Company</u>. At the Closing, the Company shall deliver or cause to be delivered to Parent:
- (a) <u>Board Resolutions</u>. A copy of the resolutions of the Board of Directors of the Company, certified by its Secretary as having been duly and validly adopted and as being in full force and effect, authorizing execution and delivery of this Agreement and performance by the Company of the transactions contemplated hereby.
  - (b) <u>Certificate</u>. The certificate required by <u>Section 8.2(a)</u>.
- (c) <u>Good Standing Certificate</u>. A certificate of good standing, dated not more than five (5) days prior to the Closing Date, with respect to the Company, issued by the Secretary of State of the State of Delaware.
  - (d) <u>Escrow Agreement</u>. The Escrow Agreement validly executed by the Representative.
- (e) <u>FIRPTA Certificate</u>. A certification, under penalties of perjury, stating that interests in the Company are not and have not been during the relevant period United States real property interests, as defined in Section 897(c)(2) of the Code, and notice to the IRS, in accordance with Treasury Regulations Section 1.1445-2(c)(3) and 1.897-2(h); <u>provided</u>, that, in the event the Company fails to deliver such certification, the sole recourse of Parent and Merger Sub shall be to withhold on payments of Merger Consideration to the extent required by Law.
- (f) <u>Resignation Letters</u>. Written resignation letters, in form and substance reasonably satisfactory to Parent, effective as of immediately prior to the Closing, from those directors, managers and officers of the Company and any of the Company Subsidiaries identified by Parent.

- 9.2 <u>Deliveries by Parent and Merger Sub</u>. Parent will deliver or cause to be delivered to the Company or the Company Equityholders, as applicable:
- (a) <u>Merger Consideration</u>. Payment of the Merger Consideration and other amounts payable as provided in <u>Section 3.1</u>.
- (b) <u>Board Resolutions</u>. A copy of the resolutions of the Board of Directors or Managers, as applicable, of each of Parent and Merger Sub, certified by its Secretary as having been duly and validly adopted and as being in full force and effect, authorizing execution and delivery of this Agreement and performance by Parent and Merger Sub of the transactions contemplated hereby.
  - (c) <u>Certificate</u>. The certificate required by Section <u>8.3(a)</u>.
- (d) <u>Escrow Agreement</u>. The Escrow Agreement validly executed by Parent and the Escrow Agent.

# ARTICLE X POST-CLOSING COVENANTS

#### 10.1 Tax Covenants.

- (a) All transfer, documentary, sales, use, stamp, registration or similar Taxes and fees applicable to, imposed upon or arising out of the Merger ("<u>Transfer Taxes</u>") shall be borne 50% by the Company Equityholders and 50% by the Parent. The party required by law shall, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes, fees and charges, and, if required by applicable Law, the other Parties will, and will cause their Affiliates to, join in the execution of any such Tax Returns and other documentation.
- (b) Parent and the Representative shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the preparation and filing of any Tax Return and any audit, litigation or other proceeding with respect to Taxes. Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information which are reasonably relevant to any such Tax Return, audit, litigation or other proceeding or any tax planning and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Parent and the Representative further agree, upon request, to use their commercially reasonable efforts to obtain any certificate or other document from any Governmental Authority or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed as a result of the transactions contemplated by this Agreement (including, but not limited to, any Transfer Taxes).
- (c) Parent shall prepare or cause to be prepared and file or cause to be filed all Tax Returns for the Acquired Companies for any Pre-Closing Tax Period that are filed or required to be filed after the Closing Date. Any such Tax Returns shall be prepared in a manner consistent with the past practices of the Acquired Company, as applicable, except as required by applicable Law. At least twenty (20) days prior to filing any such Tax Return showing a material amount of Tax due or that is reasonably likely to result in the Company Equityholders having an indemnification obligation, Parent shall provide a copy of such Tax Return to the Representative

for its review and shall consider in good faith any reasonable comments received from Representative no later than five (5) days prior to the due date of such Tax Return.

- (d) Tax Refunds. All refunds (or credits for overpayments) of Taxes attributable to Pre-Closing Tax Period that have been paid by the Company prior to Closing or indemnified by the Company Equityholders pursuant to this Agreement, except to the extent such refund or credit included in the calculation of the Current Assets, shall be for the benefit of the Company Equityholders. Promptly upon receipt of any such Tax refund (or credits for overpayments), and in no event later than ten (10) days after receipt by Parent or any of their Subsidiaries (including the Acquired Companies), Parent will, and will cause its Subsidiaries to, deliver and pay over such Tax refunds (or credits for overpayments), including any interest thereon, to the Payments Administrator (for further distribution to the Company Equityholders), net of the amount of any Taxes or reasonable out-of-pocket expenses incurred by Parent or any of its Subsidiaries in obtaining such refund (or credits for overpayments).
- (e) Certain Post-Closing Actions. Without the consent of the Representative (not to be unreasonably withheld, conditioned or delayed), Parent shall not, except as otherwise required by Law or contemplated by this Agreement, amend any Tax Return or election made in connection with such Tax Return with respect to the Acquired Companies for any Pre-Closing Tax Period or change any method of accounting for Tax purposes or Tax accounting period with respect to any Pre-Closing Tax Period, in each case that would reasonably be expected to give rise to a claim against the Company Equityholders for indemnification of Pre-Closing Taxes. Parent may voluntarily approach a Tax authority to correct a Tax filing practice of the Acquired Companies for any Pre-Closing Tax Periods after reasonable consultation with the Representative, provided that Parent shall keep the Representative reasonably informed of any material developments in such discussions and shall consider the Representative's comments in good faith.

### 10.2 <u>Director and Officer Liability and Indemnification.</u>

- (a) For a period of six (6) years after the Closing, Parent shall not, and shall ensure that each of the Acquired Companies does not, amend, alter, repeal or modify any provision in any of the Acquired Companies' Organizational Documents and any indemnification agreements to which the Acquired Companies are bound relating to, in each case, the exculpation or indemnification of any officers and directors in any way that diminishes or adversely affects the indemnification or exculpation provided therein (unless required by Law), it being the intent of the Parties that the officers and directors of any of the Acquired Companies who were officers and directors at any time prior to the Closing (each, a "D&O Indemnified Person") shall continue to be entitled to such exculpation and indemnification to the fullest extent provided for under applicable Law.
- (b) At the Closing, Parent shall, or shall cause the Company to, obtain irrevocable "tail" insurance policies naming the D&O Indemnified Persons as direct beneficiaries, with a claims period of at least six (6) years from the Closing Date, from an insurance carrier with the same or better credit rating as the Company's current insurance carrier with respect to directors' liability insurance and in an amount and scope at least as favorable as the Company's existing policies, with respect to matters existing, occurring or arising at or prior to the Closing Date. The cost of such tail policy or policies will be borne by the Company's equityholders and will be a part

of Transaction Expenses. Parent shall not, and shall cause the Company not to, cancel or change such insurance policies in any respect.

(c) In the event Parent, the Company or any of their respective successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity in such consolidation or merger, or (ii) transfers all or substantially all of its properties and assets to any Person, then and in either such case proper provision shall be made so that the successors and assigns of Parent or the Company, as the case may be, shall assume the obligations set forth in this <u>Section 10.2</u>. The provisions of this <u>Section 10.2</u> shall survive the consummation of the Merger and are expressly intended to benefit each of the D&O Indemnified Persons.

### 10.3 <u>Employee Matters</u>.

- (a) Except as set forth in Section 10.3(d), for a period of one (1) year following the Closing (or, if earlier with respect to any Continuing Employee (as such term is defined below), the date that such Continuing Employee terminates employment from the Surviving Corporation and its Affiliates), Parent shall provide (or cause the Surviving Corporation or another Affiliate of Parent to provide) to employees of the Surviving Corporation or any other Affiliate of Parent who were Company Employees immediately prior to the Effective Time ("Continuing Employees") with (i) base compensation and target annual cash bonus opportunities that are not less than the base compensation and target annual cash bonus opportunities received by the Continuing Employees immediately prior to the Effective Time, and, (ii) employee benefits (other than equity or equity-based compensation, change in control, retention, defined benefit pension and nonqualified deferred compensation plans and arrangements) ("Employee Benefits") that are substantially comparable in the aggregate to the Employee Benefits provided by Parent to its similarly situated employees.
- (b) For purposes of determining eligibility to participate, vesting of retirement benefits (but not benefit accrual), and for purposes of determining future vacation accruals under any employee benefit plans of Parent, the Surviving Corporation or any of their respective Subsidiaries, Continuing Employees shall receive service credit for service with the Acquired Companies to the same extent and for the same purposes such service credit was granted under the Employee Benefit Plans or PEO Plans set forth under Schedule 4.15(a), provided that such credit does not result in a duplication of benefits or coverage. Parent and the Surviving Corporation shall use commercially reasonable efforts to (i) waive all waiting periods with respect to participation and coverage requirements applicable to the Continuing Employees under any welfare benefit plans that such employees may be eligible to participate in after the Effective Time, other than waiting periods that are already in effect with respect to such employees and that have not been satisfied as of the Effective Time under any welfare benefit plan maintained for the Continuing Employees immediately prior to the Effective Time and (ii) provide each Continuing Employee with credit for any co-payments and deductibles paid prior to the Effective Time during the plan year in which the Closing Date occurs in satisfying any applicable deductible or out-of-pocket requirements for the plan year in which the Closing Date occurs under any group health plans that such employees are eligible to participate in after the Effective Time.
  - (c) Parent shall provide certain Continuing Employees as selected by Parent

and mutually agreed with the Chief Executive Officer of the Company with awards of Parent RSUs having an aggregate grant date fair market value equal to at least \$15,000,000. Each Continuing Employee who receives an award of Parent RSUs shall enter into a restricted stock unit agreement ("RSU Agreement") that shall provide for vesting over a four (4) year period following the date of grant, and shall otherwise have terms and conditions consistent with Parent's standard form of RSU Agreement.

- (d) Notwithstanding anything to the contrary in this <u>Section 10.3</u>, Company's and Parent's obligations under this <u>Section 10.3</u> shall not (i) apply to any individual who is furloughed, temporarily laid off, or suffers a termination of employment or reduction in hours or benefits because of COVID-19-related circumstances at any time, or (ii) limit the Company's and Parent's right, in its sole discretion, to furlough, temporarily layoff, terminate the employment of, or reduce the hours or benefits of, any employee.
- (e) The Acquired Companies shall (or shall cause the applicable plan sponsor to), (i) adopt written resolutions (or take other necessary and appropriate action(s)) to terminate the Acquired Companies' participation in the TriNet 401(k) Plan (the "401(k) Plan"), effective at least one (1) Business Day prior to the Closing Date (the "401(k) Plan Termination Date"), in compliance with its terms and the requirements of applicable Law; and (ii) on or prior to the 401(k) Plan Termination Date, or if necessary, as soon as administratively practicable following the 401(k) Plan Termination Date, make all employee and employer contributions to the 401(k) Plan on behalf of the Continuing Employees with respect to compensation paid to the Continuing Employees on or before the 401(k) Plan Termination Date, including such contributions that would have been made on behalf of the Continuing Employees had the transactions contemplated by this Agreement not occurred (regardless of any service or end-of-year employment requirements) but prorated for the portion of the plan year that ends on the Closing Date.
- (f) The Company and Parent acknowledge and agree that all provisions contained in this Section 10.3, (i) are included for the sole benefit of the respective parties and shall not create any right in any other Person, including any employees, former employees, any participant in any Employee Benefit Plan, PEO Plan or any beneficiary thereof or any right to continued employment (or any term or condition of employment) with an Acquired Company, Parent or any Subsidiary of Parent, (ii) shall not require Parent, any Acquired Company or any Subsidiary of Parent to continue or amend any particular benefit plan after the consummation of the transactions contemplated by this Agreement for any employee or former employee of any Acquired Company, and shall not alter or limit the Parent's, the Surviving Corporation's, or any of their Affiliates' ability to amend, modify or terminate any particular benefit plan, program, agreement or arrangement in accordance with its terms and applicable Law, and (iii) shall not be construed to establish, amend, or modify any benefit plan, program, agreement or arrangement.

# ARTICLE XI TERMINATION

11.1 <u>Termination of Agreement</u>. Certain of the Parties may terminate this Agreement as provided below:

- (a) Parent and the Company may terminate this Agreement by mutual written consent at any time prior to the Closing;
- (b) Parent or the Company may terminate this Agreement by written notice to the other Party prior to the Closing, if:
  - (i) the Merger has not been consummated on or before the date that is ninety (90) days after the date of this Agreement (the "End Date"); provided, that neither Parent nor the Company will be entitled to terminate this Agreement pursuant to this Section 11.1(b)(i) if such Party's willful breach of this Agreement has hindered the consummation of the transactions contemplated by this Agreement on or prior to the End Date; provided, further, that Company and Parent each shall not have the right to terminate this Agreement pursuant to this Section 11.1(b)(i) if the extending Party is pursuing specific performance of, or other equitable remedies with respect to, Parent's obligations hereunder; or
  - (ii) (A) there shall be any Law that shall make illegal the consummation of the Merger or (B) a Governmental Authority shall have enacted, issued, promulgated or entered an order or injunction prohibiting, enjoining or restraining the Merger and such order or injunction shall have become final and non-appealable; <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 11.1(b)(ii)</u> shall not be available to a Party if such order or injunction was primarily due to the failure of such party to comply with this Agreement;
- (c) Parent may terminate this Agreement by giving written notice to the Company at any time prior to the Closing if (i) a breach of any representation or warranty or failure to perform any covenant or agreement on the part of the Company set forth in this Agreement shall have occurred that causes or would reasonably be expected to cause any of the conditions set forth in Section 8.1 or Section 8.1(c) not to be satisfied at Closing and (ii) such breach or failure to perform is incapable of being cured or, if curable, is not cured by the Company by the earlier of (A) the End Date and (B) within twenty (20) Business Days after giving the Company written notice of such breach or failure; provided that, at the time of such termination, neither Parent nor Merger Sub shall be in breach of this Agreement which causes or would reasonably be expected to cause a condition to Closing set forth in Section 8.1 or Section 8.3 to fail to be satisfied; and
- (d) The Company may terminate this Agreement by giving written notice to Parent at any time prior to the Closing if (i) a breach of any representation or warranty or failure to perform any covenant or agreement on the part of Parent or Merger Sub set forth in this Agreement shall have occurred that causes or would reasonably be expected to cause any of the conditions set forth in Section 8.1 or Section 8.3 not to be satisfied at Closing and (ii) such breach or failure to perform is incapable of being cured or, if curable, is not cured by Parent or Merger Sub by the earlier of (A) the End Date and (B) within twenty (20) Business Days after the giving to Parent or Merger Sub written notice of such breach or failure; provided that, at the time of such termination, the Company shall not be in breach of this Agreement which causes or would reasonably be expected to cause a condition to Closing set forth in Section 8.1 or Section 8.1(c) to fail to be satisfied.

The Party seeking to terminate this Agreement pursuant to this <u>Section 11.1</u> (other than <u>Section 11.1(a)</u>) shall give prompt written notice of such termination to the other Parties.

11.2 <u>Effect of Termination</u>. In the event of termination of this Agreement, by either Parent or the Company in accordance herewith, the provisions of this Agreement will immediately become void and of no further force or effect (other than this <u>Section 11.2</u>, and <u>Article XII</u> hereof, which will survive the termination of this Agreement in accordance with their terms); provided, however, that the last sentence of <u>Section 6.4</u> and the Confidentiality Agreement referred to therein will survive the termination of this Agreement according to its terms. Nothing herein will relieve any Party from any liability or damages arising out of Fraud or any willful and material breach by such Party of its representations, warranties, covenants or agreements set forth in this Agreement occurring prior to such termination, in which case, the non-breaching party shall be entitled to all rights and remedies available at law or in equity.

#### ARTICLE XII MISCELLANEOUS

- 12.1 <u>Expenses</u>. Except as otherwise expressly set forth in this Agreement, each of Parent, Merger Sub, the Company and the Representative (solely on behalf of the Company Equityholders) will bear his, her or its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the transactions contemplated hereby; provided, however, that in the event the Merger is consummated, Transaction Expenses shall be borne and paid as provided in this Agreement.
- 12.2 <u>No Third Party Beneficiaries.</u> Except as otherwise set forth in <u>Section 12.9</u>, <u>Section 12.15(c)</u> and <u>Section 12.16</u>, this Agreement and its provisions are for the sole benefit of the Parties, Company Equityholders and the D&O Indemnified Persons and their respective successors and permitted assigns and shall not confer any rights or remedies upon any Person.
- 12.3 <u>Entire Agreement</u>. This Agreement, including the Schedules and Exhibits hereto, the Escrow Agreement, the Confidentiality Agreement, the Payments Agreement, the Support Agreements, the Employment Agreements and the other documents referred to herein, constitute the entire agreement among the Parties and supersede any prior understandings, agreements or representations by or among the Parties, written or oral, to the extent they relate in any way to the subject matter hereof. Notwithstanding any oral agreement or course of conduct of the Parties or their representatives to the contrary, no Party shall be under any legal obligation to enter into or complete the transactions contemplated hereby unless and until this Agreement shall have been executed and delivered by each of the Parties.
- 12.4 <u>Succession and Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No Party may assign either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of Parent, the Representative and the Company, provided, that Parent may assign this Agreement and any of its rights and interests hereunder, without the consent of any Person, to an Affiliate (including Affiliates formed before or after the date of this Agreement), to any purchaser of all or any portion of the assets or business of Parent or any of its Affiliates, or as collateral security to any lender or prospective lender to Parent or the

Company or to any purchaser of all or substantially all of the equity or assets of Parent or any of its Subsidiaries (including the Company and the Company Subsidiaries following the Closing); provided, further that no assignment to any such Affiliate, lender or purchaser shall in any way affect Parent's obligations or liabilities under this Agreement.

- 12.5 <u>Counterparts</u>. This Agreement may be executed in counterparts (including by means of electronic transmission), each of which shall be deemed an original but all of which together will constitute one and the same instrument.
- 12.6 <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.
- 12.7 <u>Notices</u>. All notices, requests, demands, claims and other communications hereunder will be in writing. Any notice, request, demand, claim or other communication hereunder shall be deemed duly given (a) when delivered personally to the recipient, (b) one (1) Business Day after being sent to the recipient on a priority basis by reputable overnight courier service (charges prepaid), or (c) if sent by electronic mail, when received if received before 5:00 p.m. local time of the recipient on a Business Day, and otherwise on the next following Business Day, in each case addressed to the intended recipient as set forth below:
  - (a) If to Parent, Merger Sub or the Surviving Corporation (after the Closing), addressed to it at:

JAMF Software, LLC 100 Washington Ave. S., Suite 1100 Minneapolis, MN 55401 Attention: Jeff Lendino, Chief Legal Officer Email: jeff.lendino@jamf.com

With a copy to (which shall not constitute notice):

Vista Equity Partners Management, LLC Four Embarcadero Center, 20th Floor San Francisco, CA 94111 Attention: Christina Lema, Email: clema@vistaequitypartners.com]

And

Kirkland & Ellis LLP c/o 3330 Hillview Ave.

Palo Alto, CA 94304

Email: marc.browning@kirkland.com and dilen.kumar@kirkland.com

Attention: Marc D. Browning, P.C. and Dilen Kumar

(b) If to the Company (prior to the Closing), addressed to it at:

Wandera, Inc.

Attention: Eldar Tuvey and Roy Tuvey

Email: eldar.tuvey@wandera.com and roy.tuvey@wandera.com

With a copy to (which shall not constitute notice):

Fenwick & West LLP 801 California St., Mountain View, CA 94041

Mountain view, CA 34041

Attention: Stephen M. Fisher and Sayre Stevick

Email: sfisher@fenwick.com and sstevick@fenwick.com

(c) If to the Representative, or to the Company Equityholders after Closing, addressed to it at:

Shareholder Representative Services LLC 950 17<sup>th</sup> Street, Suite 1400 Denver, CO 80202 Email: deals@srsacquiom.com

Attention: Managing Director

Any Party may change the address to which notices, requests, demands, claims and other communications hereunder are to be delivered by giving the other Parties notice in the manner set forth in this <u>Section 12.7</u>.

- 12.8 Governing Law; Waiver of Jury Trial.
- (a) This Agreement and any Legal Proceeding of any kind or nature (whether at law or in equity, based upon contract, tort or otherwise) that is in any way related to this Agreement or any of the transactions related hereto (including the interpretation, construction, validity and enforcement of this Agreement, or the negotiation, execution or performance of any of the transactions related hereto (including any Legal Proceeding based upon, arising out of, or related to any representation or warranty expressly made in this Agreement)) shall be governed by and construed in accordance with the domestic Laws of the State of Delaware, including its statutes of limitations, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware and without regard to any borrowing statute that would result in the application of the statute of limitations of any other jurisdictions. In furtherance of the foregoing, the laws of the State of Delaware will control even if under such jurisdiction's

choice of law or conflict of law analysis, the substantive law of some other jurisdiction would ordinarily or necessarily apply.

- (b) Each Party hereto hereby irrevocably and unconditionally waives any right such Party may have to a trial by jury in respect of any litigation directly or indirectly arising out of or relating to this Agreement, the other agreements contemplated hereby or the transactions contemplated by this Agreement. Each Party certifies and acknowledges that (i) no representative, agent or attorney of any other Party has represented, expressly or otherwise, that such other Party would not, in the event of litigation, seek to enforce the foregoing waiver, (ii) such Party understands and has considered the implications of this waiver, (iii) such Party makes this waiver voluntarily, and (iv) such Party has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this Section 12.8.
- (c) The Parties submit to the exclusive jurisdiction of the Chancery Court of the State of Delaware and any state appellate court therefrom within the State of Delaware (or if the Chancery Court of the State of Delaware declines to accept jurisdiction over a particular matter, any federal court sitting in the State of Delaware and any federal appellate court therefrom) in respect of the interpretation and enforcement of the provisions of this Agreement and any related agreement, certificate or other document delivered in connection herewith and by this Agreement waive, and agree not to assert, any defense in any action for the interpretation or enforcement of this Agreement and any related agreement, certificate or other document delivered in connection herewith that they are not subject to such jurisdiction or that such action may not be brought or is not maintainable in such courts or that this Agreement may not be enforced in or by such courts, that the action is brought in an inconvenient forum, or that the venue of the action is improper.
- (d) Each Party agrees that service in person or by certified or by nationally recognized overnight courier to its address set forth in <u>Section 12.7</u> shall constitute valid in personam service upon such Party and its successors and assigns in any arbitration proceeding commenced pursuant to this <u>Section 12.8</u>. Each Party hereby acknowledges that this is a commercial transaction, that the foregoing provisions for consent to arbitration, service of process and waiver of jury trial have been read, understood and voluntarily agreed to by each Party and that by agreeing to such provisions each Party is waiving important legal rights.

### 12.9 <u>Survival; Acknowledgment; Waiver</u>

(a) Each of the representations and warranties contained in this Agreement and in any certificate delivered pursuant hereto shall terminate at, and shall not survive, the Closing. Each of the covenants or agreements of the Parties contained in this Agreement shall terminate at, and shall not survive, the Closing, except for (i) those covenants that by their terms contemplate performance in whole or in part after the Closing, including, for the avoidance of doubt, the obligation of Parent and the Company to pay (or cause to be paid) the Deferred Merger Consideration pursuant to Section 3.1(e), which shall continue in full force and effect until the last date on which such covenant is to be performed, (ii) Fundamental Representations which shall continue in full force and effect after the Closing until the applicable statute of limitations, and (iii) the indemnification for the Pre-Closing Taxes under Section 12.9(c)(i)(A), which shall continue in full force and effect after the Closing until 60 days following the expiration of the applicable statute of limitations.

- (b) THE REPRESENTATIONS AND WARRANTIES IN <u>ARTICLE IV</u>, AS QUALIFIED BY THE COMPANY DISCLOSURE SCHEDULES, AND THE REPRESENTATIONS AND WARRANTIES BY THE COMPANY PURSUANT TO ANY ANCILLARY AGREEMENTS ENTERED INTO IN CONNECTION WITH THE TRANSACTION, CONSTITUTE THE SOLE AND EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF THE COMPANY TO PARENT AND MERGER SUB IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY, AND EACH OF PARENT AND MERGER SUB UNDERSTANDS, ACKNOWLEDGES AND AGREES THAT ALL OTHER REPRESENTATIONS AND WARRANTIES OF ANY KIND OR NATURE EXPRESS OR IMPLIED (INCLUDING ANY RELATING TO THE FUTURE OR HISTORICAL FINANCIAL CONDITION, RESULTS OF OPERATIONS, ASSETS OR LIABILITIES OF THE COMPANY AND ANY OF THE OTHER ACQUIRED COMPANIES) ARE SPECIFICALLY DISCLAIMED BY THE COMPANY.
- (c) Notwithstanding the foregoing or anything else contained in this Agreement or otherwise, (i) from and after the Closing, the Company Equityholders shall severally (but not jointly), in accordance with its Indemnity Pro Rata Share, indemnify Parent and its Affiliates for (A) any Pre-Closing Taxes paid by Parent or its Affiliates (including any of the Acquired Companies) and (B) any breach of Fundamental Representations, and (ii) the indemnification provisions of this Section 12.9 are the sole and exclusive remedy of Parent and its Affiliates against the Company Equityholders following the Closing for any losses or damages incurred by Parent and its Affiliates (including any of the Acquired Companies), provided that nothing shall restrict the Party's rights to specific performance, or Parent's remedies in the event of, or prevent Parent from bringing a claim alleging, Fraud against the Person committing such Fraud. Notwithstanding anything herein to the contrary, Company Equityholders shall not have any indemnity obligation resulting from a reduction or unavailability of any net operating loss or other similar Tax attribute of any Acquired Company in any post-Closing Tax period (or portion thereof).
- (d) Subject to Section 12.9(c)(ii) in respect of a particular Company Equityholder's own Fraud, the maximum liability of the Company Equityholders for indemnification claims hereunder shall not exceed the aggregate Merger Consideration actually received by such Company Equityholder. If and solely to the extent that an amount or adjustment was already included in the calculation of the Closing Merger consideration (including pursuant to Section 3.4), then the same amount or adjustment may not be recovered under this Section 12.9 as an indemnification payment.
- (e) To the extent Parent is entitled to indemnification payments pursuant to this Agreement, such indemnification payments shall be satisfied:
  - (i) first, from the R&W Insurance Policy, subject to applicable exclusions and limitations under the R&W Insurance Policy;
  - (ii) second, solely to the extent not recoverable from the R&W Insurance Policy, directly from the Company Equityholders (in proportion to each Company Equityholder's Indemnity Pro Rata Share);
    - (iii) thereafter, and solely with respect to losses that are not recoverable

under the R&W Insurance Policy as a result of the limit under the R&W Insurance Policy being satisfied, directly from the Company Equityholders (in proportion to each Company Equityholder's Indemnity Pro Rata Share); and

- (iv) in all cases, reduced by actual insurance recoveries actually received by Parent and its Affiliates, net of any insurance premiums (including any increase thereof) and costs of collection.
- 12.10 Amendments and Waivers. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by Parent, the Company and the Representative. No waiver by any Party of any provision of this Agreement or any default, misrepresentation or breach of warranty or covenant hereunder, whether intentional or not, shall be valid unless the same shall be in writing and signed by the Party making such waiver, nor shall such waiver be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.
- 12.11 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable Law by which this Agreement is governed, such invalidity, illegality or unenforceability shall not affect any other provision; provided that such provision shall be construed to give effect to the Parties' intent regarding such provision to the maximum extent permitted by applicable Law.
- 12.12 <u>Construction; Disclosure</u>. The language used in this Agreement shall be deemed to be the language chosen by the Parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any Person. The specification of any dollar amount or the inclusion of any item in the representations and warranties contained in this Agreement or the Schedules or Exhibits attached hereto is not intended to imply that the amounts, or higher or lower amounts, or the items so included, or other items, are or are not required to be disclosed (including whether such amounts or items are required to be disclosed as material or threatened) or are within or outside of the Ordinary Course of Business, and no Party shall use the fact of the setting of the amounts or the fact of the inclusion of any item in this Agreement or the Schedules or Exhibits in any dispute or controversy between the Parties as to whether any obligation, item or matter not described or included in this Agreement or in any Schedule or Exhibit is or is not required to be disclosed (including whether the amount or items are required to be disclosed as material or threatened) or is within or outside of the Ordinary Course of Business for purposes of this Agreement. Prior to the Closing, the Company shall have the right from time to time to supplement, modify or update the Company Disclosure Schedules to reflect actions, omissions, circumstances, occurrences and/or developments occurring after the execution and delivery of this Agreement and, except for purposes of Section 8.1(c), any such supplements, modifications and updates shall supplement, modify and amend the Company Disclosure Schedules for all purposes. From and after the time of any such supplement, modification and/or update, references to the Company Disclosure Schedules shall be references to the Company Disclosure Schedules as so supplemented, modified and/or updated. The information contained in this Agreement and in the Company Disclosure Schedules and Exhibits hereto is disclosed solely for purposes of this Agreement, and no information contained herein or therein shall be deemed to be an admission by

any Party hereto to any third party of any matter whatsoever (including any violation of law or breach of contract).

- 12.13 <u>Incorporation of Exhibits and Schedules</u>. The Exhibits and Schedules identified in this Agreement are incorporated herein by reference and made a part hereof.
- 12.14 Specific Performance. The Parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed by the Parties hereto in accordance with their specific terms or were otherwise breached. It is accordingly agreed that Parent and Merger Sub, on the one hand, and the Company, on the other hand, shall be entitled specific performance of the terms hereof, including an injunction or injunctions to prevent breaches of this Agreement by the other (as applicable) and to enforce specifically the terms and provisions of this Agreement in the courts described in Section 12.8, this being in addition to any other remedy to which such Party is entitled at law or in equity. Each of the Parties hereby waives (a) any defenses in any action for specific performance that a remedy at law would be adequate and (b) any requirement under any Law to post a bond or other security as a prerequisite to obtaining equitable relief. Without limiting the generality of the foregoing, and notwithstanding Section 6.8 hereof, Parent and Merger Sub acknowledge that the Company shall be entitled specific performance of Parent's and Merger Sub's obligation to consummate the transactions contemplated hereby regardless of whether the Debt Financing has been funded or will be funded at the Closing, and that Parent and Merger Sub shall use available cash, debt or equity financing sources to pay the amounts required to be paid by Parent or Merger Sub hereunder. If, prior to the End Date, any Party brings any action, in each case in accordance with Section 12.8, to enforce specifically the performance of the terms and provisions hereof by any other Party, the End Date shall automatically be extended by (x) the amount of time during which such action is pending, plus ten (10) Business Days or (y) such other later date established by the Delaware court presiding over such action.

#### 12.15 <u>Representative</u>.

- (a) <u>Designation</u>. The Representative is hereby authorized to serve as the representative of each Company Equityholders with respect to the matters expressly set forth in this Agreement to be performed by the Representative.
- (b) <u>Authority.</u> Each Company Equityholder, by his, her or its acceptance of any portion of the Merger Consideration, as well as through the execution and delivery of his, her or its Stockholder Letter of Transmittal, Option Cancellation Agreement or Warrant Cancellation Agreement pursuant to the terms of this Agreement, irrevocably appoints the Representative as of the Closing as the representative, agent, proxy and attorney-in-fact for such Company Equityholder for all purposes of this Agreement, the Escrow Agreement and the Payments Agreement (including the full power and authority on such Company Equityholder's behalf (i) to consummate the transactions contemplated herein, (ii) to pay such Company Equityholder's expenses incurred in connection with the negotiation and performance of this Agreement (whether incurred on or after the date hereof), (iii) to make any determinations and settle any matters in connection with the adjustment of the Merger Consideration contemplated by <u>Article III</u>, (iv) to cause the Payments Administrator to distribute any funds payable by Parent hereunder which are for the account of the Company Equityholders, (v) to deduct and/or hold back any funds which

may be payable to any Company Equityholder pursuant to the terms of this Agreement in order to pay, or establish a reserve for, any amount which may be payable by such Company Equityholder hereunder, (vi) to execute and deliver on behalf of such Company Equityholder any amendment to the terms hereof, (vii) to take all other actions to be taken by or on behalf of such Company Equityholder in connection herewith, (viii) to retain funds for reasonably anticipated expenses and liabilities, and (ix) to do each and every act and exercise any and all rights which such Company Equityholder or the Company Equityholders collectively are permitted or required to do or exercise under this Agreement, the Escrow Agreement and the Payments Agreement). All decisions and actions by the Representative made in accordance with the authority granted to it hereunder, including the defense or settlement of any claims for losses for which the Parent, Merger Sub and their respective officers, directors, employees, agents, representatives, successors and permitted assigns may be entitled to indemnification, shall be binding upon all of the Company Equityholders, and no Company Equityholder shall have the right to object, dissent, protest or otherwise contest the same. The Representative Holdback Amount shall be retained by the Representative for such time as the Representative shall determine in its sole discretion.

(c) <u>Replacement</u>. In the event that the Representative becomes unable to perform its responsibilities hereunder or resigns from such position, the Company Equityholders (or, if applicable, their respective heirs, legal representatives, successors and assigns) who held a majority of the voting power represented by the capital stock of the Company issued and outstanding immediately prior to the Effective Time shall select another representative to fill such vacancy, and such substituted representative shall be deemed to be the Representative for all purposes of this Agreement, and the Company Equityholders shall provide notice of the identity of the new Representative to Parent within ten (10) days of such replacement.

### (d) Exculpation.

- (i) Neither the Representative nor any agent employed by it shall incur any liability to any other Person by virtue of the failure or refusal of the Representative for any reason to consummate the transactions contemplated hereby or relating to the performance of its other duties hereunder or any of its omissions or actions with respect thereto. For the avoidance of doubt, Parent and Merger Sub shall be entitled to rely on the acts, to the extent authorized hereunder, of the Representative.
- (ii) The Representative shall have no obligations to make any payments, including on behalf of any Company Equityholder or any other Person. Parent agrees that it will not look to the personal assets of the Representative, acting in such capacity, for the satisfaction of any obligations to be performed by the Company (pre-Closing) or the Company Equityholders. The Representative will incur no liability of any kind with respect to any action or omission by the Representative in connection with the Representative's services pursuant to this Agreement and any agreements ancillary hereto, except in the event of liability primarily resulting from the Representative's gross negligence or willful misconduct. The Representative shall not be liable for any action or omission pursuant to the advice of counsel. The Company Equityholders will indemnify, defend and hold harmless the Representative from and against any and all losses,

liabilities, damages, claims, penalties, fines, forfeitures, actions, fees, costs and expenses (including the fees and expenses of counsel and experts and their staffs and all expense of document location, duplication and shipment) (collectively, "Representative Losses") arising out of or in connection with the Representative's execution and performance of this Agreement and any agreements ancillary hereto, in each case as such Representative Loss is suffered or incurred; provided, that in the event that any such Representative Loss is finally adjudicated to have been primarily caused by the gross negligence or willful misconduct of the Representative, the Representative will reimburse the Company Equityholders the amount of such indemnified Representative Loss to the extent attributable to such gross negligence or willful misconduct. If not paid directly to the Representative by the Company Equityholders, any such Representative Losses may be recovered by the Representative from (i) the Representative Holdback Amount, and (ii) the amounts in the Escrow Funds at such time as remaining amounts would otherwise be distributable to the Company Equityholders; provided, that while this section allows the Representative to be paid from the aforementioned sources of funds, this does not relieve the Company Equityholders from their obligation to promptly pay such Representative Losses as they are suffered or incurred, nor does it prevent the Representative from seeking any remedies available to it at law or otherwise. In no event will the Representative be required to advance its own funds on behalf of the Company Equityholders or otherwise. Notwithstanding anything in this Agreement to the contrary, any restrictions or limitations on liability or indemnification obligations of the Company Equityholders set forth elsewhere in this Agreement (including Section 12.1 and Section 12.16 hereof) are not intended to be applicable to the indemnities provided to the Representative under this Section 12.15. The Company Equityholders acknowledge and agree that the foregoing indemnities will survive the Closing, the resignation or removal of the Representative or the termination of this Agreement.

(iii) The Representative Holdback Amount will be held in with an FDIC-insured institution in a segregated account. The Company Equityholders will not receive any interest or earnings on the Representative Holdback Amount and irrevocably transfer and assign to the Representative any ownership right that they may otherwise have had in any such interest or earnings. The Representative will not be liable for any loss of principal of the Representative Holdback Amount other than as a result of its gross negligence or willful misconduct. The Representative will hold these funds separate from its corporate funds, will not use these funds for its operating expenses or any other corporate purposes and will not voluntarily make these funds available to its creditors in the event of bankruptcy. As soon as practicable following the completion of the Representative's responsibilities, the Representative will deliver any remaining balance of the Representative Holdback Amount to the Payments Administrator for further distribution in accordance with Section 3.4(d)(i)–(ii) (the date of such distribution, the "Representative Holdback Distribution Date"). For tax purposes, the Representative Holdback Amount will be treated as having been received and voluntarily set aside by the Company Equityholders at the time of the Closing.

- (iv) This Section  $\underline{12.15(d)}$  shall survive any termination of this Agreement pursuant to Article XI.
- (e) <u>Irrevocability; Successors</u>. The provisions of this <u>Section 12.15</u> are independent and severable, are irrevocable and coupled with an interest and shall be enforceable notwithstanding any rights or remedies that any Company Equityholder may have in connection with the transactions contemplated by this Agreement. The provisions of this <u>Section 12.15</u> shall be binding upon the heirs, legal representatives, successors and assigns of each Company Equityholder, and any references in this Agreement to a Company Equityholder shall mean and include the successors to the rights of the Company Equityholders hereunder, whether pursuant to testamentary disposition, the laws of descent and distribution or otherwise.
- (f) Access. The Representative shall have reasonable access to relevant information about the Acquired Companies and the reasonable assistance of the Company's, the Company Subsidiaries' and Parent's employees for purposes of performing its duties and exercising its rights hereunder; provided that the Representative shall treat confidentially and not disclose any nonpublic information from or about the Company or its Subsidiaries to anyone (except as required by law or on a need to know basis to individuals who agree to treat such information confidentially).
- 12.16 Waiver of Conflict; Privilege. After the Effective Time, Parent shall not, and shall cause each of its Affiliates (including the Surviving Corporation) not to use any legal advice provided by Fenwick & West LLP to the Company, the Representative or any Company Equityholder, relating to the Merger ("Company Transactions Legal Advice") in connection with any indemnification claim dispute hereunder or any other legal proceeding or potential legal proceeding against, with or involving Parent, the Surviving Corporation or any of their Affiliates or agents. The Company Transactions Legal Advice shall be deemed to be privileged and confidential communications of the Representative and the Company Equityholders and control of the confidentiality and privilege applicable thereto shall be retained by the Representative and the Company Equityholders. After the Effective Time, the Representative shall be permitted to access and use Company Transactions Legal Advice in connection with any indemnification claim dispute hereunder or any other legal proceeding or potential legal proceeding against, with or involving Parent, the Surviving Corporation or any of their Affiliates or agents; and Fenwick & West LLP, the Representative and any Company Equityholder may make any such Company Transactions Legal Advice available to Fenwick & West LLP or the Representative, as the case may be. In connection with the foregoing, Parent hereby irrevocably waives and agrees not to assert, and agrees to cause the Surviving Corporation to irrevocably waive and not to assert, any conflict of interest arising from or in connection with (i) Fenwick & West LLP's prior representation of the Company and (i) Fenwick & West LLP's representation of the Representative prior to and after the Effective Time. For the avoidance of doubt, Parent, the Surviving Corporation and any of their Affiliates and agents may access and use for any purpose facts, data and any other information contained in any communications between Fenwick & West LLP, on the one hand, and the Company, the Representative or any Company Equityholder, on the other hand, to the extent such communications belong to the Surviving Corporation even if such communication also contains Company Transactions Legal Advice, including as evidence in any indemnification claim dispute or any other legal proceeding or potential legal proceeding involving the Representative or any Company Equityholder, but for the avoidance of doubt, excluding the Company

Transactions Legal Advice contained in such communications. For the avoidance of doubt, nothing in this <u>Section 12.16</u> or in this Agreement shall be deemed to be a waiver of any applicable privileges or protections that can or may be asserted to prevent disclosure of any client communications to any third party. The foregoing shall apply *mutatis mutandis* to the post-Closing representation of the Surviving Corporation by Kirkland & Ellis LLP, which it is agreed shall be permissible in all respects.

12.17 Non-Recourse. Other than for claims or causes of action pursuant to the Stockholder Letters of Transmittal, Option Cancellation Agreements and Warrant Cancellation Agreements contemplated by Article III, all claims or causes of action (whether in contract or in tort, in law or in equity) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), may be made only against the entities that are expressly identified as Parties hereto. Other than for claims or causes of action pursuant to the Stockholder Letters of Transmittal, Option Cancellation Agreements and Warrant Cancellation Agreements contemplated by Article III, no Person who is not a named Party to this Agreement, including any past, present or future director, officer, employee, incorporator, member, partner, equityholder, Affiliate, agent, attorney or representative of any named party to this Agreement ("Non-Party Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or based upon any theory that seeks to impose liability of an entity party against its owners or affiliates) for any obligations or liabilities arising under, in connection with or related to this Agreement or for any claim based on, in respect of, or by reason of this Agreement or its negotiation or execution; and each Party hereto waives and releases all such liabilities, claims and obligations against any such Non-Party Affiliate. The Non-Party Affiliates shall be a third party beneficiaries of this Agreement for purposes of this Section 12.16. Notwithstanding the foregoing, this Section 12.17 shall not apply to Section 12.15, which shall be enforceable in its entirety by the Representative against the Company Equityholders.

\*\*\*\*

IN WITNESS WHEREOF, the parties hereto have executed this Agreement and Plan of Merger as of the date first written above.

### **<u>ULTIMATE PARENT:</u>**

### JAMF HOLDING CORP.

By: /s/ Dean J. Hager

Name: Dean J. Hager

Title: Chief Executive Officer

### **PARENT**:

### JAMF SOFTWARE, LLC

By: /s/ Dean J. Hager

Name: Dean J. Hager

Title: Chief Executive Officer

## **MERGER SUB:**

By: /s/ Dean J. Hager

Name: Dean J. Hager

Title: Chief Executive Officer

[SIGNATURE PAGE TO THE AGREEMENT AND PLAN OF MERGER]

# **COMPANY**:

# WANDERA, INC.

By: /s/ Roy Tuvey
Name: Roy Tuvey
Title: President

[SIGNATURE PAGE TO THE AGREEMENT AND PLAN OF MERGER]

# **REPRESENTATIVE:**

### SHAREHOLDER REPRESENTATIVE SERVICES

**LLC**, solely in its capacity as the Representative

By: /s/ Sam Riffe
Name: Sam Riffe

Title: Managing Director

[SIGNATURE PAGE TO THE AGREEMENT AND PLAN OF MERGER]

### Schedule 1.1

## **Principal Stockholders**

- Eldar Tuvey
- Jim Walker
- John Edwards
- 83North II Limited Partnership83North III Limited Partnership
- Bessemer Venture Partners VIII Institutional L.P
- Bessemer Venture Partners VIII L.P
- Sapphire Venture Fund II, L.P.SAPPHIRE, SAP HANA FUND OF FUNDS, L.P.
- The Gaskaar Trust
- The Twegg Trust

# Schedule 1.2

# **Non-Compete Parties**

- Roy TuveyEldar Tuvey
- Ben Oxnam
- Stijn Paumen
  Mike Campin
  Garry Scott
  John Edwards
  Jim Walker

## Schedule 1.3

### **Non-Solicit Parties**

- 83North II Limited Partnership
  83North III Limited Partnership
  Bessemer Venture Partners VIII Institutional L.P
  Bessemer Venture Partners VIII L.P
  Sapphire Venture Fund II, L.P.

- SAPPHIRE, SAP HANA FUND OF FUNDS, L.P.

# Exhibit A

# **Working Capital Schedule**

# Exhibit B

# Form of Escrow Agreement

# Exhibit C

# Form of Stockholder Letter of Transmittal

# Exhibit D-1

# Form of Warrant Cancellation Agreement

# Exhibit D-2

# Form of Option Cancellation Agreement

#### Certification Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

- I, Dean Hager, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Jamf Holding Corp.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 11, 2021

/s/ Dean Hager
Dean Hager
Director and Chief Executive Officer

#### Certification Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

#### I, Jill Putman, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Jamf Holding Corp.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 11, 2021 /s/ Jill Putman

Jill Putman Chief Financial Officer

#### **Certification of the Chief Executive Officer**

#### Pursuant to Rule 18 U.S.C. Section 1350

In connection with the Quarterly Report on Form 10-Q of Jamf Holding Corp. (the "Company") for the period ended March 31, 2021, as filed with the U.S. Securities and Exchange Commission (the "Report"), I, Dean Hager, Chief Executive Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 11, 2021

/s/ Dean Hager

Dean Hager

Director and Chief Executive Officer

#### **Certification of the Chief Financial Officer**

#### Pursuant to Rule 18 U.S.C. Section 1350

In connection with the Quarterly Report on Form 10-Q of Jamf Holding Corp. (the "Company") for the period ended March 31, 2021, as filed with the U.S. Securities and Exchange Commission (the "Report"), I, Jill Putman, Chief Financial Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 11, 2021

/s/ Jill Putman

Jill Putman

Chief Financial Officer